

Integrated Pollution Prevention and Control (IPPC) & Integrated Pollution Control (IPC)

Interim Sector Guidance for the incineration of waste and fuel manufactured from or including waste



Commissioning Organisation Environment Agency Rio House Waterside Drive Aztec West Almondsbury Bristol BS32 4UD

Tel 01454 624400 Fax 01454 624409

© Environment Agency

First Published 2001

ISBN Applications for reproduction should be made in writing to:

Liz Greenland Environment Agency Scientific and Technical Information Service 2440 The Quadrant Aztec West Almondsbury Bristol BS32 4AQ

All rights reserved. No part of this document may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, electronic, mechanical, photocopying, recording or otherwise, without the prior permission of the Environment Agency.

Record of changes

Version	Date	Change
Consultation	July 2001	Initial draft issue (incinv1)

Note:

Queries about the content of the document should be made to Paul James (07721 390065) or any member of the IPPC Project or Technical Guidance Teams. This draft report is for Internal Consultation and represents the views of the Author. It does not necessarily represent the views or policy of the Environment Agency.

Written comments or suggested improvements should be sent to Graham Winter at the Environment Agency's Technical Guidance Section by email at graham.winter@environment-agency.gov.uk or at:

Environmental Protection National Service Environment Agency Block 1 Government Building Burghill Road Westbury-on-Trym Bristol. BS10 6BF

Telephone 0117 914 2868

Executive Summary

This guidance has been produced by the Environment Agency for England and Wales in collaboration with the Scottish Environment Protection Agency (SEPA) and the Northern Ireland Environment and Heritage Service (EHS). Together these are referred to as "the Regulator" in this document. Its publication follows consultation with industry, government departments and non-governmental organisations.

This document provides interim technical guidance on both Integrated Pollution Prevention and Control, and updated guidance for processes authorised under Integrated Pollution Control.

What are IPPC & IPC

The aims of this Guidance

Integrated Pollution Prevention and Control (IPPC) and its predecessor regime, Integrated Pollution Control (IPC) are regulatory systems that employ an integrated approach to control the environmental impacts of certain industrial activities. They each involve determining the appropriate controls for industry to protect the environment through a single permitting process. To gain a Permit, Operators will have to show that they have systematically developed proposals to apply the 'Best Available Techniques' (BAT) and meet certain other requirements, taking account of relevant local factors.

The Regulators intend to implement IPPC to:

- protect the environment as a whole;
- promote the use of "clean technology" to minimise waste at source ;
- encourage innovation, by leaving significant responsibility for developing satisfactory solutions to environmental issues with industrial Operators; and
- provide a "one-stop shop" for administering applications for Permits to operate.

Once a Permit has been issued, other parts of IPPC come into play. These include compliance monitoring, periodic Permit reviews, variation of Permit conditions and transfers of Permits between Operators. IPPC also provides for the restoration of industrial sites when the permitted activities cease to operate.

This Guidance and the BREF This UK Guidance for delivering the PPC (IPPC) Regulations in the Incineration sector is produced in advance of the BAT Reference document (BREF). The European Commission programme states that it will commence work on the waste incineration BREF in 2001. This guidance will be reviewed and updated as required when the BREF is available.

The aims of this Guidance are to:

- provide a clear structure and methodology which operators making an application should follow to
 ensure that all aspects of the PPC Regulations (see Appendix 2 for equivalent legislation in
 Scotland and Northern Ireland) and other relevant Regulations have been addressed and it should
 thereby assist the Operator to make a satisfactory application;
 - minimise the effort by both Operator and Regulator in the permitting of an installation by use of clear indicative standards and the use of material from previous applications, and from accredited Environmental Management Systems (EMSs);
 - improve the consistency of applications by ensuring that all relevant issues are addressed;
- increase the transparency of the permitting process by having a structure in which the operators
 response to each issue, and any departures from the standards, can be seen clearly;
- improve consistency of regulation across installations and sectors by facilitating the comparison of applications;
- provide a very brief description of the activities to assist the reader to understand the context of the requirements;
- provide a summary of the BAT techniques for pollution control from UK experience which are relevant in the UK context expressed, where possible, as clear indicative standards and which need to be addressed by Applicants;
- provide an arrangement of information which allows the reader to find, quickly all of the guidance associated with:
 - a subject (e.g. accidents, energy or noise) (Sections 2.1 and 2.5 2.11);
 - the technical areas (e.g. furnace requirements) (Sections 2.3 2.4);
 - particular emissions (e.g. NOx or dioxins) (Section 3).

Additionally, to assist Operators in making applications, separate, horizontal guidance is available on a range of topics such as waste minimisation, monitoring, calculating stack heights etc. The majority of this guidance is available free through the Environment Agency, SEPA or EHS (Northern Ireland) web sites (see References).

CONTENTS

1	INTR	ODUCTI	ON	1
	1.1		STANDING IPPC AND BAT, IPC AND BATNEEC	1
	1.2	MAKING		4
	1.3	INSTALL	ATIONS AND PROCESSES COVERED BY THIS NOTE	5
		1.3.1	Existing IPC authorised processes	5
		1.3.2	New and existing IPPC installations	5
		1.3.3	Activities NOT covered by this note	6
	1.4	REVIEW	PERIODS	8
	1.5	KEY ISS	UES FOR THIS SECTOR	9
	1.6	SUMMAR	RY OF RELEASES	.12
	1.7	OVERVI	EW OF THE ACTIVITIES IN THIS SECTOR	.13
	1.8	ECONO	міс Aspects	.18
		1.8.1	Sub-sector specific information	. 18
		1.8.2	Generic cost information required	. 21
n	TECL			22
2	TECF		S FOR FOLLOTION CONTROL	.23
	2.1	Manage	EMENT TECHNIQUES	.24
	2.2	MATERI	ALS INPUTS	.27
		2.2.1	Raw materials selection	. 27
		2.2.2	Waste minimisation (minimising the use of raw materials)	.30
		2.2.3	Water use	.32
		2.2.4	Typical waste compositions	.36
	2.3	THE MA	IN ACTIVITIES (INCLUDING ABATEMENT)	.39
		2.3.1	Incoming waste management	.41
		2.3.2	Waste charging	.51
		2.3.3	Furnace types	. 54
		2.3.4	Furnace requirements	.60
		2.3.5	Chimneys and vents	.68
		2.3.0	Cooling systems (local pollution prevention aspects)	. 69
		2.3.1	Abstement of point source emissions to sir	.// 72
		2.3.0	Abatement of <u>point source</u> emissions to air	.73
		2.3.9	to air by plant type	0 1
		2310	Abstement of point source emissions to surface water and sewer	.0 4 87
		2.3.10	Control of fugitive emissions to air	.07 00
		2.3.11	Control of fugitive emissions to surface water, sewer and groundwater	90
		2313	Odour	02 03
	24	EMISSIC	DNS TO GROUNDWATER	95
	2.5		NG OF WASTES PRODUCED	97
	2.0	WASTE	RECOVERY AND DISPOSAL	
	2.7	ENERGY	/	100
		2.7.1	Basic energy requirements (1)	100
		2.7.2	Basic energy requirements (2)	101
		2.7.3	Further energy efficiency requirements	103
	2.8	ACCIDE	NTS AND THEIR CONSEQUENCES	106
	2.9	NOISE A	ND VIBRATION	110
	2.10	Μονιτο	RING	113
	2.11	DE-CON	IMISSIONING	122
	2.12	INSTALL	ATION WIDE ISSUES	124
3	FMIS		ENCHMARKS	125
•		_		
	3.1		DNS INVENTORY AND BENCHMARK COMPARISON	125
	3.2		ISSION BENCHMARKS	126
		3.2.1	Emissions to air associated with the use of BAT	126
		3.2.2	Emissions to water associated with the use of BAT	120
		J.∠.J 2 0 1	Initoduction to Standards and obligations	120
		J.∠.4 2 2 F	Statistical basis for benchmarks and limits in Permits	120 170
		3.2.3 2 2 6	Deference conditions for releases to sir	120
		3.2.0		120

	3.3	RELEAS	ES TO AIR	129		
		3.3.1	Introduction to the European Directive Emission Limits	129		
		3.3.2	Waste Incineration Directive 2000/76/EC	129		
		3.3.3	Hazardous Waste Incineration Directive 94/67/EC	131		
		3.3.4	Municipal waste incinerators	133		
		3.3.5	Clinical waste incinerators	134		
		3.3.6	Chemical waste incinerators	135		
		3.3.7	Sewage sludge incinerators	136		
		3.3.8	Animal carcass and animal remains incinerators	137		
		3.3.9	Drum incinerators	138		
		3.3.10	Pyrolysis and gasification plants	139		
		3.3.11	Refuse derived fuel	141		
		3.3.12	Co-incinerators	142		
	3.4	RELEAS	ES TO WATER	143		
		3.4.1	Standards and obligations for releases to water	143		
		3.4.2	Legislative emission limit values	144		
4	IMPA	ст		146		
	4.1	Assess	MENT OF THE IMPACT OF EMISSIONS ON THE ENVIRONMENT	146		
	4.2	THE WA	STE MANAGEMENT LICENSING REGULATIONS	148		
	4.3	THE HAR	BITATS REGULATIONS	149		
REF	EREN	CES		150		
DEF	INITIO	NS		152		
APPENDIX 1 - SOME COMMON MONITORING AND SAMPLING METHODS						
APF	ENDI)	(2 - EQ	UIVALENT LEGISLATION IN SCOTLAND & NORTHERN IRELAND	154		

TABLE OF FIGURES

Figure 1-1	- Outline of the Main	Incineration Tec	chniques7	7
i iguio i i			on inquoo	

INTRO	DUCTION	TECHNIQUES			EMISSIONS		IMPACT	
IPPC and BAT	Making an application	Installations covered	allations Review vered periods		issues Summary of releases		Sector overview	Economic aspects

1 INTRODUCTION

1.1 Understanding IPPC and BAT, IPC and BATNEEC

IPPC and the Regulations	Integrated Pollution Prevention and Control (IPPC) and its predecessor, Integrated Pollution Control (IPC) are regulatory systems that employ an integrated approach to control the environmental impacts of certain industrial activities. They involve the determination of appropriate controls for industry to protect the environment through a single permitting process. To gain an IPPC Permit, Operators will have to show that they have systematically developed proposals to apply the 'Best Available Techniques' (BAT) and meet certain other requirements, taking account of relevant local factors.
	Although no new authorisations will be issued under the old IPC regime, certain existing processes will continue to be regulated under IPC (where the Best Available Techniques Not Entailing Excessive Cost or BATNEEC must be used) until IPPC applies to them (see Refs. 3 & 4). This guidance therefore outlines the techniques and standards that are considered to represent both BAT and BATNEEC. While IPPC considerations are wider in many places than IPC the cost/benefit balance for any technique will often be the same. Where the differences between BAT and BATNEEC arise these are highlighted in the text. Where the term BAT is used it should be taken to also mean BATNEEC unless specifically stated.
	The essence of BAT and BATNEEC is that the selection of techniques to protect the environment should achieve an appropriate balance between realising environmental benefits and costs incurred by Operators.
	IPPC operates under the Pollution Prevention and Control (England and Wales) Regulations, (see Ref. 2 and Appendix 2). These Regulations have been made under the Pollution Prevention and Control (PPC) Act 1999 and implement the EC Directive 96/61 on IPPC. Further information on the overall system of IPPC, together with Government policy and more detailed advice on the interpretation of the Regulations, can be found in the Department of the Environment, Transport and the Regions (DETR) document <i>IPPC: A Practical Guide</i> , (see Ref. 3).
Installation based, NOT national emission limits	The "BAT" approach of IPPC is different from regulatory approaches based on fixed national emission limits (except where General Binding Rules have been issued by the Secretary of State). The legal instrument which ultimately defines BAT is the Permit and this can only be issued at the installation level.
Indicative BAT standards	Indicative BAT standards (essentially for BAT but also covering other aspects) are laid out in national guidance (such as this) and should be applied unless there is strong justification for another course of action. It should be noted that BAT includes both the technical components of the installation given in Section 2 and the benchmark levels identified in Section 3. Departures from those standards, in either direction, can be justified at the local level taking into account the <u>technical characteristics</u> of the installation concerned, its <u>geographical location</u> and the <u>local environmental conditions</u> . Notwithstanding this, the mandatory EU emission limits and process standards that relate to waste incineration processes must be met first, although BAT may dictate even tighter controls.
BAT and EQSs	The "BAT" (and BATNEEC) approach is also different from, but complementary to, regulatory approaches based on Environmental Quality Standards (EQS). Essentially BAT requires measures to be taken to prevent or, where this is not practicable, to reduce emissions. That is, if emissions can be reduced further, or prevented altogether, at reasonable cost, then this should be done irrespective of whether any environmental quality standards are already being met. It requires us not to consider the environment as a recipient of pollutants and waste, which can be filled up to a given level, but to do all that is practicable to minimise the impact of industrial activities. The process considers what can be reasonably achieved within the installation first (this is covered by Sections 2 and 3 of this Guidance) and only then checks to ensure that the local environmental conditions are secure, (Section 4 of this Guidance and Ref. 5). The BAT (and BATNEEC) approach is, in this respect, a more precautionary one, which may go beyond the requirements of Environmental Quality Standards.
	Conversely, it is feasible that the application of what is BAT/BATNEEC may lead to a situation in which an EQS is still threatened. The Regulations therefore allow for expenditure beyond BAT/BATNEEC where necessary. However, this situation should arise very rarely assuming that the EQS is soundly based on an assessment of harm. The BAT/BATNEC assessment, which balances cost against benefit (or prevention of harm) should in most cases have come to the same conclusion about the expenditure which is appropriate to protect the environment.
	Advice on the relationship of environmental quality standards and other standards and obligations is given in <i>IPPC: A Practical Guide</i> (see Ref. 3). General information relevant to this sector and specific requirements for each substance are given in Section 3.

INTRODUCTION		TEC	TECHNIQUES			SSIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Assessing BAT/BATNEEC at the sector level	The assessment of BAT takes place at a number of levels. At the European level, the EC issues a BAT reference document (BREF) for each sector. The BREF is the result of an exchange of information which member states should take into account when determining BAT, but which leaves flexibility to member states in its application. The European BREF for waste incineration is not expected to be published until 2003. This UK Guidance is therefore produced in advance of the BREF and lays down the indicative standards and expectations in the UK. This guidance will therefore be revised as necessary when the BREF is available. At the national level, techniques which are considered to be BAT should, first of all, represent an appropriate balance of costs and benefits for a typical, well-performing installation in that sector. Secondly, the techniques should normally be affordable without making the sector as a whole uncompetitive either on a European basis or worldwide.
	Similar considerations apply to the assessment of BATNEEC for existing IPC authorised processes. Although there is no formal link to European guidance, UK guidance on BATNEEC takes account of industry sector performance and standards outside as well as inside the UK.
Assessing BAT/BATNEEC at the installation level	When assessing the applicability of the sectoral, indicative BAT/BATNEEC standards at the installation level departures may be justified in either direction as described above. The most appropriate technique may depend upon local factors and, where the answer is not self evident, a local assessment of the costs and benefits of the available options may be needed to establish the best option. Individual company profitability is not considered.
	In summary, departures may be justified on the grounds of the technical characteristics of the installation concerned, its geographical location and the local environmental conditions but not on grounds of individual company profitability. Further information on this can be found in the IPPC Guide for Applicants, (see Refs. 3 and 4).
	While BAT/BATNEEC cannot be limited by individual company profitability, company finance may be taken into account in the following limited circumstances:
	 where the BAT cost/benefit balance of an improvement only becomes favourable when the relevant item of plant is due for renewal/renovation anyway (e.g. BAT for the sector may be to change to a different design of furnace when a furnace comes up for rebuild). In effect, these are cases where BAT for the sector can be expressed in terms of local investment cycles.
	• where a number of expensive improvements are needed, a phasing programme may be appropriate as long as it is not so extended that it could be seen to be rewarding a poor performing installation, (see Ref. 5 for more details).
Innovation	The Regulators encourage the development and introduction of new and innovative techniques which meet the BAT/BATNEEC criteria and are looking for continuous improvement in the overall environmental performance of the process as a part of progressive sustainable development. This Note describes the appropriate indicative standards at the time of writing. However, Operators should keep up to date with the best available techniques relevant to the activity and this Note may not be cited in an attempt to delay the introduction of improved, available techniques. The technical characteristics of a particular installation may allow for opportunities not foreseen in the Guidance; as BAT/BATNEEC is ultimately determined at the installation level (except in the case of GBRs) it is valid to consider these even where they go beyond the indicative standards.
New installations	The indicative requirements apply to both new and existing activities but it will be more difficult to justify departures from them in the case of new activities. For new installations, the indicative requirements should normally be in place before the commencement of operations. In some cases, such as where the requirement is for an audit of ongoing operations, this is not feasible and indicative upgrading timescales are given for such cases.
Existing installations - standards	For an existing activity, a less strict proposal (or an extended timescale) may, for example, be acceptable where the activity already operates to a standard that is very close to an indicative requirement (see Section 2 for further guidance).
	Upgrading timescales will be set in the improvement programme of the Permit. Improvements fall into a number of categories:
	 the many good practice requirements in Section 2, such as management systems, waste, water and energy audits, bunding, good housekeeping measures to prevent fugitive or accidental emissions, energy baseline measures, waste handling facilities and monitoring equipment;
	the larger, usually more capital intensive improvements;
	longer term studies required for control, environmental impacts etc.
	All improvements should be carried out at the earliest opportunity and to a programme approved by the Regulator.

INTRODUCTION			TECHNIQUES			EMISSIONS		IMPACT	
IPPC and BAT	Making an application	Insta co	allations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Specific improvements may be required to ensure compliance with the standards outlined in the Waste Incineration Directive (2000/76/EC) and national legislation implementing the Directive. These improvements should be carried out within the timescales given below. The whole programme (including any other items outlined in this guidance but not required by the Directive) should be completed <u>at the latest</u> within 3 years of the issue of the Permit. Any longer timescales will need to be justified by the Operator in accordance with the principles above.

Improvement	By:
<i>New</i> incineration installations covered by the Waste Incineration Directive (2000/76/EC) to meet at least the standards outlined in WID	From the first day of operation – with immediate effect.
<i>New</i> incineration installations that are <i>not</i> covered by WID to meet those standards outlined in the WID that are also BAT (i.e. Operators must specifically justify any departures from WID).	28 December 2005 or earlier of BAT dictates
<i>Existing</i> incineration installations covered by WID to meet the standards of WID	28 December 2005 or earlier if BAT dictates
<i>Existing</i> incineration installations that are <i>not</i> covered by WID to meet those standards outlined in the WID that are also BAT (i.e. Operators must specifically justify any departures from WID).	28 December 2005 or earlier if BAT dictates

The Applicant should include a proposed timetable covering all improvements.

This guidance provides further information on techniques that may be appropriate for meeting BAT and WID.



- **Note 1** The amount of detail needed to support the application should be sufficient to support the Applicant's contention that either the conditions of the guidance have been met or an alternative measure has been justified. The level of detail should be commensurate with the scale of the operation and its ability to cause pollution. An Applicant is not required to supply detail that could not reasonably be expected to contribute to a decision to issue a Permit.
- **Note 2** For existing IPC or Waste Management Permit holders, your response to each point in Sections 2, 3 or 4 may rely heavily on your previous application. The Regulator does not wish you to duplicate information as long as the previous information adequately addresses the issues. However, the more the information can be reorganised to demonstrate that all the issues have been adequate addressed the better. You will need to send us copies of any information referred to.
- **Note 3** The contents of the outlined BAT boxes in Sections 2, 3 and 4, and additional blank tables etc., are available electronically on the Environment Agency's Website, for the assistance of Applicants.

INTRO	DUCTION	TEC	TECHNIQUES		EMISSIONS		IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

1.3 Installations and Processes Covered by this Note 1.3.1 Existing IPC authorised processes

Existing IPC Authorised Processes described as Part A processes in schedule 1 of The Environmental Protection (Prescribed Processes and Substances) Regulations 1991 SI472 (as amended) under the following sections:

5.1(a) & 5.1(b)	- chemical waste incinerators
5.1(bb)	- certain plants burning certain solid or liquid "hazardous waste"
5.1(c)	 incinerators (for any waste) of capacity > 1te/hr
5.1(d)	- the burning out of metal containers
1.3(c)	- the burning of fuels derived from or comprising wastes in dedicated plant

Although not primarily intended for processes whose primary purpose is the generation of energy or production of material products, the information in this guidance may also be applicable these processes where they burn waste or waste derived fuels. These processes are likely to be most aptly authorised under other sections of SI472. Guidance on the interpretation of the sections outlined above is provided in SI472 (as amended).

No new IPC authorisations can be issued after 31 December 2000. The specific IPC information provided in this guidance will therefore only be applicable to processes which are already authorised under IPC and do not require an IPPC permit. Existing IPC processes will come under IPPC in one of three ways:

- Permitting according to the transitional schedule in the Regulations (Ref. 2);
- Permitting ahead of the schedule by agreement;
- Permitting ahead of the transitional schedule in the case of a "substantial change".

Further guidance on when an IPPC permit is required is provided elsewhere (Ref. 3).

1.3.2 New and existing IPPC installations

New and existing IPPC installations described as Part A(1) activities in schedule 1 of the Pollution Prevention and Control (England and Wales) Regulations 2000 S.I.1973, under the following sections:

5.1(a) & 5.1(b)	- chemical waste incinerators
5.1(c)	- certain plants burning certain solid or liquid "hazardous waste"
5.1(d) & (e)	- incinerators (for any waste) of capacity > 1te/hr
5.1(f)	- the burning out of metal containers
1.1(c)	- the burning of fuels derived from or comprising waste in dedicated plant

It is important to note that the Regulations give detailed guidance on the interpretation of the sections outlined above. For example they state that "incineration includes pyrolysis". As such, the pyrolysis of waste materials is included in this guidance.

The installation includes the main activities as stated above and associated activities which have a technical connection with the main activities and which may have an effect on emissions and pollution. They include, as appropriate:

- storage and handling of raw materials;
- storage, handling, preparation, selection and management of incoming waste;
- the control and abatement systems for emissions to all media;
- water abstraction and treatment plant;
- cooling systems;

- heat transfer systems;
- energy recovery and power production systems;
- storage and despatch of waste and other materials;
- on-site waste handling and recycling facilities.

Figure 1-1 shows the main operations.

INTRODUCTION		TEC	TECHNIQUES			SSIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

However, the impact of the activities on the environment may be wider than just the on-site activities. The Note, and the Regulations, cover issues downstream of the installation such as the disposal of wastes and wastewaters.

Advice on the extent of the physical site which is contained within the installation, for example split sites, is given in *IPPC Part A(1) Installations: Guide for Applicants,* see Ref. 4. Operators are advised to discuss this issue with the Agency prior to preparing their application. Particular examples relevant to incineration installations would be:

- A site includes an ash treatment and/or recycling centre;
- A site includes a waste transfer operation for separately delivered recyclable materials;
- A site includes a materials recovery facility, the residues of which are sent to the incinerator.

In each of these cases it will be necessary to consider guidance provided in *IPPC: A Practical Guide* (Ref. 3). The extent to which the associated activity is dedicated to the incineration unit will become a part of the Regulator's decision.

Where associated activities are carried out in conjunction with the main activities and are not covered in this guidance note (for example ash treatment operations), reference should be made to:

- other relevant IPPC Guidance Notes (e.g. S0.01 Guidance for the Waste Treatment Sector) and,
- other relevant guidance notes issued under EPA 90 (e.g. Ref. 20 Monitoring),
- where appropriate, the Secretary of State's Guidance for Local Authority Air Pollution Control. (NB In Northern Ireland this guidance is produced by the Department of the Environment),
- Agency guidance on waste management (e.g. Guidance on Clinical Waste Management)

For this sector, this would apply in particular to guidance on solid waste (ash) treatment plant, the design and operation of materials recovery facilities and waste derived fuel *production* plants.

1.3.3 Activities NOT covered by this note

This note provides guidance where the **primary purpose** of the installation is the destruction of wastes or waste derived fuels in **dedicated plant**. There are many situations where wastes or waste derived fuel are burned as a substitute for, or in addition to, primary fuels in the course of a manufacturing or other operation.

For example, waste or waste derived oils are sometimes burned in power stations, and waste solvents or tyres may be burned in a cement works. Such processes are likely to be "most aptly" authorised not as "incinerators" but under separate, specific sections of the Regulations as, for example a power station or cement works. The techniques that represent BAT for these installations will be **outlined in their own sector specific guidance**. As such, this guidance stops short of outlining the design or operational techniques appropriate in such circumstances but concentrates upon "dedicated incineration plant". This guidance does, however, outline techniques that should be **considered** when determining BAT for the prevention of releases at all plats where waste is burned.

It is important to note that, where wastes are co-combusted with other fuels, the provisions of European legislation (Ref. 25) will require implementation via the IPPC Permit. This may include a requirement to determine emission limit values for the installation on a pro-rata basis, according to the heat derived from the primary and substitute fuel, or specific emission limits. Each of these situations will need to be assessed individually. Therefore, if an Operator proposes to use fuels derived from, or comprising wastes it is essential that this is discussed with the Regulator at an early stage.

INTRO	DUCTION	TEC	TECHNIQUES			SIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects



Figure 1-1 - Outline of the Main Incineration Techniques

INTRODUCTION			TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application	Insta co	allations overed	Re pei	eview riods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

1.4 Review Periods

Permits can be reviewed or varied at any time. However, the PPC Regulations impose a requirement on Regulators to review Permits in certain specific circumstances such as where the pollution caused by the installation is of such significance that the existing emission limit values need to be revised or new limits set.

In addition, Regulators are required to review the conditions of Permits "periodically". The Government stated in its third consultation paper (England, Wales and Scotland) on the implementation of IPPC, that the new sector-specific IPPC Sector Guidance Notes would provide guidance on appropriate review periods for each sector. These would take into consideration guidance on the relevant criteria, to be provided by the Government. Examples of the likely relevant criteria for setting these review periods are "the risk and level of environmental impacts associated with the sector" and "the cost to the Regulators and regulated industry of undertaking the reviews".

The Regulators consider that at the present time, having regard to those criteria, it is in fact appropriate to set indicative minimum review periods which differ only between those sectors which have been subject to integrated permitting (i.e. IPC or Waste Management Licensing), and those which have not. It is therefore proposed that Permit conditions should normally be reviewed on the following basis:

- for individual activities NOT previously subject to regulation under IPC or Waste Management Licensing, a review should normally be carried out within four years of the issue of the IPPC Permit;
- for individual activities previously subject to regulation under IPC or Waste Management Licensing, a review should normally be carried out within six years of the issue of the IPPC Permit.

This means that activities/installations not currently in IPC or Waste Management Licensing will be initially reviewed within four years and thereafter within six years.

An exception to this is where discharges of List I or List II substances have been permitted or where disposal or tipping for the purposes of disposal, of any matter which might lead to an indirect discharge of any substance on List I or II. In such cases the review must be carried out within four years.

This period will be kept under review and, if any of the above factors change significantly, may be shortened or extended.

The information in this guidance will itself be reviewed following issue of the BREF. It is anticipated that the BREF will be available in 2004.

INTRO	TE	TECHNIQUES				SIONS	IMPACT		
IPPC and BAT	Making an application	Installation covered	s I	Review periods	Key	issues	Summary of releases	Sector overview	Economic aspects

1.5 Key Issues for this Sector

European legislation

The incineration of waste is subject to specific European legislation, setting baseline operational and emission standards for the majority of installations burning waste or fuels derived from waste, either alone or in combination with other fuels. Where they apply, the standards of the Directives must be met as a minimum. If the application of BAT / BATNEEC results in tighter limits than these standards then the tighter limits must be applied.

The Environment Agency intends to produce separate "regulatory guidance" on the new Waste Incineration Directive (WID 2000/76/EC). The enabling legislation will aim to implement the Directive requirements, and WID will eventually apply in the majority of situations. The standards set out in this guidance take direct account of the directive (as well as other relevant legislation).

The scope of WID is wide and will cover the majority of situations where waste is burned in installations either alone or in combination with other fuels. For definitive guidance on the scope of the directive, readers should refer to Article 2 of the directive where provision is made for the exclusion from WID requirements of those installations burning ONLY certain types of waste. Interpretation of the directive is complex and it is recommended that this is discussed with the Regulator well before an application is submitted.

Of particular interest to this sector will be the directive emission limit values for releases to air which are outlined in Section 3. In addition, WID (and other Directives that it will replace by 28 December 2005, e.g. HWID) sets a variety of operational standards in respect of: delivery and reception of waste; combustion requirements; residue composition; control and monitoring; abnormal operating conditions. *These requirements have been incorporated in the relevant sections of this guidance when outlining what is considered to be the BAT.*

Feedstock composition

Many wastes vary in terms of physical and chemical composition. The nature of the waste to be treated is an important factor in determining what is BAT for an individual installation. Operators must demonstrate that they have designed their installations such that the operating envelope is sufficiently wide to ensure emission limits are met over the range of operational conditions that will be encountered. In some cases it may be necessary for Operators to restrict the wastes types burned or pre-treat the waste feed in order to prevent short-term exceedences of emission limits.

Feedstock management / storage / preparation

Where waste feed stocks are more highly heterogeneous in their nature it is more likely that greater effort will be required to manage that waste such that the installation design envelope is not exceeded. This may include consideration of upstream waste collection as well as on-site sorting and pre-treatment. Further details regarding suitable techniques and the criteria for their selection are provided in Sections 2.2 and 2.3.

Furnace requirements and combustion control

There are a wide variety of furnace types. Whilst one design may be well suited to a particular waste stream it may not be capable of treating another waste. The physical properties of the waste are usually the key parameters that should be considered when selecting a particular furnace design e.g. fluidised bed furnaces require wastes fed to them to be of a particular particle size range.

In all cases it is necessary to ensure that the overall process ensures wastes are effectively and thoroughly treated so as to minimise polluting emissions. In particular this includes a need to ensure that releases attributable to inefficient combustion are prevented through the optimisation of combustion conditions. Optimisation should aim to achieve steady state process conditions through the application of suitable controls. These may include waste pre-treatment as well as in-process methods. Furnace types and requirements are further discussed in Sections 2.3.3 and 2.3.4 respectively.

Energy recovery and boiler design

The process of combustion gives rise to very significant quantities of heat. The heat generated should be recovered as far as practicable (this is a WID requirement). This will include consideration of opportunities to increase energy recovery through for example, electricity generation, combined heat and power, the generation of process steam or district heating, or combinations of these.

INTRODUCTION		TEC	TECHNIQUES			SIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Key	/ issues	Summary of releases	Sector overview	Economic aspects

Whilst increasing energy recovery is important, the use of wastes (and waste derived fuels) as a source of energy must be a secondary consideration after ensuring their safe destruction. For example, in many incineration plant energy recovery achieved by raising steam in a boiler. It is important that the energy recovery itself does not give rise to the production of additional polluting species. Boiler design and energy recovery are therefore important and are further considered in Sections 2.3.7 and 2.7 respectively. Where waste are used as a regular or additional fuel it is a requirement of European legislation that its co-incineration should not cause higher emissions of polluting substances than those dedicated to waste incineration.

Abatement technology and its operation

The provision and effective operation of flue gas cleaning equipment is essential and a key step to meeting mandatory emission standards and minimising the release of pollutants that cannot be prevented through the use of up-stream techniques. Air abatement issues are discussed in Section 2.3.8. A breakdown by plant type (i.e. municipal, clinical etc) is also included in that section.

Residues handling and disposal

The purpose of an incinerator is to maximise the safe destruction waste and to minimise the production of residues in terms of their quantity and harmfulness. The nature of the wastes treated and the throughput dictate the quantities of waste produced. Waste is in the form of grate or bottom ash, fly ash, and residues from the air pollution control (APC) equipment. In some plants the fly ash and APC residues are combined. All of these ashes have the potential to be difficult to handle on account of their physical characteristics, some may be classed as special wastes (e.g. APC residues). All installations must make adequate provision for the on-site management of these wastes.

It is possible that significant quantities of some types of ash can be recycled. If disposal of ash is planned, the Operator will be required to justify why recovery is "technically and economically impossible" for it to be recovered, together with "the measures planned to avoid or reduce any impact on the environment". These matters are further discussed in Sections 2.2, 2.5 and 2.6.

Monitoring and reporting

European Directives include minimum monitoring requirements. Continuous emission monitors (CEMs) are specified in many cases. Regular spot monitoring is required to back up CEM measurements, along with CEM calibration and confidence level assurances. On-line dial-up Regulator access to monitors and historical emissions data is considered BAT in some circumstances. Further details are included in Section 2.10.

Public perception and planning issues

Public perception of incineration is generally poor. Proposals for new waste incinerators or the incineration of waste derived fuels are frequently accompanied by significant concern in local communities. The most common concerns relate to the perceived impact of incinerator emissions to air upon human health and the environment. The alternatives to incineration, and the selection of the site concerned are also frequently guestioned.

Although still the thermal treatment of waste, and usually incorporating a combustion stage, pyrolysis and gasification may be proposed as alternatives to incineration. For many waste types, the emission performance of such plant has yet to be demonstrated consistently on a commercial scale. Whilst well suited to some waste types and offering some potential advantages, the data concerning the general application of these techniques and their economics is in many cases not fully developed (see also Section 1.7 - pyrolysis and gasification)

Some of the issues raised (e.g. traffic, visual amenity, location) fall within the remit of the planning authority rather than to the Regulator and should be addressed in a planning application. Guidance on the content of planning applications should be sought from the relevant planning authority. Environmental impact assessments (EIA) are a likely requirement of the planning process. There is considerable scope for assessments submitted in respect of the EIA to be provided in support of the Operator's application. In order to avoid unnecessary duplication it is very important that Operators discuss the scope of such assessments with the Regulator, the planning authority and other stake holders at an early stage.

The Regulator considers that Operators and developers of waste burning installations have a duty to inform and consult with local communities. The Regulator will work closely with local planning authorities in order to improve communication of the facts relating to particular proposals and will place information regarding emissions in the public domain.

INTRODUCTION			TECHNIQUES			EMIS	SIONS	IMPACT	
IPPC and BAT	Making an application	Insta co	allations overed	Review periods	Key	y issues	Summary of releases	Sector overview	Economic aspects

Waste strategy

Waste strategy is of particular relevance to municipal waste incineration. Waste strategies for Scotland, England and Wales were published in 2000. Statutory targets for the recycling and recovery of waste were published and a hierarchy of waste disposal options (with landfill as the least desirable option) was included. Restrictions upon the quantities of certain types of waste that may be landfilled were set in European legislation (Ref. 25) in 1999.

Although waste prevention will become increasingly important, evidence of waste arising trends suggests there is a continued need to plan suitable disposal outlets. The wastes that are produced, that can no longer be landfilled, and cannot practicably be recycled will therefore require disposal. This could lead to an increase in the number of municipal incineration plants.

When considering the strategic need for incineration the requirements of Government policies (including consideration of the Best Practicable Environmental Option), regional and local waste disposal plans should be reflected in the size and location of plant. Matters relating to this are considered further in Sections 2.6 and 4.2.

Incineration is likely to have a continuing role in ensuring the BPEO is selected for the management of waste.

Noise

There are noise sources at waste incineration installations that should be addressed (see Section 2.9).

Odour

Odour may arise from storage and handling of raw waste, poor site hygiene or from stack emissions where combustion is poor. All or these situations are avoidable through the use of BAT (see Section 2.3.13).

Accident risk

Apart from the normal process and spillage risks, risks of releases to air through abnormal operating conditions must be considered. Interlocks are required to prevent waste feed when combustion conditions do not meet the required standards. Other accidental release scenarios should be identified and suitable engineering or procedural measures put in place. Shut-down of the process is required under certain circumstances. Further details are provided in Section 2.8.

Site restoration

Some incinerators will have been operating on the same site for many years. Others may be constructed on the site of former industrial sites including power stations and waste management sites. There may well be historical ground contamination that could be confused with potential future contamination from the activities as they will be operated under IPPC. In such cases it will be necessary to assess the degree of contamination as a baseline for future operations.

Long range trans-boundary pollution

Dedicated waste incineration plants are unlikely to be of sufficient size to have significant transboundary effects. Power stations and other processes that co-incinerate waste will need to consider this aspect. Further guidance may be found in the sector specific guidance for those processes.

INTRODUCTION			TECHNIQUES			EMI	SSIONS	IMPACT	
IPPC and BAT	Making an application	Installa cove	ations ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

1.6 Summary of Releases

RELEASES	Particulate matter	Hydrogen chloride	Oxides of sulphur	Oxides of nitrogen	Oxides of carbon	Dioxins	Organic compounds / odour	Mercury & cadmium	Other heavy metals	Alkali metals & oxides and alkali earth metals & oxides	Acids/alkalis/salts	Hydrogen fluoride
Applicable to:												
Chemical waste	Y	Y	Y	Y	Y	Y	Y	Y	Y		Y	Y
Clinical waste	Y	Y	Y	Y	Y	Y	Y	Y	Y		Y	Y
Municipal waste	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Animal carcasses	Y	Y	Y	Y	Y	Y	Y			Y	Y	
Sewage sludge	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Occurs in:												
Combustion gases	А	А	А	А	А	А	А	А	А			А
Waste storage and heating Note1,2	AL									WL		
Boiler water treatment & blowdown											W	
Scrubber liquor / sludge residues	WL					WL	WL	WL	WL	WL	WL	
Grate ash, fly ash and sorbents	AL					AL	AL	AL	AL	AL	AL	
Occasional releases from ash quenching	W					W	W	W	W	W	W	
KEY	KEY A – Release to Air, W – Release to Water, L – Release to Land											

Notes:

1 2 3

Includes separated sludges released to land. Includes disinfectant releases to sewer. Heavy metal releases to land, may be present in both organic and inorganic forms, and similarly for organic compounds, which may be organometallic compounds.

INTRODUCTION			TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application	Insta co	allations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects	

1.7 Overview of the Activities in this Sector

No. of UK Part A Installations	
Municipal waste incinerators	12 sites
Clinical waste incinerators	?
Chemical waste incinerators	3 sites
Refuse derived fuel incinerators	3 sites
Animal carcass incinerators	?
Sewage sludge incinerators	?
Waste pyrolysers	?
Waste gasifiers	?
Drum incinerators	2 sites

Summary of the activities

This section provides a very brief description of incineration activities.

The industry provides a means for the disposal of a variety of waste types, and depending upon the type of waste, to recover energy from that waste. If not designed and operated properly installations have the potential for significant releases to water, air and land.

Common process stages

Although technologies differ slightly depending upon the precise nature of the feedstock (e.g. grate, hearth and abatement plant design differences) there are common purpose process stages to all waste incinerators.

The main components are:

- Feed stock storage, handling and possibly treatment;
- Furnace loading and charging operations;
- Solid waste transport in furnace;
- Furnace design, e.g. rotary hearth, moving grate, fluidised bed;
- Residue removal and management, i.e. ash;
- Combustion control, e.g. air staging, NOx control;
- Gas residence, cooling and heat transfer;;
- Energy recovery, e.g. steam boilers, turbine generation;
- Gas cleanup equipment, e.g. reagent injection, contact vessels, scrubbers, baghouses etc.;
- Discharge of gases.

Pyrolysis and Gasification:

Pyrolysis and gasification plants differ from this general layout in that they include a starved air stage as the initial thermal treatment. However, when combined with a subsequent combustion stage in order to recover energy (rather than converting the waste feed to a waste derived fuel or other product) they too contain essentially the same process steps.

Municipal waste incineration (MWI)

Installations accept domestic waste and some commercial and industrial wastes of a similar character to domestic waste. The number of municipal waste incineration installations is predicted to increase significantly over the next 10 to 20 years as alternatives to landfill are sought for wastes that cannot practicably be reduced or recycled.

Some of the new installations will replace those that closed in the 1990s because they could not be upgraded to meet the requirements of European legislation. Modern MWIs must achieve the tightest standards required by European legislation. The emission levels set for waste incineration are generally lower than for other industrial combustion sectors. Although each case will need to be assessed on its merits, at the legislative emission levels, incinerators are not expected to give rise to a significant deterioration in local environmental quality.

Enhanced recycling of the municipal and commercial wastes usually disposed of at such plant is not expected to alter the calorific value of feed-stocks as the removal of combustible fractions (paper, plastic etc) is counteracted by the removal of metals, glass and (higher moisture content) vegetable matter. Despite this, it is still necessary to consider the mode of waste collection and segregation in the catchment area, and any pre-treatment, or sorting stages, in order to predict the composition of the intended waste feedstock. Only once knowledge of the feedstock is secured will it be possible to select the BAT for the incineration of that waste.

INTRODUCTION			TEC	HNIQUES	5	EMIS	SSIONS	IMPACT	
IPPC and BAT	Making an application	Installa cove	ations ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Energy should be recovered from the waste in the form of electricity and / or steam. Heat may be supplied for local use in some cases.

INTRO	DUCTION	TEC	TECHNIQUES			SSIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Clinical waste incinerators (CWI)

CWIs provide for the disposal of healthcare wastes arising typically from hospitals, veterinary and dental practices, and pharmaceutical research plants. Some of the wastes produced in these locations may be classed as hazardous wastes and it is therefore important that good segregation of such waste is practised at the point of the waste arising in order to demonstrate that the provisions of the Hazardous Waste Incineration Directive (Ref. 26) will not apply. In practice, the quantities and proportion of such waste arising tends to be very small and are unlikely to alter the performance of a typical CWI.

Many CWIs are situated on the site where the waste is produced. The capacity of the majority of these is below 1 te/hr and are therefore part B installations and are not covered here. This guidance only relates to CWIs of a capacity greater than 1te/hr, unless they burn certain hazardous wastes as defined in the legislation (Ref. 26). Where Operators are unsure of the requirements they should discuss their situation with the Regulator.

Sanitation of certain clinical wastes may be achieved using autoclaves, or microwave technology. Having used these techniques to reduce the immediate biological hazards, the resultant waste may then be incinerated as non-clinical waste.

The provision of an effective, rapid and hygienic disposal route is of particular importance in this sector. This is particularly the case where wastes are of an infectious or unpleasant nature. Particular care is required to ensure that the health and safety of plant operators is considered when designing the waste reception, storage and transfer stages of the installation. Every attempt should be made to eliminate physical contact with the waste, ensure it is soundly packaged in accordance with relevant health and safety guidance – needles and other sharps are a particular concern.

Energy is usually recovered, commonly in the form of steam to heat the hospital at which the incinerator is located, thus saving on primary fuels.

Chemical waste incinerators

Chemical waste incinerators provide a means of disposal for hazardous wastes. They may be operated in-house by a company wishing to destroy its own wastes, or merchant incinerators which receive wastes of a wide array of types, and from a range of sources.

At merchant incinerators or those in-house ones taking a variety of waste streams, the variation in the types of waste, and their high hazard and environmental pollution potential means that such process must adopt the very highest technological and management standards.

The Hazardous Waste Incineration Directive (HWID) must be complied with at all installations burning hazardous wastes (including co-incineration at production plants or power stations).

Opportunities for energy recovery are significant for some waste streams (e.g. solvent wastes).

Sewage sludge incinerators

Interest in the disposal of sewage sludge by thermal means has increased with the removal of alternative disposal outlets (e.g. disposal at sea). There are a number of plants operating and proposed.

Fluidised bed technology has proven effective in this sub-sector and may help to achieve compliance with WID owing to the reduced thermal NOx generation compensating for relatively high fuel nitrogen. There are a number of projects that are considering the use of pyrolysis and gasification.

Industrialised sewage catchments can lead to elevated levels of metals in the waste feed. Abatement design must therefore carefully consider this. The scope of the IPPC installation may in some cases extend to the sewage treatment works on the same site.

Energy recovery opportunities are significant. It may be possible for gasification / pyrolysis to be coupled with engines or gas turbines in order to increase electricity recovery. It is common for combustion heat (or steam) to be re-circulated for sludge drying or other applications.

Animal carcass and animal remains incinerators

Between 1995 and 2000 the number of ACIs in England and Wales increased from 1 to 6. The key driver for this expansion has been the response to BSE. Capacity of ACIs was reported to be 4000 carcasses per week in the year 2000. With culled cattle arising from the over thirty months scheme (OTMS) numbering in excess of 10,000 per week, stock piles of rendered meat and bone meal (MBM), specified risk material (SRM) and insufficient overall alternative disposal capacity, the number of plants in this sector looks set to rise.

INTRODUCTION			TECHNIQUES			EMIS	SSIONS	IMPACT	
IPPC and BAT	Making an application	Insi C	tallations overed	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Installations burning <u>only</u> animal carcasses are presently exempted from the Waste Incineration Directive (WID) but must comply with other specific legislation. Installations that burn other animal remains (including MBM) do fall within WID from the dates specified.

In general the design and operational standards of this sector has been lower than other sectors, and will need improvement to meet WID standards.

Drum incinerators

This sector provides a means of burning out the contents of contaminated metal drums such that they may be re-used.

The application of HWID has lead to a decline in the number of plants operating in this sector. It would appear unlikely that there will be growth in this sector.

Pyrolysis & gasification plants

Pyrolysis and gasification are a type of thermal waste treatment that may have application across a range of waste types and offer a potential alternative to conventional combustion. Commercial scale plants have been developed for the treatment of selected waste types that fall within a reliable physical and chemical specification e.g. some sewage sludges and wood wastes.

There are *theoretical* advantages in that the thermal degradation stages are split into two by first generating a gas, and then subsequently combusting that gas. This can improve opportunities for process control and hence reduce emissions. Liquid and solid residues can also be recovered – these may have further uses with some, well segregated waste streams.

A further *theoretical* advantage is to be gained from the burning of the gas produced in gas engines or gas turbines. This would improve the efficiency of energy recovery as these generation technologies do not have the same inherent losses a steam turbines.

Despite these theoretical advantages the techniques have yet to be developed widely and reliably at a commercial scale, although some successful plants have been developed for specific waste streams (e.g. sewage sludge). The economics of this sub-sector would appear to suggest that wider application, to include general municipal waste streams would only become viable at disposal costs greater than those in the UK. However, coupled with systems to segregate, select and pre-treat waste they may become competitive for individual waste fractions.

Despite some claims for greatly reduced emissions compared to mass burn incineration there is insufficient data to draw this conclusion on a general basis.

Comparison of Pyrolysis and Gasification with Incineration:

The following points should be noted:

- There is wider variation in the processes with pyrolysis and gasification plants than is the case for combustion plants. Generalisations regarding performance must therefore be regarded with caution.
- Many applications are at an early stage of development and performance data limited and sometimes conflicting.
- Pyrolysis and gasification generally require waste pre-treatments to produce a more homogeneous fuel than is required by combustion plants. This pre-treatment may lead to reductions in emissions spikes although the same would be true for incineration with pre-treatment.
- Pyrolysis and gasification plants may be able to recover a higher proportion of energy from the fraction of waste burned by burning the gas produced in engines or gas turbines. However, achieving directive requirements regarding combustion residence time and CO and VOC emissions limit values is a concern.
- Fuel bound nitrogen may be converted to ammonia and hydrogen cyanide which may require removal prior to combustion to prevent elevated NOx levels.
- Solid and liquid residues have the potential to contain higher levels of organic pollutants and heavy metals than incineration and further treatment may be required.
- There would appear to be little evidence to support claims that pyrolysis and gasification emit lower amounts of dioxins to air than modern incineration installations.
- Smaller plants appear to be more cost competitive with incineration than larger plants in the municipal sector.
- Niche applications for some well-characterised wastes streams would appear to offer the main market for this technology.

INTRO	DUCTION	TEC	TECHNIQUES			SSIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Refuse derived fuel plants

Fuels may be derived from a variety of combustible wastes with either no pre-treatment or complex treatment depending on the source and nature of the waste.

Types of RDF include:

- Waste oil
- Waste solvents e.g. secondary liquid fuels (SLF)
- Municipal RDF
- Agricultural residues e.g. straw, poultry litter
- Tyres
- Industrial bio-mass residues e.g. wood, food residues, paper sludge

Some of these are already subject to HWID. In the majority of cases (excluding certain clean wood and biomass residues specified in WID) the remainder will be required to meet the standard outlined in the WID.

Co-incinerators

Co-incineration relates to the burning of wastes or waste derived fuels with other non-waste (primary) fuels in installations whose primary purpose is the generation of energy or the production of material products. Common examples are cement works and power stations. There can be advantages in recovering the energy from the waste burned provided it does not give rise to unacceptable pollution hazards (and can meet European emission requirements).

These installations will be covered by WID (or HWID). Although BAT for this sector is outlined in the relevant sector guidance, this guidance does include information relevant to the emission limit and other operational standards that must be met, (see Sections 2 and 3).

INTRODUCTION		TEC	TECHNIQUES			SSIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

1.8 Economic Aspects

Waste disposal is an essential service. Disposal costs are eventually, largely borne by the consumer. Options for disposing of waste are limited, and it is with these other options, primarily landfill, that the incineration sector competes.

Legislative pressures on incineration and other disposal methods presently play a key role in predicting the viability of incineration as a disposal option. The Landfill Directive in particular, sets challenging targets for the reduction of reliance upon landfill in the UK. Current UK waste strategy requires significant increases in recycling performance in the municipal sector. Other factors, specific to each sector are outlined below. Increases in the cost of waste disposal can enable individual incineration sectors to provide for enhanced environmental performance and may facilitate the development of otherwise uneconomic disposal techniques.

In general there are economies of scale to be gained at larger throughput plant. This and the greater mass emissions generated may justify the application of tighter environmental standards at the larger plant in each sector. However, in many cases plant throughput will be limited by factors such as: land availability; a need to comply with proximity requirements; and transport costs.

At the small plant scale it is possible that the minimum operational standards required by European legislation (Ref. 25) will render some existing plants uneconomic, unless disposal costs increase further. Larger existing plant requiring upgrade to meet the new standards may also incur significant expenditure. Current trends indicate that disposal costs in many sectors are likely to continue to increase significantly ahead of inflation.

1.8.1 Sub-sector specific information

Municipal waste incineration sub-sector

There are currently 12 operational MWIs in the UK. Waste growth of 3% per annum has been seen in recent years and predictions are that this is set to continue for at least the next decade. Even after allowances have been made for the increased recycling rates required by Government, the move from landfill needed to fulfil the requirements of the Landfill Directive is widely predicted to result in capacity growth in this sub-sector.

Estimates of the number of new municipal incinerators required in the UK are greatly dependent upon plant size and recycling levels achieved. Estimates (Ref. 27) based upon incineration plant capacity of 200 Kte/annum and a recycling rate of 33%, are that 8 to 26 million tonnes of capacity would be required. This equates to between 40 and 130 new plants.

All plants are required to meet the standards of existing European Directives. New plants must meet the standards of the most recent Waste Incineration Directive (2000/76/EC). Existing must meet these standards not later than 28 December 2005. Although not common, in some cases BAT may require higher environmental performance standards than the minimum standards set in the Directives. This is most likely to be the case at large plant, where economies of scale may be evident, or those situated in areas where local environmental conditions dictate additional controls are required to further reduce installation air emission contributions to background pollution.

Planning requirements are in the majority of cases likely to restrict the development of very large throughput (>500 Kte/annum approx.) MWIs, because the Operator is required to demonstrate compliance with policies relating to the proximity of the plant to the waste produced. New plants (which meet WID standards) of throughput below 100 Kte/annum are being introduced in the UK. Even smaller plant may be considered to be economical for selected or pre-treated waste streams.

Gate fees for municipal waste disposal are paid by local authorities. The size of the local authority waste management budget and the contracts agreed with disposal service providers will therefore have an important influence upon the development of both recycling and disposal outlets within a region. Local and regional waste disposal plans have a key role in the development of new facilities.

Financial support for energy from waste projects has been obtained by some projects under the Government's non fossil fuels / renewable energy programmes. It is important that the maximisation of electricity production should not be achieved at the expense of the exploitation of realistic opportunities for overall increased recovery of heat (e.g. through use of combined heat and power, district heating), or without consideration of opportunities to further reduce (mainly NOx) emissions through the use of selected catalytic reduction.

Evidence suggests that the developers of new municipal waste incinerators face significant costs (up to 7 figures) in respect of the preparation and submission of planning applications and pollution control licences.

INTRODUCTION			TECHNIQUES			EMIS	SSIONS	IMPACT	
IPPC and BAT	Making an application	Inst co	allations overed	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Developers of MWIs tend to be part of large utility companies with diverse interests. Cost sensitivity to individual environmental improvements may therefore limited. Investment capital is available in the sector but this may, at least in part, need to be raised against wider company assets.

Clinical waste sub-sector

There is currently considered to be incineration capacity in excess of demand. It is estimated (Ref. 28) that in the period 2003/2006 there will be 30 CWIs, providing an approximate disposal capacity of 175 Kte/annum. Factors that may influence capacity include the use of alternative treatments such as sterilisation, micro-waving and increased segregation of waste at source. The Landfill Directive may result in an increase in clinical waste requiring disposal, but its impact on the UK incineration market as a whole is expected to be limited (see below).

The WID may lead to the closure of a number of smaller CWIs (smaller part Bs) and improve the economics of the remaining, compliant processes. The upgrading of some CWIs to meet the standards of the HWID in 2000, (in order to allow them to incinerate small quantities of certain hazardous wastes) is likely to mean that these processes will have the least difficulty in meeting the standards of the WID. Furthermore, it is possible that these HWID authorised plants may be in a good position to exploit any opportunities arising from the banning of the landfill of certain wastes under the Landfill Directive.

There is some sensitivity to costs and improvements beyond Directive standards may be difficult to justify. The anticipated closure of some plants is evidence of this.

Chemical (hazardous) waste sub-sector

There are currently only three merchant ChWIs in the UK following the closure of a fourth owing to a reported lack of buoyancy on the market. Competition within the sector, with the co-incineration of liquid wastes in cement kilns and the need for compliance with HWID may be responsible.

No new capacity is currently anticipated in this sector. There may be opportunities for increased plant utilisation as the Landfill Directive takes effect and the landfill of certain hazardous wastes is banned...

Although this sub-sector does not appear set for significant expansion, it is noted that the parent companies may operate on a much wider footing and have continued to be capable of raising capital to finance significant projects in similar fields.

In-house ChWIs may be operated for economic or policy reasons. Unit operational costs may be in excess of those found on the open market, but in-house disposal remains attractive where security or reliability of an available disposal route are factors. For waste gases, export may be impractical and in-house disposal will remain the favoured option. The provisions of the WID do not apply to the incineration of waste gases although BAT will apply.

The Landfill Directive may result in increased disposal costs for some waste streams and result in the development of in house waste treatment facilities, including incineration plant. In-house plants often represent a relatively small investment in relation to the overall facility they serve. These plants are therefore relatively insensitive to the cost of environmental improvements.

Sewage sludge sub-sector

The majority of new sewage sludge incinerators built in recent years provide an outlet for the disposal of sewage sludge diverted from dumping at sea following the 1998 ban, and for additional sludge arising from the wider use of secondary treatments installed to comply with the Urban Waste Water Directive. Some water companies have elected for alternative means of disposal including:

- Sludge pelletisation for use as fuel
- Sludge treatment for use as fertiliser
- Other thermal waste treatments e.g. Gasification or pyrolysis

The choice of disposal option will be based on a variety of factors including cost and availability of alternatives. This is influenced by:

- Proximity to and availability of suitable agricultural land for use as fertiliser / conditioner
- Sludge quality compared to land spreading criteria (influenced by industrial effluents and legislation)
- Ability to gain permission for incineration or alternatives

Incineration capacity increased four fold between 1995 and 1998 (6 new plants commissioned with a combined capacity of 344,600 tonnes dry solid per annum). Whether sufficient capacity exists at present or expansion is required will depend upon continued availability and economics of alternative disposal routes.

INTRODUCTION		TEC	TECHNIQUES			SSIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Although water companies have come under increasing pressure from their fiscal Regulator in respect of cost reductions, where compliance with legislative obligations is required they will be required to make the necessary investment and meet these obligations. The disposal of sludges by incineration will therefore be expected to meet the standards required by WID.

Animal carcass and animal remains sub-sectors

Between 1995 and 2000 the number of ACIs in England and Wales increased from 1 to 6. The key driver for this expansion has been the response to BSE. Capacity of ACIs was reported to be 4000 carcasses per week in the year 2000. With culled cattle arising from the over thirty months scheme (OTMS) numbering in excess of 10,000 per week, stock piles of rendered meat and bone meal (MBM), specified risk material (SRM) and insufficient overall alternative disposal capacity, the number of plants in this sector looks set to rise.

The cost of cattle disposal will be directly influenced by the capital and revenues cost of the disposal option chosen. In the incineration sector reduced emission limit values have a great influence over operational costs and hence gate disposal fees.

With the introduction of the WID (and other potential legislation influencing animal carcass only incinerators) the common emission limits concentrations and other requirements may serve to improve the economics of relatively larger plant. Smaller (part B) plant may need to upgrade their throughput to increase revenues in order to offset the application of WID.

Existing installations that are exempted from WID (animal carcass ONLY incinerators) must justify why they cannot meet WID standards. This justification is less likely to be acceptable for new installations. Particular attention should be paid to proposed legislative changes that may eventually harmonise incineration standards for ACIs. The eventual need to meet these standards is likely to mean that new plant will find it cheaper to design for them from day one rather than undertake expensive and technologically difficult upgrading at a later date.

Drum incineration sub-sector

The application of HWID has lead to a dramatic decline in the number of plants operating in this sector. It would appear unlikely that there will be growth in this sector.

Pyrolysis and gasification sub-sector

Commercial scale plants have been developed in the UK for the treatment of selected waste types that fall within a reliable physical and chemical specification e.g. some sewage sludges and wood wastes. Operators are developing plants for thermal treatment of municipal wastes. The majority of these plants are of relatively smaller scale.

In general it would appear that it is at the smaller end of the market (under approx. 100Kte pa) that costs of incineration (per unit disposed) increase and may converge with those of pyrolysis and gasification. However, the fact that relatively few proposals have yet reached full commercial scale exploitation means that the available cost data is sparse, and may be based on technology suppliers estimates rather than realised costs of operating plants.

Pyrolysis and gasification may benefit from potentially greater energy recovery revenues (if high efficiency generation can be used and guarantee the required emission limit values) and provide niche solutions for certain waste types. However, there remains a need for technology providers to develop and demonstrate UK operational plant to provide consistently reported data upon which conclusions regarding economics and environmental performance can be judged, and compliance with WID requirements demonstrated. Until this data is available, significant market penetration may be delayed owing to the additional risk perceived by financiers and purchasers.

Refuse derived fuel and co-incineration sub-sectors

Generalisations are difficult for these sectors, but it is possible that the waste hierarchy will encourage the recovery of energy from wastes that would otherwise be disposed of. Increased waste segregation at source may reveal additional opportunities for the economic recovery of energy from individual waste fractions.

At the sub-sector level, economics will be determined by the specific type of fuel, its availability and the type of plant required to secure its incineration and compliance with the WID (or HWID) standards. In particular the need for pre-treatment stages may result in significant additional costs. Where pre-treatment has already been carried out, costs (capital and revenue) will be reduced, but this can be balanced by a relatively more expensive price for the pre-prepared fuel.

INTRODUCTION		TEC	TECHNIQUES			SSIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

WID provides mandatory environmental performance criteria for all installation in these sectors. Only at installations that have very large waste throughputs (and consequent potential impacts) is it possible that improvements beyond WID may be achieved within BAT. Further improvements may also be appropriate at installations where the process or abatement design in place for the primary fuel is such that its operation with a refuse derived fuel may already allow for standards better than the directive standards to be achieved.

Individual Situations:

- Municipal RDF some plants burning RDF are undergoing upgrade at the time of writing to meet new emission standards. The fact that these have been financed would appear to indicate that sector economics are adequate. The role of recycling credits and energy generation subsidies may be significant in supporting such schemes.
- 2. **Poultry litter** whilst a number of plant upgrades have taken place, there have been few new plants developed in this sub-sector in recent years.
- 3. **MBM Combustion** significant new projects have been financed. The quantity of MBM requiring disposal may lead to further proposals (this would also be true of animal remains and carcass incinerators)
- 4. Tyre Combustion tyres offer a significant disposal problem but at an average CV of 32Mj/kg represent a significant energy recovery opportunity. Preparing the tyres to allow incineration has proved problematic in some instances. Opportunities for co-incineration in power stations and cement kilns offer a means of recovering energy and may even reduce emissions at those installations compared with the primary fuel. The relatively low level of exploitation of these opportunities may reflect technical difficulties, problems gaining permission or the public relations aspects of waste burning. Pyrolysis and gasification may be suited to this sector as they may be able to capitalise on the high volatile content of the tyres to produce a fuel rich off gas.
- 5. Secondary Liquid Fuels the relatively widespread use of SLF and compliance with the existing HWID would indicate that the use of SLF remains economically viable.

1.8.2 Generic cost information required

Capital, revenue and annualised cost data for a variety of abatement and monitoring techniques are provided in the BAT report - Review of IPC Technical Guidance for Chapter 5 and 1.3c Vol1 March 2000 (Ref. 29).

Operators will be expected to justify their selected techniques using the H1 methodology (Ref. 5) (E1 for IPC) – thus comparing the costs of different techniques against the environmental gains. The key cost related aspects outlined below are of particular interest in this sector, and will require justification in the application. This section is intended as a guide. It should not be considered to be all encompassing.

Where it is not possible for uncertain (but potentially significant) factors such as "reliability" or "operational experience" to be quantified, these may still be included in any accompanying qualitative discussion that justifies the final BAT selection. In many cases such factors will be the dominant indicators of environmental performance:

1. Waste pre-treatment:

Waste is often of a highly heterogeneous nature. This can create problems during its incineration including short-term exceedences of emissions limits and unnecessary waste production through the over-dosing with reagents to control peak emission rates.

The degree of pre-incineration waste treatment will therefore require justification and should take into account the following:

- What are the benefits to be accrued from pre-treatment?
- Additional capital and revenue costs of the pre-treatment options?
- Costs and consequences associated with not implementing pre-treatment? e.g. waste disposal costs for additional reagent waste arising?

See also Section 2.3.1 (incoming waste management) and section 2.3.8 (abatement of point source emissions to air).

2. NOx control techniques:

There are a variety of approaches available for the prevention and control of NOx, (see Sections 2.3.3 - furnace types, 2.3.4 - furnace requirements and Section 2.3.8 - abatement for details).

INTRODUCTION		TE	TECHNIQUES			SSIONS	IMPACT	
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Operators will in particular be required to provide cost arguments comparing the NOx control techniques selected and take into account the following:

- Could waste pre-treatment reduce NOx production?
- Could a lower NOx furnace be used? e.g. fluidised bed
- · Could staged combustion be used?
- Can FGR be used to reduce the need for reagent injection?
- Could SCR be used instead of or as well as SNCR?

Note: Recent cost comparisons for SCR and SNCR appear to have shown some conversion. Figures quoted in a recent application (year 2001) of £1,181/te NOx avoided with SNCR and £3,513/te of NOx avoided with SCR. Further reductions in SCR cost could lead to it becoming BAT in some situations and its possible use should be examined (see Section 2.3.8 for further comment on secondary NOx abatement and NOx cost benefit study requirements).

3. Acid Gas Treatment:

There are a variety of approaches to acid gas control. The techniques are outlined in Section 2.3.8 - abatement of point source emissions to air.

The key cost related elements that must be considered in order to demonstrate that the selection chosen is the BAT are:

- · Can waste elimination / pre-treatment be used to control inputs?
- What are the costs and benefits of waste pre-treatment? e.g. reduction in reagent and APC residue disposal costs? Capital and revenue costs of pre-treatment plant?
- Can reagent dosing be linked to acid gas load using fast response monitoring and adjustment?
- Would an alternative reagent assist?
- · Justify why wet scrubbing is not employed?

4. Stack height

Stack height increases will result in reduced ground level impacts but there are restrictions on this.

· Could the stack height be increased to reduce ground level impacts?

See Section 2.3.5 - Chimneys and vents, and Section 4 - Impact.

5. Energy recovery

The primary purpose of the installations in this sector is the safe disposal of waste. The recovery of energy from the waste is an important, but secondary consideration. Nonetheless, installations should be designed and operated such that *"the heat generatedis recovered as far as practicable"* (ref WID). See also Section 2.3.7 - Boiler design and Section 2.7 - Energy.

Operators must therefore justify how the degree of energy recovered is the BAT. This will require consideration of:

- How much energy is produced per tonnes of waste incinerated? What is the "net" energy
 production / consumption (i.e. less parasitic loads) of the installation per unit of waste incinerated?
- Will increased energy recovery result in increased emissions from the installation?
- · What are the costs (capex / opex) of the options for increasing the energy recovered?
- Justify why combined heat and power and / or district heating cannot be used?
- Can waste heat be used to heat primary / secondary air? Or to heat the final gas discharge (to reduce plume visibility, improve dispersion, allow for SCR?)

INTRODUCTION TECHNIQUES			E	EMISSIONS			IMPACT			
Management	Materials inputs	Main activities	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

	2 TECHNIQUES FOR POLLUTION CONTROL
BAT Boxes to	This section summarises, in the outlined BAT boxes,
help in	what is required in the application
preparing applications	 the indicative BAT requirements (i.e. what is BAT in most circumstances) against which the application will be judged.
	At the top of each BAT box is the question which is being addressed. It will be seen that these deal with the questions in the Application Form relating to environmental performance of the installation.
	Although referred to as "BAT", the requirements also cover the other requirements of the PPC Regulations and requirements of other Regulations (such as the Waste Management Licensing Regulations (see Appendix 2 for equivalent legislation in Scotland and Northern Ireland) and the Groundwater Regulations insofar as they are relevant to an IPPC Permit).
Indicative BAT requirements	Where it has been possible for the Regulator to make a judgement on what will normally be BAT, the, indicative requirements are clear and prescriptive. In such cases:
	• If you propose to comply with the indicative requirement, you need only describe how you will do so, if this is not obvious from the wording of the requirement itself.
	• If you propose to depart from any indicative requirements, you should justify your proposal. Such departures may be stricter or less strict than the indicative requirements:
	Stricter proposals may be appropriate where:
	- new techniques have become available after the publication of the guidance;
	- the particular technical configuration at your installation makes higher standards practicable;
	 The local environment is particularly sensitive. Less strict proposals may be justified due to particular factors relating to your installation or the
	local environment. For example, you may operate to a standard that is very close to an indicative requirement, but using different plant or processes from that upon which the indicative requirement is based. In such a case it may impose a disproportionate cost to replace the old plant with the new techniques for only a small decrease in emissions.
	In other cases, the main BAT candidates are identified, but the final choice can only be made on an installation-specific basis. In further cases, aspects of the installation may not be covered by the guidance at all.
lug tift sin a	Whether you are:
Justifying proposals	 justifying departures from clear indicative requirements;
p. op come	 assessing options to determine which of those identified by guidance is best for a your site; or developing proposals for parts (or possibly all) of an installation that are not covered by guidance.
	The costs and benefits of a range of options should be compared. However, the level of detail required depends on the environmental significance of the matter in question. In the more complex cases (e.g., where the options available would lead to significantly different environmental effects, or where the cost implications are a major factor) it will be necessary to develop proposals through a more detailed analysis of the costs and benefits of options. The Agency's methodology for such assessments is set out in the IPPC H1 "Assessment of BAT and Environmental Impact for IPPC (in preparation).
	In many situations, however, it will not be necessary to carry out a detailed analysis of options. For example, where an indicative standard is inappropriate for obvious technical reasons, or where there are only minor additional emissions, it may be possible to justify a departure in just a few words.
Prevention is the priority.	 In responding to the requirements the Operator should keep the following general principles in mind. As a first principle there should be evidence in the application that full consideration has been given to the possibility of PREVENTING the release of harmful substances, e.g. by:
	- substituting materials or processes (see Section 2.2.1);
	 preventing releases of water altogether (see Section 2.2.3); or by
	- preventing waste emissions by reuse or recovery.
	 Only where that is not practicable should the second principle be adopted of reducing emissions which may cause harm.

Further explanation of the requirements of Section 2 is given in Section B2 of the Guide for Applicants.

INTRODU	I TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.1 Management Techniques

Within IPPC, an effective system of management is a key technique for ensuring that all appropriate pollution prevention and control techniques are delivered reliably and on an integrated basis. The Regulators strongly support the operation of environmental management systems (EMSs). An Operator with such a system will find it easier to complete not only this section but also the technical/regulatory requirements in the following sections.

The Regulators recommend that the ISO 14001 standard is used as the basis for an environmental management system. Certification to this standard and/or registration under EMAS (EC Eco Management and Audit Scheme) (OJ L168, 10.7.93) are also strongly supported by the Regulator. Both certification and registration provide independent verification that the EMS conforms to an assessable standard. EMAS now incorporates ISO 14001 as the specification for the EMS element. For further details about ISO 14001 and EMAS contact British Standards Institute (BSI) and the Institute of Environmental Management and Assessment (IEMA) respectively.

The steps required in this and subsequent sections may help the Operator to make good any shortfalls in their management system. An effective EMS will help the Operator to maintain compliance with regulatory requirements and to manage other significant environmental impacts. While the requirements below are considered to be BAT for IPPC, they are the same techniques as required in a formal EMS and are also capable of delivering wider environmental benefits. However, it is information on their applicability to IPPC which is primarily required in this application.



1. Describe their management system to demonstrate how it meets the *"Requirements for an effective management system"* below. The description should make clear who holds responsibility for each of the requirements. The second column explains where in the application the response to each requirement is best dealt with to avoid duplication. Copies of all procedures are not needed, but examples may be included in your application.

If you are certified to ISO 14001 or registered under EMAS (or both), you may provide a statement derived from certification records/assessments to support your application.

Further specific management procedures are dealt with under the appropriate section on the remainder of the document. It is recommended that you understand all the requirements of the application before completing this section, as many management issues are dealt with in other sections.

2. The type of management system employed will depend upon the scale and complexity of the operations undertaken. The Operator should demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements below, or by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures.

Indicative BAT Requirements

The Operator should have a management system in place for the activities which delivers the requirements given in column 1 below. The development of any aspects of the management system not already in place should be completed within the timescale given in Section 1.1.

Ree	quirement for an effective management system	How delivered for IPPC
1.	Clear management structure and allocated responsibilities for environmental performance, in particular meeting the aspects of the IPPC Permit	Describe in this section who has allocated responsibilities
2.	Identification, assessment and management of significant environmental impacts	By responding to the requirements in Section 4.1 in the application
3.	Compliance with legal and other requirements applicable to activities impacting on the environment	Compliance with the Permit satisfies this requirement

Cont.

BAT for management techniques

INTRODUCTI	ON TECHNIQUES EMISSION	IS IMPACT
Management Materia	Is Activities/ Ground Waste Energy Accidents Noi	se Monitoring Closure Installation issues
4. 5.	Establishing an environmental policy and setting objectives and targets to prevent pollution, meet legal requirements and continually improve environmental performance Environmental improvement programmes to implement policy objectives and targets	The Applicant should make proposals in response to each of Sections 2.2 to 2.12. These proposals may be incorporated within the Permit improvement programme
6.	Establish operational controls to prevent and minimise significant environmental impacts	By responding to the requirements in Sections 2.2 to 2.7, 2.11 and 2.12 in the application
7.	Preventative maintenance programmes for relevant plant and equipment – method of recording and reviews	Describe system here. List procedures in Section 2.3
8.	Emergency planning and accident prevention	By responding to the requirements in Section 2.8 in the application
9.	Monitoring and measuring performance	Describe in this Section
	Identify key indicators of environmental performance and establish and maintain a programme to measure and monitor indicators to enable review and improvement of performance	
10.	Monitoring and control systems:	By responding to the requirements
	 to ensure that the installation functions as intended; 	In Section 2.10 in the application
	 to detect faults and unintended operations; 	
	to detect slow changes in plant performance to trigger preventative maintenance	
11.	Training	To be described in this Section
	Provision of adequate procedures and training for all relevant staff (including contractors and those purchasing equipment and materials), which should include:	the areas covered by Sections 2.2 to 2.3 and 2.5 to 2.10 are covered
	 a clear statement of the skills and competencies required for each job; 	
	 awareness of the regulatory implications of the Permit for the activity and their work activities; 	
	 awareness of all potential environmental effects from operation under normal and abnormal circumstances; 	
	 prevention of accidental emissions and action to be taken when accidental emissions occur; 	
	• implementation and maintenance of training records; Expertise required depends on the activities being carried out. However, both technical and managerial staff upon whom the installation's compliance depends need sufficient qualifications, training and experience for their roles. This may be assessed against any industry sector standards or codes of practice	
12.	Communication and reporting of incidents of actual or potential non-compliance and complaints	Describe in this Section
	Actions taken in response, and about proposed changes to operations	
13.	Auditing	Describe in this Section
	Regular, (preferably) independent, audits to check that all activities are being carried out in conformity with these requirements. All of these requirements should be audited at least once per year	

Cont.

INTRODUCTION TECHNIQUES			EMISSIONS			IMPACT				
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

14.	Corrective action to analyse faults and prevent recurrence	Describe in this Section how this is dealt with for each of Sections
	Define responsibility and authority for handling and investigating non-conformance, taking action to mitigate any impacts caused and for initiating and completing corrective and preventive action	appropriate
	Recording, investigating, taking corrective action and preventing recurrence, in response to environmental complaints and incidents	
15.	Reviewing and Reporting Environmental Performance	
	Senior management review environmental performance and ensure appropriate action taken where necessary to ensure that policy commitments are met and that policy remains relevant. Review progress of the Management Programmes at least annually.	Describe in this Section
	Incorporate environmental issues in all other relevant aspects of the business, insofar as they are required by IPPC, in particular:	Describe in this Section
	the control of process change on the installation;	
	 design and review of new facilities, engineering and other capital projects; 	
	capital approval;	
	the allocation of resources;	
	 planning and scheduling; 	
	 incorporation of environmental aspects into normal operating procedures; 	
	 purchasing policy; 	
	 accounting for environmental costs against the process involved rather than as overheads. 	This will become a Permit requirement
	Report on environmental performance, based on the	Describe in this Section
	audit cycle), for:	Describe in this Section
	 information required by the Regulator; and 	
	 effectiveness of the management system against objectives and targets, and future planned improvements. 	
	Report externally preferably via public environmental statement	
16.	Managing documentation and records	
	List the core elements of the EMS (policies, responsibilities, procedures etc) and links to related documentation in order to be able to control, locate and update documentation.	Describe in this Section
	Describe how environmental records and results of audits and reviews are identified, maintained and stored	

INTRODUCTION TECHNIQUES			EMISSIONS			IMPACT				
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Selection of raw materials

2.2 Materials Inputs

This section covers the use of raw materials, water and waste input and the relevant techniques for both minimising their use and minimising their impact by selection. It also provides a summary (Section 2.2.2 - Waste minimisation) of the main techniques that should be considered for minimising the waste produced by the installation. The choice of fuels is covered under Section 2.7.3 - Energy.

As a general principle, the Operator will need to demonstrate the measures taken to:

- reduce the use of chemicals and other materials (Section 2.2.2);
- substitute less harmful materials or those which can be more readily abated and when abated lead to substances which in themselves are more readily dealt with;
- understand the fate of by-products and contaminants and their environmental impact (Section 4).

Special considerations for the waste incineration sector

Unlike other sectors, where the production of a material product by the consumption of the minimum quantity of raw materials is the aim, the waste incineration sector has the primary purpose of consuming the waste that is fed to it, for the purpose of its safe disposal.

Whilst it remains valid to consider reducing consumption of other raw materials (e.g. reagents used for pollution control purposes and water) in the same way as other sectors, the approach in respect of waste that will actually be disposed of in the installation is necessarily different. This is because the quantity and type of waste to be disposed of, and hence the capacity of, the installation will be selected according to the amount of waste of a particular type that requires disposal. At the installation level, there will therefore be limited scope to consider how the overall quantity of waste might be reducedthis is a wider consideration, beyond the scope of an individual permit. Such wider, strategic factors are however important when sizing the installation, and it will remain for the Operator to demonstrate how the installation fits within local, regional and national waste strategy and plans.

Despite the comments made in the paragraph above, from a pollution control perspective at the sector level, it remains valid to carefully consider the composition of the waste. The collection, selection and treatment chain will determine the type and composition of waste that will finally be incinerated. This in turn will determine the techniques that will represent the BAT at the incinerator. The influence of waste type on plant design is discussed in Section 2.2.1.2.

2.2.1 Raw materials selection

This section looks at the general issues concerning the selection and substitution of raw materials (Section 2.2.1.1) and wastes (Section 2.2.1.2). Section 2.2.2 describes the general techniques to minimise the use of raw materials.

2.2.1.1 Reagent selection

The principal raw materials that may be consumed (excluding waste feed) in the incineration sector are.

- Lime Calcium hydroxide (Ca(OH)2) reagent for gas treatment;
- Sodium Bicarbonate (NaHCO₃) - reagent for gas treatment;
- Sodium Hydroxide (NaOH)
- reagent for gas treatment; Water - make up for neutralisation reagents / boiler water / cooling towers;
- reagent for NOx reduction; Urea or Ammonia
- Activated Carbon - reagent for dioxin / heavy metal absorption;
- Catalysts - where SCR is used; or in catalytic bag filters
- Water treatment chemicals for boiler water conditioning;
- mainly acids and alkalis for pH balancing and precipitation: Effluent treatment plant chemicals
- Fuels - either gas or fuel oil for start up and temperature stabilisation;
- Biocides - to reduce fouling in direct cooling systems and for biological safety in cooling water.

Summary of materials in use

Reduce Substitute Understand

INTRODUC			HNIQU	JES	EM	IISSION	IS	I	MPAC	Т
Management Ma	aterials	Activities/	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation
	nputs	abatement	water							133063
Selection of raw materials	Applio Ques	cation Form tion 2.2 (part ⁻	1)	ldentify water t	/ the raw hat you p	and auxilia propose to u	ry mate use.	rials, other s	substanc	es and
	With	the Applic	ation the	e Opera	ator she	ould:				
	1. s ir	upply a list of	the materia	als used,	which ha	ve the poter	ntial for s	ignificant env	vironmen	tal impact,
	•	the chemica	al composi	tion of th	e materia	Is where rele	evant;			
	•	the quantities used,								
	•	the fate of t	he materia	ll (i.e. app whore kr	proximate	percentage	s to eacl	h media and	to the pro	oduct), tovioitu to
	•	relevant sp	ecies).	where Ki	iown (e.g		ity, Dioac	cumulation	Jolential,	
	•	any reason impact inclu (the substit	ably practi uding, but r ution princ	cable alte not be lim iple).	ernative ra hited to, a	aw materials ny alternativ	which n es descr	nay have a lo ibed in BAT	wer envir Requiren	onmental nent 5 below
	Generic information about materials, and grouping information of those of a similar type, is normally adequate rather than listing every commercial alternative used. A common sense approach to the level of detail should be used; ensuring that any material could have a signific effect of the environment is included. Product data sheets should be available on-site.								be, is sense a significant	
	 justify (e.g. on the basis of impact on product quality), the continued use of any substance which there is a less hazardous alternative and that the proposed raw material section is BAT: 							ance for is therefore		
	3. fo c	or existing acti ertain substar	ivities, ider nces, which	ntify short n the Ope	falls in the erator beli	e above info eves require	rmation, longer t	e.g. the envi term studies	ironmenta to establi	al impact of sh.
	Indic	ative BAT	Require	ments						
BAT for	1. T	he Operator s	should:							
selection	•	complete a	ny longer-t	erm stud	ies (Item	3 above),				
	• a	carry out ar s improvemer	ny substitu nt conditior	tions ider ns to a tin	ntified, nescale to	be approve	ed by the	Regulator.		
	2. T	he Operator s	should mai	ntain a de	etailed inv	ventory of ra	w materi	als used on-	site.	
	3. T m	he Operator s naterials and t	should have he implem	e procedi entation	ures for th of any sui	ne regular re table ones v	view of r	new developr e less hazard	ments in ı ous.	raw
	4. T n	he Operator s naterials.	should have	e quality	assuranc	e procedure:	s for the	control of the	e content	of raw
	 The following raw material substitution criteria should be applied where appropriate: Raw material Selection criteria 									
	Alkal	ine reagents		• low	concentra	ations of per	sistent p	ollutants in th	ne reager	nt itself e.g.
				 high 	ais i pollutani	t absorption	efficienc	y is required		
				 low was 	waste pro	oduction i.e.	low cond	centrations o	funused	reagent in
				• pos	sibility to	recycle to de	ecrease	waste produc	ction	
	Activ	ated Carbon		• low	concentra	ations of per	sistent p	ollutants e.g.	metals	
				high	porosity	to enhance	absorpti	on efficiency		od as
				 care abs 	e required	aracteristics	ying sup s may ch	piler / source lange	is require	eu as

Cont.

Only "low mercury" NaOH should be used. (#)

•

NaOH

INTRODUCTION TECHNIQUES			EMISSIONS			IMPACT				
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Selection of raw materials

Raw material	Selection criteria
Support Fuels	• Support fuels shall not give rise to higher emissions than burning gas oil, liquified gas or natural gas.
	 All uses of support fuel other than natural gas will require justification.
Dispersants/ surfactants	Only chemicals with high biodegradability and known degradation products should be used.
	Alkylphenolethoxylates should be avoided. (#)
Biocides	Only chemicals with high biodegradability and known degradation products should be used.
	Environmental assessment should consider site specific nature of receiving waters before deciding on material suitability e.g. saline or freshwater environments

2.2.1.2 Waste selection - Influence of waste type on plant design

This section deals with the influence that the composition of **the waste that is fed to** the incineration installation has upon BAT. Some data for waste compositions is included in Section 2.2.4.

Understanding the physical and chemical nature of the waste stream is of paramount importance. BAT for an incineration installation will depend to a large extent upon the type of wastes that are being incinerated. Without this knowledge it will not be possible for an installation to be designed so as to meet the requirements of BAT.

Applicants must identify and consider:

- All waste streams that are to be incinerated.
- expressed according to the European Waste Catalogue categories;
- according to the premises where the waste arises and the consequent general character of the waste;
- the throughput of each line for each waste type in tonnes per year and per hour,
- The physical nature of the waste:
 - its CV range;
 - its moisture range;
 - its ash content;
 - its density / friability;
- The chemical composition of the waste:
 - Identify the range of pollutant loads that will be generated pre-abatement and show how their emission will be prevented or minimised (e.g. PCBs, Halogens, Metals),
- How the plant design envelope accounts for waste composition variation,
- Particular wastes that may cause difficulties and how they will be prevented from entry to the installation or managed to avoid emission exceedences or other problems,
- How the waste will be managed on-site and upstream of the plant to prevent exceedence of plant design envelope,
- How the nature of the waste may change over the plant lifetime and how this has been taken into account.

The Applicant should demonstrate that their plant has been designed and will be managed and operated such that the heterogeneity of the waste is accounted for. Operational plant will be able to demonstrate this by reference to actual plant data for emissions and other operational parameters. New plant may be able to make reference to the performance of other operational plant of the same design but must consider the possibility of local variations in waste character, plant modifications and management.

INTRODUCTION		TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

Waste	2.2.2 Waste minimisation (minimising the use of raw materials)
minimisation	The prevention and minimisation of waste and emissions to the environment is a general principle of

Principles

IPPC. Operators will be expected to consider the application of waste minimisation techniques so that, wherever practicable, all types of wastes and emissions are prevented or reduced to a minimum. The

steps below will also help to ensure the prudent use of natural resources.

Waste minimisation can be defined simply as:

"a systematic approach to the reduction of waste at source, by understanding and changing processes and activities to prevent and reduce waste".

A variety of techniques can be classified under the general term of waste minimisation and they range from basic housekeeping techniques through statistical measurement techniques, to the application of clean technologies. In the context of waste minimisation and this Guidance, waste relates to the inefficient use of raw materials and other substances at an installation. A consequence of waste minimisation will be the reduction of gaseous, liquid and solid emissions.

Key operational features of waste minimisation will be:

- the ongoing identification and implementation of waste prevention opportunities;
- the active participation and commitment of staff at all levels including, e.g. staff suggestion schemes:
- monitoring of materials usage and reporting against key performance measures.

For the primary inputs to waste activities e.g. the waste to landfill, the requirements of this section may have been met "upstream" of the installation. However, there may still be arisings which are relevant.

See Ref. 7 for detailed information, guides and case studies on waste minimisation techniques.



Using this information, opportunities for improved efficiency, changes in process and waste reduction should be generated and assessed, and an action plan prepared for the implementation of improvements to a time-scale approved by the Regulator.

minimisation
INTRODUC		V TEC	CHNIQUES EMISSIONS IMPACT							
Management In International Ma	iterials iputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Waste minimisation	2. F tr	References pl echniques. S specific to the	rovide deta ection 2.3 main activ	ailed infor covers cl ities in th	rmation, g leaner teo is sector.	guides and ca chnologies ar	ase studi nd waste	es on waste minimisatior	minimisa ı opportu	tion nities
	3. F	eedstock He	eterogenei	ty						
	lı ti e	mproving feed hroughout the environmental	dstock hete installatio performar	erogeneit n. This v nce and r	y can mir vill in turr educe the	nimise residu I lead to impr e amount of r	es by im oved abi eagents	proving oper lity to optimis used and wa	ational st se operat astes proc	ability ional and duced.
	(Operators sho neterogeneity:	uld consid	er at leas	st the follo	owing technic	ues for i	mproving fee	edstock	
		Procedure:	waste man	agement al of pro	t blem was	stes				
		On or off s	ite waste ti	reatment	/ mixing					
	4. F	urnace Cond	ditions							
	T h ti	The prime pur narmfulness of o securing this	pose of inc f the residu s.	ineration les arisin	i is to the	rmally treat w her disposal.	vastes in Good co	order to min ombustion co	imise the onditions	amount and are the key
	•	Dperators sho burnout in the bed / b SNCR read	uld conside the furnace urnout time gent dosine	er at leas e should e and ten g should	at the follo achieve l nperature be optim	owing key teo ess than 3% e exposure) ised to preve	hniques: TOC (e.ز nt ammo	to minimise i g. by improvi nia slip to as	residue p ng waste h	roduction: agitation on
	5. C	Sas Treatmer	nt Conditio	ons	·	·		·		
	(c	Optimising alk contaminated)	aline (and reagent.	other) re	agent use	e will prevent	the prod	luction of wa	stes (unu	sed or
	(•	Dperators sho alkaline rea	uld consid agent recy	er at leas cle	st the follo	owing technic	lues:			
	•	wet scrubb	ing n of reade	nt dosino	and rea	ction conditio	ne			
	6 1	Vaste Manac	ement	in uosing			110			
	U. V	Desertant			4 4h - C 1					
	•	On site or o	off-site was	er at leas	nents to i	mprove suita	bility for	recovery or c	lisposal.	
	N s ti ti	Aixing of wast maller amour hat the plant of heir recovery	es produce ht such tha design app or disposa	ed on site t it canno ropriately I. This n	e can cau ot be reco y segrega nust inclu	se contamina vered or eas ites waste str ide at least:	ation of a ily dispos reams wi	l large amound sed of. Oper thin the plan	nt of was ators <i>mu</i> t, in orde	te with a f st ensure r to facilitate
	Furthe	er guidance or	n each of th	nese tech	niques n	hay be found	in Sectio	on 2.3 - Main	activities	5.
	. artife									-

INTROD	UCTION	J TEC	HNIQU	JES	E١	/IISSION	IS	IN	ЛРАС	Т
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Water use	2.2.3 Inciner tend to and so water of	Water ators are not be a primary me cooling sy consumption y	use generally c environme /stems can will howeve	considere ental con consum er make a	ed to be n cern altho le relative a contribu	najor users o bugh it is imp ly larger qua tion to impro	f water. ortant to ntities of	Water use do recognise th water. The ironmental p	oes not t nat wet so minimisa erformar	herefore crubbing ation of ace of the
Summary of the activities	installa Major • gas • ash • eva Other r	ition, but shou water uses in s scrubbing – n discharge qu aporation from uses include t	uld be cons incineration particularly uench bath wet coolir poiler water	idered o n plants v wet scru s ng towers r make u	n a case are: ubbing s p and wa	by case basi sh down ope	erations.			
	Appli Ques	cation Form tion 2.2 (part	3)	ldent <u>wate</u>	ify the ra <u>r</u> that you	w and auxil u propose to	iary mai o use.	terials, other	r substa	nces and
	With 1. s 2. s 3. d 4. d	the Applic upply informa upply a diagra escribe the cu ny other techn emonstrate th	tion on wat am of the w urrent or pr niques whic nat the prop	e Operater consu- vater circ oposed p ch are pe	ator sh imption a uits with i position w ertinent to e BAT, b	ould: nd comparise ndicative flow ith regard to the installati y confirming o	on with a ws; the indic ion; compliar	ny available cative BAT re nce with the in	benchm quireme ndicative	arks; nts below, or
	5. d J	equirements, l pplicants) or a escribe, in pa lanned. ative BAT	by justifying alternative rticular, an Require	g departi measure y water a ments	ures (as c es; audits alre	lescribed in S eady conduct	Section 1	.2 and in the	Guide fo	or de or
BAT for water efficiency	1. A b o s ⁱ	regular revie een carried of pportunity wit ome time befo PPC Permit re	w of water ut recently, hin the imp pre an audi eviews. Th	use (wat an initia provemer t will be e audit s	ter efficie l comprel nt prograr meaningf hould be	ncy audit) sh nensive audit nme. New p ul. Further a carried out a	ould be o t should lants will udits sho s follows	carried out. V be carried ou need to have ould be at lea s:	Where or it at the e e been o ast as fre	ne has not earliest perating for quent as the
	•	The Operat Water effici where not a (see Sectio and those i reducing wa usually inst	tor should p ency object available, n in 1.2), or v n the existi ater use be allation-spe	broduce stives sho ational b where be ng secto byond a c ecific.	flow diagi ould be es enchmar nchmarks r guidanc certain lev	rams and wa stablished by ks (see Ref. s are not ava æ should be vel should be	ter mass compar 9). In ju- ilable, th taken int identifie	 balances for ison with sec stifying any d e techniques techniques account. T d by each Op 	r the acti etor guida leparture describe The cons perator, a	vities. ance or, s from these ed below traints on as this is
	•	Water pincl chemical pl (see ETBP Using this i	h technique ant, to ider P publication nformation	es should ntify the c ons, Ref. , opportu	d be used opportuni 8). Inities for	in the more ties for maxin reduction in	complex nising re water us	situations, p use and mini se should be g	articularl imising u generate	y on se of water ed and a timescale
	2. T	approved b he following g	y the Regu	ilator.	nould be	applied in se	quence t	o reduce emi	issions to	o water:
	:	water-efficient water shout necessary. which has a	ent techniq ld be recyc Where thi a lower wat	ues shou led withi s is not p ter qualit	uld be use n the prop practicabl y requirer	ed at source cess from wh e, it should b ment;	where p lich it iss e recycle	ossible; ues, by treati ed to another	ing it first part of t	if he process

 in particular, uncontaminated roof and surface water, which cannot be used, should be discharged separately.

Cont.

INTRODUC)TIO	N TEC	HNIQU	JES	E	VISSIO	NS	I	MPAC	Т
Management Ma	terial iputs	S Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Water use	3.	Measures sho (see Section 2	uld be imp .3.12).	lemented	to minin	nise contamir	nation risl	k of process	or surface	e water,
BAT for water efficiency (cont)	4.	To identify the associated with waters, should treatment.	scope for h each use be kept se	substituti should t sparate w	ng water be identifi /here the	from recycle ied. Less co re is scope f	ed source ntaminate or reuse,	s, the water ed water stre possibly afte	quality ree ams, e.g. ar some fo	quirements . cooling orm of
	5.	Ultimately was applications, th be usable in th can vary, it can the quality falls treated water f	stewater wi ne best cor ne process n be recycl s below that rom the ef	Il need so rventiona directly c ed select it which t fluent trea	ome form al effluent or when n ively whe he syster atment pl	n of treatmen t treatment pr nixed with fre en the quality m can tolerat lant could be	t (see Se roduces a sh water is adequ e. The C used and	ction 2.3.10) a good water While trea Jate, revertin Operator shou d justify when	Howeve quality w ted efflue g to disch uld identifi re it is not	er, in many hich may nt quality narge when y where
		In particular, the be applied to in Ultimately, the There remains particularly wh system could be	ne cost of r ndividual p y could cou , however, ere waste pe produce	membran rocess st mpletely a conce heat is av	e and de treams or replace tl ntrated e vailable f	i-ionisation te r to the final e he ETP plant ffluent strear or further trea	echnology effluent fr t, leading n but, wh atment by	/ continues to om the efflue to much red ere this is su y evaporation	o reduce. ent treatm luced efflu ifficiently s n, a zero e	They can lent plant. lent volume. small, and effluent
		The advantage page 78. Ope	es and disa rators shou	advantag uld asses	es of wet s the cos	scrubbing s sts and bene	ystems a fits of pro	re discussed viding such a	in <mark>Sectio</mark> a system.	n 2.3.8.3 on
	6.	Water used invacuumingevaluatingtrigger cont	cleaning a , scraping the scope trols on all	nd washi or moppi for reusir hoses, h	ing down ng in pre ng wash v and lance	should be m ference to he water; es and wash	iinimised osing dov ing equip	by: vn; ment.		
	Ben	chmarks fo	r water	consur	nption					
	7.	Dry scrubbing for ash quench	systems d 1 and cond	o not con itioning.	isume sig	gnificant qua	ntities of v	water, with o	nly a little	required
	8.	Semi dry gas s	scrubbing t	ypically c	onsume	250-350Kg /	tonne of	waste incine	erated.	
	9.	MWIs using we should be redu liquors are rec therefore be me allow for furthe	et scrubbir uced by scr irculated, t uade at the reductior	ng can co rubber liq his does final (pol ıs in wate	nsume u uor re-cil not resul lishing) si er consur	p to 850Kg / rculation. Ca t in higher er crubber. Liq nption than s	tonne of are must nissions t uor treatr ample ble	waste incine be taken to e to air. Clean nent to remo ed and top u	erated, alth ensure that water inp ive polluta up system	nough this at, where out should ants will is.
	10.	The nature of 1100 Kg/tonne primary object circulating the final stage clea exhaust gases plume dispersi	the wastes of waste) ive. Multi- used stack an scrubbe but consi ion charact	treated i may be j stage we end scru r water h deration eristics.	n HWIs r ustified to t scrubbin ubber wa elps to po should al	neans that h o ensure emi ng systems p iter to earlier revent scrub lso be given	igher leve issions to provide fo scrubbin ber water to stack e	els of water c air are conti r lower wate g / quench s · losses by e exit temperat	consumpti rolled – w r consum tages. Co vaporation ures and	on (up to hich is the ption by re- ooling of the n with consequent
	11.	Most CWIs err	ploy dry s	crubbing	and there	efore consun	ne relative	ely little wate	؛r.	
	12.	There is little c waste feed (e. techniques tha closed loop typ	lata availal g. drum inc it can be as be.	ble for oth cineration ssociated	ner incine ı) the gre I with hig	eration plant ater the justi her levels of	types. In fication fo water co	general the or the use of nsumption if	more vari the wet so they are r	iable the crubbing not of the
	13.	In justifying an taken into accordidentified by eabeing used for In general the use is similar a	y departur ount. The ach Operat gas scrub following n and closed	es from t constrain tor, as thi bing it wil nay repre loop wet	hese ber its on red is is usua Il be impo sent BAT systems	nchmarks the lucing water illy installatio ortant that Op I with regard are not prac	e techniqu use beyo n-specific perators j l to water :ticable):	ues described and a certain b. With the n ustify their cl consumption	d below sl level shou najority of hoice of te n (provide	hould be uld be fresh water chnique. d reagent
		IVIVIS - (HWIS -)	wet scrubb	i-ury scru ing;	g;וועטי					
		• CWIs - (dry or semi	i-dry scru	bbing.					Cont

INTRODUC	TION	TEC	HNIQL	JES	EN	/IISSIO	IS	I	MPAC	Т
Management Ma	iterials iputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
	_									
Water use BAT for water efficiency (cont)	14. Oth just con nee was of v the was poll scru	er incinerato ify why lowe sumption). d to ensure ste produced vater, althou re is sufficien ste pre-treati utant load ito ubbing syste	brs should or water cc The natur emissions d from the gh closed nt space a ment (to ir ems) shou ems.	be asse onsumption to of the way to air ar gas treat loop effli nd wet p nprove h ild also b	ssed on a on technic waste fee re controll tment are uent recy lumes are eterogen re conside	a case by cas ques cannot d in terms of ed within em all factors th cle systems e not an issu eity) or feed ered as these	se basis be used its comp hission lin hat may j may med e. Other manage e may re	and the Ope (in order to position and I mit values an ustify the use at all of these r prevention ment (to rem duce the nee	rator requiminimise heteroger ad the quate of greate criteria poptions si ove or dill ed for wate	uired to water neity, the antities of ter quantities providing uch as lute high ter
	Other 3	Techniqu	es for re	ducin	g <mark>gro</mark> ss	water us	е			
	15. Oth	er technique	es include	:						
	•	In wet syste	ems - prov	ision of	multi-stag	e scrubbers	in series	8:		
		- with the	effluents	from the	clean scr	ubbers used	l as feed	for the dirty	scrubber	/ quench;
		- clean wa	ater feeds	to final p	olishing /	clean end so	crubbers	;		
		- dirty wat	er bleeds	only or p	orimarily fi	rom dirty end	l scrubbi	ng / quench	stages;	
		- consider	the possi	bility of s	crubber I	iquor treatme	ent and r	e-circulation		
	•	In semi-dry compromisin emission lim consumed, i difficulties, c	r systems ng the abi nit values. it should n or poor rea	the quar lity of the Whilst the ot result agent rea	ntity of wa abateme ne use of in reduct ction con	ater should b ent plant to e BAT will ent ions that cou ditions (e.g.	e measu ffectively ail demo Ild result moisture	red and mini treat stack on nstration that in reagent h the temperatur	imised, bu gases and it water is andling (p e or conta p atroom	ut without d meet not over oumping) act time).
	·	concentration consumption through the changes) - t concentration systems.	n of wet an ons (fast re n of both v alteration his will rec on change	vater and of alkalir uire very	monitors d reagent ne reagen y small m iter softw	and feedbac It may also t concentrati ixing tanks ir are will be re	be poss obe poss ion (rathe order to quired to	Is are require ible to reduce er than volun o effect a suf o automatica	p-stream ed) can pr se water c netric pur ficiently fa lly manag	revent over consumption mping rate ast ge such
	•	Water used	in cleanin	g and wa	ashing do	wn should be	e minimis	sed by:		
		- evaluatir	ng the sco	pe for re	using was	shwater;				
		 trigger c 	ontrols on	all hose	s, hand la	inces and wa	ashing e	quipment.		
	•	Fresh water	should or	nly be us	ed for:					
		- dilution o	of chemica	als (e.g. f	or gas sc	rubbing med	lia);			
		- vacuum	pump sea	ling (not	e, below,	that this can	be mucl	h reduced or	even elir	ninated);
		- to make	up for eva	aporative	losses.					
	·	Fresh water daily basis.	consump	tion shou	and be dire	ectly measur	ed and r	ecorded regu	ularly - typ	ocally on a
	•	Specific poir monitored p	nts of fres articularly	h water u the discl	ise, circui harge to t	t overflows a he ETP.	and recyc	cled water qu	ality sho	uld be
	•	Water-seale should be re	ed vacuum eviewed by	y conside	may acco ering impr	ount for consi ovements su	derable uch as:	water use ar	id arrange	ements
		 cascadir by using casing (u PLUS 	ng seal wa modern cup to 50%	iter throu lesigns w reductio	gh high to vith impro n);	o low pressu ved internal	re pump: recircula	s; tion of water	within th	e pump
		- filtering a reduction	and coolin n potentia	g seal wa l), or	ater with	a heat excha	inger prio	or to re-use i	n the pun	nps (90%
		 filtering a reduction 	and coolin n potentia	g seal wa l), or	ater with a	a cooling tow	ver prior	to re-use in t	he pump	s (95%
		 filtering a reduction 	and coolin n potentia	g seal wa I),	ater with i	injected fresh	n water p	prior to re-use	e in the p	umps (65%
		OR								
		- recycling	g the hot s	eal wate	r.					
	•	any other co wherever pr	ooling wate acticable,	ers shoul possibly	d be sepa after son	arated from one form of tre	contamin eatment,	ated process e.g. re-cooli	s waters ang and so	and re-used creening.
										Cont

INTRODUC	IOIT		HNIQ	JES	E	MISSION	٧S	II	MPAC	Т
Management Mat in	erials puts	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Water use BAT for water efficiency (cont)	Recy 16. C til ii s tu s c c	Consideration ne re-use of s quors for re-u Site drainage reatment. Su crubber feed onsidered an n some cases rocess direct ecycled selec elow that whi reated water t	ater should be crubber ef ise. or roof wat ch uses ra Operator d justify th s effluent tr ly or in a n tively, whe ch the sys from the E	given to fluents as rer may b inge from rs should e techniq reatment nixture, w en the qua tem can f TP is, or	multiple u s a quence on-site fo demonst ues selece produces ith fresh ality is ad tolerate. is planne	uses of water ch media (in e for a wide v eed to toilet f rate in their a cted and reje a good qual water. While equate, reve The Operato d to be, used	r to minin HWIs) an variety of facilities, applicatio cted. ity water e treated of rting to d or should d and just	hise consum id the treatm uses even a wash down n that such u which may b effluent qual ischarge whi confirm the ify where it is	ption. Th ent of scr fter rudim water, que uses have uses have be usable ty can va en the que positions s not.	is includes ubber entary ench or been in the ry it can be ality falls in which

INTRODUCTION		TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

2.2.4 Typical waste compositions

feedstock

Waste

This section provides general information regarding the composition of wastes.

Physical characteristics

Waste chemical characteristics

Waste chemicals in bulk liquid form are handled by pipeline or tanker and may be burned on the site where they arise. They are often suitable as a substitute fuel. In some cases these types of waste can be potentially less harmful to the environment than the fossil fuels they displace. Bulk quantities produced by large continuous processes tend to be more consistent than those from smaller or batch operations where wastes may not be well segregated.

Waste chemicals can also be in solid form or in drums (containing either liquid or solids). These and occasional waste clearouts (e.g. laboratories etc) can be extremely variable and expensive to identify and process.

Gaseous waste chemicals can cover a wide spectrum from a high CV gas which could be used as a fuel to VOCs at a concentration of a few mg/m³ which are causing an odour problem.

Municipal waste characteristics

Municipal waste varies in composition both seasonally, (e.g. wet garden refuse can make combustion difficult in the autumn) and geographically depending on the type of commercial operations and socioeconomic conditions in the catchment area.

While putrescible materials lead to the odour and health hazards of the raw incoming waste, municipal waste will also be contaminated with a variety of materials that may be a hazard to the environment, e.g. waste paint, oil, pesticide or household cleaners. It may contain batteries, plastics, packaging etc, which may contain toxic constituents such as cadmium or lead, although other controls are tending to diminish these problems. Municipal waste may also contain some clinical wastes.

Wastes that may give rise to high levels of acid gases (e.g. PVC and plasterboard) require careful management to prevent overloading of abatement plant and potential exceedences of emission limits.

Improvements in UK recycling performance are unlikely to effect the combustion performance of plants in the foreseeable future. This is because non-combustibles, such as glass and metal, will be removed as well as some of the combustible fraction.

Clinical waste characteristics

Clinical waste is defined in "The Collection and Disposal of Waste Regulations 1988[°] as including:

- any waste which consists wholly or partly of human and animal tissue, blood or any other body fluids, excretions, drugs or pharmaceutical products, swabs or dressings, or syringes, needles, or other sharp instruments, being waste which unless rendered safe may prove hazardous to any person coming into contact with it;
- any other waste arising from medical, dental, veterinary (including small carcasses in some CWIs), pharmaceutical or similar practice, investigation, treatment, care, teaching, or research, or the collection of blood for transfusion, being waste which may cause infection to any person coming into contact with it.

Clinical waste may contain small quantities of certain hazardous wastes (e.g. prescription only medicines) that cannot practicably be segregated from the waste stream. Where this is the case the waste stream would not be considered by the Regulator to fall within the scope of the Hazardous Waste Incineration Directive (EC/96/67). Where larger quantities of hazardous waste are received (e.g. pharmacy clear outs) that could be segregated without undue difficulty the provisions of the HWID will apply until 28 December 2005 when those provisions will be replaced and revised by those under the Waste Incineration Directive (2000/76/EC).

Sewage sludge characteristics

Sewage sludge arises from two principal sources, either of which require de-watering:

- the removal of solids from raw sewage. This primary sludge has a solids content of about 5% and consists of both organic and inorganic substances;
- the removal by settlement of solids produced during biological treatment processes i.e. surplus activated sludge and humus sludge. This is known as secondary sludge.

The physical and chemical composition will vary according to the treatment processes at the particular works and the nature of the catchment. Sludges from works in industrial catchments, or those with certain industries (e.g. metal plating) may have elevated levels of some pollutants.

Some sludges may be dried or even pelletised to produce a highly combustible waste or for use as a fertiliser.

INTRODUCTION 1			HNIQ	JES	EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Waste feedstock

Drum characteristics

Drums containing residual chemical waste will usually be of the 210 litre steel variety. They may of the "open head" type, used for solids, powders, waxes, greases etc, in which the whole lid is removable, held in place with a clamping ring or of the "tight head" type, used for liquids, with a filler hole and screwed bung in the top. Wherever possible drums will be washed rather than burned out as the former is much less expensive. However in many cases drums cannot be cleaned adequately by this method and incineration is the only method. It is these drums which form the feedstock for this process. The types of contamination are extremely variable.

Animal carcass characteristics

Animal carcasses (which include remains) come typically from domestic pets brought in by individuals, animals from vets and currently BSE bovine stock. As with clinical waste, odour and the potential health hazard from the incoming waste can be significant.

Incoming waste composition

Table 2-1 - Typical constituents

				Typica	l constit	uents - %	% of dry	weight			
Wastes	Hydrocarbon content	Oxygen	Chlorine	Sulphur	Fluorine	Nitrogen	Phosphorus	Mercury	Cadmium	Other heavy metals	Iron
Chemical waste					V	ery variab	le				
Clinical waste	42 - 56	21	1-4	0.07- 0.17	-	0.07- 0.45	-				
Municipal waste	18 - 36	12 - 21	0.25 - 0.8	0.08 - 0.6	0.01 - 0.02	0.4 - 0.8					0.5 - 2.4
Animal carcasses	7	3	as salt	in fur	-	in protein	-	-	-	-	-
Sewage sludge	35 - 57	13 - 19	0.07- 1	0.2 - 1.3	0.03 - 0.06	3 - 5		*	*	*	
Refuse Derived Fuel	Very va specific note S2	Very variable – may comprise a single consistent waste stream (e.g. tyres) or the broader specification of a pelletised or floc fuel derived from municipal waste. For details see earlier guidance note S2 1.05.									
Drum recovery					V	ery variab	le				

* Mercury - typically < 1.0 mg/kg DS (Dry Solids) in non-industrial areas. Average 3 mg/kg rising to EC land spreading limit of 25 mg/kg in industrial areas. However the trend is for the industrial causes to diminish and as dentistry becomes the major source for rural and industrial values to become more closely aligned.</p>

Cadmium - 4-6 mg/kg DS in non-industrial areas. Average 5 mg/kg rising to EC land spreading limit of 40 mg/kg in industrial areas.

Antimony, arsenic, lead, chromium, cobalt, copper manganese, nickel, vanadium and tin - total 700 mg/kg DS in non- industrial areas. Average 2000-3000 mg/kg rising beyond EC land spreading limit to 8000 mg/kg, on occasions, in industrial areas.

INTRODUCTION		TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

Waste feedstock

Table 2-2 - Combustion characteristics

Wastes	Calorific Value (MJ/kg) (Note 1)	Moisture (%) (Note 2)	Volatile hydrocarbon content (%) (Note 3)	Ash content (%)						
Chemical waste		very	variable							
Clinical waste	13-25(av 21)	9-15	42-56	7-20						
Municipal waste	6.3 -15.8	33	18-36	13-32						
Animal carcasses	<3 whole carcass	85								
Sewage sludge	12-19	60-65		26-41						
Coal - for comparison	23	8	26	10						
Refuse derived fuel	Very variable – may the broader specific waste. For details s	variable – may comprise a single consistent waste stream (e.g. tyres) or proader specification of a pelletised or floc fuel derived from municipal te. For details see earlier guidance note S2 1.05.								
Drum recovery	little	very variable								

Note 1 Calorific values are net, i.e. after deducting the heat required to evaporate the water.

Note 2 % of weight of incoming waste as received.

Note 3 % of dry weight.

INTRODUCTION		N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

2.3 The Main Activities (including abatement)

This section deals with the main incineration installation activities (including any "directly associated activities") that may have an influence upon **preventing** the generation of potentially polluting emissions:

(see Section 2.3.1)
(see Section 2.3.2)
(see Section 2.3.3)
(see Section 2.3.4)
(see Section 2.3.5)
(see Section 2.3.6)
(see Section 2.3.7)

Abatement technology for releases to air and water can be found under Sections 2.3.8 to 2.3.11.

The minimisation of waste is dealt with as follows:

- by comment within a section dealing with a particular technique that may have significant influence over the quantities or nature of wastes produced e.g. poor burnout leading to elevated quantities of bottom ash,
- by comments in Section 2.2 where wastes arise from raw material selection,
- by summary of the key waste minimisation considerations under Section 2.6.



Describe the proposed installation activities and the proposed techniques and measures to prevent and reduce waste arisings and emissions of substances and heat (including during periods of start-up or shut-down, momentary stoppage, leak or malfunction).

With the Application the Operator should:

- provide adequate <u>process descriptions</u> of the activities and the abatement and control equipment for all of the activities such that the Regulator can understand the process in sufficient detail to assess the Operator's proposals and, in particular, to assess opportunities for further improvements. This should include:
 - process flow sheet diagrams (schematics);
 - diagrams of the main plant items where they have environmental relevance; e.g. landfill liner design, incinerator furnace design, abatement plant design etc.;
 - details of any chemical reactions and their reaction kinetics/energy balance;
 - control system philosophy and how the control system incorporates environmental monitoring information;
 - annual production, mass and energy balance information;
 - venting and emergency relief provisions;
 - summary of extant operating and maintenance procedures;
 - a description of how protection is provided during abnormal operating conditions such as momentary stoppages, start-up, and shut-down for as long as is necessary to ensure compliance with release limits in Permits;
 - additionally, for some applications it may be appropriate to supply piping and instrumentation diagrams for systems containing potentially polluting substances.

If there is uncertainty, the degree of detail required should be established in pre-application discussions.

- 2. describe the current or proposed position for all of the indicative BAT requirements for each subsection of 2.3, or any others which are pertinent to the installation;
- 3. identify shortfalls in the above information which the Operator believes require longer term studies to establish;
- 4. demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measure;

INTROD	UCTIC	N TE	CHNIQ	UES	E	MISSIO	NS	1	MPAC	T
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

In assessing the integrated impacts of proposals and balancing the impacts of different techniques it should be noted that energy should be taken into account whether or not there is a Climate Change Agreement or Trading Agreement in place (see Section 2.7.2).

Indicative BAT Requirements

See each subsection of this Section 2.3.

INTRODU	CTIO	N TEC	HNIQ	UES	EI	MISSIO	NS		MPAC	T			
Management i	aterials nputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues			
Incoming waste handling	 2.3.1 This s Se Ma Th The probability Deginn 2.3.1. Europ waste for ins 	I Incomi ection deals we couring the suit anagement of the need for pre- revention of we ning of Section I Legislat eean Directives at all installations not of	ing was with the tect tability of t the waste e-treatmen aste produ a 2.3. The tive oblig s contain n ions that f covered by	the man chniques re he waste to on the site t uction at the managem mandatory i all within the the Direct	ageme equired to o be acce e installa nent of wa or incom minimum neir remit tives.	ent manage in epted before tion is dealt astes produ ning waste requirement The stanc	icoming v e it is del i in this g ced is de e manag dards des	waste effectiv livered uidance as c ealt with in Se gement rding the han scribed may	vely. This lescribed ection 2.5 odling of a also repre	at the <i>incoming</i> esent BAT			
All new plants covered by the Waste Incineration Directive (2000/76/EC) will be required to meet least the standards outlined in the Directive from 28 December 2002. Site specific BAT may require these standards to be adopted before this date, or even standards that go beyond those required WID.													
	The "c require outline	Incoming waste management at existing municipal waste incinerators The "old" municipal incineration directives (89/369/EEC and 89/429/EEC) did not contain any specific requirements in respect of the handling of incoming waste. BAT standards therefore apply. These are outlined later in this Section (2.3.1).											
	Existin the W below standa	Existing municipal waste incinerators will be expected to meet at least the minimum requirements of the Waste Incineration Directive (2000/76/EC) by 28 December 2005 at the latest (see this Section below). Site specific BAT may require these standards to be adopted before this date, or even standards that go beyond those required by WID.											
	As the the W	e standards for ID represent B	waste ch BAT, they s	ecking, doo should be a	cument n adopted a	nanagemen at all new pl	t and wa ants fror	iste handling n the first day	that are o y of opera	outlined in ation.			
	Incor	ning waste i	manager	nent at e	xisting l	hazardou	s waste	incinerato	rs				
	The st Incine Decer	tandards below ration Directive mber 2005:	w relate to e (94/67/E	all installa C). WID v	tions that vill apply	fall within t to existing i	the contr installatio	ol of the curr ons in this se	ent Hazaı ctor from	rdous Waste 28			
	HWID	waste manag	gement re	quiremen	ts applic	able to exi	isting H\	NIs:					
	 the ord the hu 	e Operator sha der to prevent e environment man health.	all take all or, where , in particu	necessary that is not lar the poll	measure practicat lution of a	es concernir ale, to reduc air, soil, surf	ng the de ce as far face and	elivery and re as possible r ground wate	ception on negative e er, and the	f waste in effects on e risks to			
	No the	ote: This requi	rement rei Section 2.	lates to the <mark>3.1.4</mark> for gi	e design o uidance o	of the recep on waste ha	tion area Indling.	a as well as ti	he manag	ement of			
	 Pri of 	ior to accepting the waste cover	g the wast ering:	te at the ind	cineratior	n plant, the	Operato	r shall have a	available a	a description			
	-	The physical information n	, and as fa lecessary	ar as practi to evaluate	cable, the e its suita	e chemical bility for the	composit intende	tion of the wa d process.	aste and a	all			
	-	The hazard of precautions t	haracteris o be taker	tics of the in handlir	waste, th ng the wa	e substanc ste.	es with v	vhich it canno	ot be mixe	ed, and the			
		every iter onate to t ecessary analysis ne waste s ission lim	m to be he risk to evaluate regimes for such that its it values in										
	• Su ca	Ifficient informant information in the either:	ation shou	ld be obtai	ined prior	to receipt of	of the wa	iste onto the	site to en	sure that it			
	-	Be safely offl pretreatment	oaded for or	safe stora	ge and fu	rther chara	cterisatio	on prior to inc	cineration	or			

INTROD	UCTION TECHNIQUES EMIS						MISSIO	NS		MPACT		
Management	Materia inputs	ls <mark>Act</mark> aba	ivities/ tement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	nstallation issues	
Incoming waste handling		- Be su Prior t shall b	directly in fficiently c o acceptir be carried	ncinerated letailed giv ng the was out by the	without ne en the natu te at the in Operator:	ed for fur ure and s cineratior	ther charac ource of the n plant, at le	terisatior waste) east the fo	i (i.e. the des	scription is	already edures	
		- De No	etermination ote: The C ading mec	on of the m Operator sh chanism we	nass of the nould make eigh device	waste; provision s, weigh	n to determi vessels or t	ine the m hrough c	ass by the u alculation. 1	se of a wei This may be	ighbridge, e carried	
		- the tho an da	e checking ose requir d control ngerous g	g of those o ed by Cou of shipmer goods trans	documents ncil Regula nts of waste sport regula	required Ition (EEC within, in ations,	by Directive C) No. 259/9 nto and out	e 91/689/ 93 of 1 Fe of the Eu	EEC and, wi bruary 1993 ropean Com	here applic on the su munity and	able, pervision d by	
		No wa dis	o te: Speci iste is acc sposal site	al waste a cepted. Th e.	nd transfro is includes	ntier ship both pre	ment of wa -notification	ste paper and noti	rwork must b fication upor	e checked transfer to	before the the	
		- Th to ide the	e taking c verify con entify the r e incinerat	of represen formity wit nature of th tion.	tative sam h the desci ne wastes t	ples, unle ription pro reated.	ess inapprop ovided and f These samp	oriate, as to enable bles shall	far as possi the compete be kept for a	ble before ent authori at least 1 m	unloading, ties to nonth after	
		No ca sa ac av ha	te: This r ses this w mpling. T count of ti ailable fro zards that	requirement yould be im The frequent he source of the was t the waste	nt does not apractical (ency and sco of the wast ate produce a may pose	mean tha e.g. labor ope of sa e, its van r, experie in relatio	at every iten atory dispos mpling and iability, the r ence of deal on to the pro	n requires sals). Bu /or analy reliability ling with pocess con	s sampling o Ik loads are sis should be and adequa wastes of the ocerned.	r analysis, more likely e adapted t cy of the in at type and	in many to require to take formation the likely	
	·	Exem under same	otions ma takings ind level of pr	y be grante cinerating f rotection is	ed from the their own w met.	e second vaste at tl	and third of he place of	the abov productio	e bullets for on of the was	industrial p te provideo	plants and d that the	
		<i>Note:</i> the sa accep greate	This mea me length ting waste er knowled	ns that in-i ns (in relati e from othe dge of the v	house plan ion to docu er sites. It i waste strea	ts, burnir ment che reflects th am and th	ng their own ocking, samp ne fact that p nat the risks	wastes o bling and bersons i are cons	do not neces analysis) as ncinerating t requently rec	sarily need those plar heir own w luced.	l to go to nts aste have	
	ine Wi	comin ID	g waste	managei	ment at a	ll incine	eration and	d co-inc	ineration p	olants co	vered by	
	Ex De spo be	isting ir cembe ecific B yond th	nstallation r 2005 at AT may ro lose requi	s will need the latest. equire thes red by WII	l to meet th New insta se standarc D.	e standa llations w ls to be a	rds of the "r vill need to c dopted befo	new" incir comply fro ore this d	neration director om 28 Decer ate, or even	ctive (WID) nber 2002. standards	by 28 Site that go	
	wi	D inco	ming was	ste manag	jement req	luiremen	its:					
	Th (20	ese sta)00/76/	indards aj EC):	oply to all i	nstallations	s that fall	within the c	ontrol of	Waste Incine	eration Dire	ective	
	•	Article precau practic and gr shall r	e 5(1) The utions con cable nega roundwate neet at lea	e Operator acerning th ative effect er as well a ast the req	of the incir e delivery a ts on the er as odours a uirements	neration of and recept nvironme and noise set out in	or co-inciner otion of was nt, in particu , and direct [Articles 5(ation pla te in orde ular the p risks to h 3) and 5(nt shall take er to prevent ollution of ain numan health 4) below].	all necessa or to limit a r, soil, surfa n. These m	ary as far as ace water neasures	
		Note: the rea guidar	This appl ception ar nce will in	lies to haza rea as well the majori	ardous and as the mai ty of cases	non-haz nagemen fulfil the	ardous was t of the was se requirem	te inciner ste. The l ents.	ation. It rela use of BAT a	ites to the o as outlined	design of in this	
	ŀ	Article to the	e 5(2) The EWC, prie	e Operator or to accep	shall deter oting the wa	mine the aste at th	mass of ea e incineratio	ch catego on or co-i	ory of waste, ncineration p	, if possible plant.	e according	
		Note: be trea make device to the	This appl ated shou provision es, weigh site and r	lies to haza Id be outlin to determi vessels or recorded ou	ardous and ned in the a ine the mas through ca n the releva	non-haza applicatio ss by the alculation. ant paper	ardous was n, using EN use of a we This may work.	te inciner VC where ighbridge be carrie	ration. The v possible. T e, loading me d out before	vaste categ he Operato echanism w the load is	gories to or should veigh delivered	
	•	Articl	e 5(3) Prio	or to accep	ting hazard	dous was	te at the inc	ineration	or co-incine	ration plan	t, the	

Operator shall have available information about the waste for the purpose of verifying, inter alia, compliance with the permit requirements specified in Article 4(5). This information shall cover:

INTRODU	CTION	TEC	HNIQ	UES	E	MISSIO	NS		MPAC	т	
Management N	laterials inputs a	Activities/ batement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	
Incoming waste	(a)	all the admi mentioned i	nistrative i in paragra	informatio ph 4(a);	n on the	generating p	orocess	contained in	the docur	nents	
handling	(b)	the physica information	l, and as fa necessary	ar as prac / to evalua	ticable, c ite its sui	hemical cor tability for th	npositior ne intenc	n of the wast led incinerati	e and all c ion proces	other ss;	
	(c)	the hazardo the precaut	ous charac ions to be	teristics o taken in h	f the was andling t	te, the subs he waste.	tances v	vith which it o	cannot be	mixed, and	
	Not san be a info pro- suff trea	te: This does npled. Whilst adopted. In p rmation nece cess". Regul ficient to enab ttment does n	not require meeting th particular th ssary to ev ators shou ole the Ope ot give rise	e everythir he minimu his is empl valuate su valuate su	ng to be t m require hasised t itability fo sampling ategorise dence of	horoughly c ements an a by 5(3)(b) – or the intenc g and analys the waste s f authorised	haracter approach which de led [haza sis regim such tha emissio	rised, nor even proportiona escribes a ne ardous waste les for incom t its subsequ n limits in pa	ery item to te to the r eed to hav incinera ing waste ent storag rticular.	be isk should e "all other tion that are ge and	
	 Suf can 	Sufficient information should be obtained prior to receipt of the waste onto the site to ensure that it can be either:									
	-	safely offloaded for safe storage and further characterisation prior to incineration or pretreatment, or									
	-	incinerated w detailed giver	ithout nee n the natur	d for furthe	er charac	terisation (i e waste).	.e. the d	escription is	already sı	ufficiently	
	• Art the	icle 5(4) Prior following rece	r to accept eption proc	ing hazaro cedures sh	lous was nall be ca	te at the inc irried out by	ineration the Ope	n or co-incine erator:	eration pla	nt, at least	
	(a)	the checkin those requi and control dangerous-	g of those red by Cou of shipme goods trar	documen uncil Regu nts of was nsport regu	ts require lation (El te within ulations;	ed by Directi EC) No 259 , into and ou	ive 91/68 /93 of 1 ut of the	39/EEC and, February 199 European Co	where ap 93 on the ommunity	plicable, supervision, and by	
	(b)	the taking of far as possi paragraph (nature of th incineration	f represen ble before 3 by carryi e wastes t	itative san unloading ng out cor reated. Th	nples, un g, to verif itrols and iese sam	less inappro y conformity I to enable t ples shall b	opriate, e / with the he comp e kept fo	e.g. for infect e information petent author or at least on	ious clinic provided ities to ide e month a	al waste, as for in entify the fter the	
	Not haz this The sou was may	te: This requi ardous waste would be imp frequency al rce of the waste producer, pose in rela	rement onl es. It does practical (e nd scope c ste, its var experience tion to the	ly relates t not mean e.g. labora of sampling iability, the e of dealin process c	o inciner that eve tory dispo g and /or e reliabilit g with wa oncerned	ation or co-i ry item requ osals). Bulk analysis sh ty and adeq astes of that d.	incinerat lires san c loads a ould be uacy of t t type an	ion installation npling or ana re more likel adapted to ta the information d the likely h	ons receiv lysis, in m y to requil ake accou on availab azards th	ing pany cases re sampling. nt of the le from the at the waste	
	• Arti und was	icle 5(5) Exer ertakings inci ste provided ti	mptions mainerating o hat the rec	ay be grar r co-incine uirements	ited from erating or of this D	[articles 5(2 aly their own Directive are	2), (3) ar n waste a met.	nd (4)] for ind at the place o	lustrial pla of generat	nts and ion of the	
	Not the acc grea	t e: This mean same lengths epting waste ater knowledg	s that in-h s (in relatio from other ge of the w	ouse plan on to docu r sites. It r vaste strea	ng their own ocking, samp ne fact that p nat the risks	wastes oling and persons are cons	do not neces I analysis) as incinerating t sequently rec	ssarily nee s those pla their own duced.	ed to go to ants waste have		
	2.3.1.2 incine	? Other in ration proc	coming v esses	waste ma	anagem	ent requir	rements	s applicabl	e to all		
	• Was EL\	ste pre-treat r /s and to prev	nent shou /ent unnec	ild be carri cessary wa	ed out to aste prod	the degree uction.	necess	ary to contro	l emission	s within	
	 A hi prov 	igh standard (vided and ma	of housek intained to	eeping sh o clean up	ould be i spilled m	maintained i aterials.	in all are	as with suita	ble equipi	ment	
	• Loa har	iding and un d standing. S	loading of such areas	f vehicles should ha	should or ave appro	nly be done opriate falls	in desig to an ad	nated areas equate drain	provided age syste	with proper m.	
	Unc con doo	contained or p trol e.g. nega rways.	otentially tive pressu	odorous v ure create	waste sh d by feed	ould be stor ling combus	ed inside ation air,	e buildings w automatic or	ith suitabl restricted	e odour I orifice size	
	• Fire	e fighting pro mical wastes	visions are	e required	, especia	lly for MWI	receptio	n bunkers, C	WI storag	e and for	

INTRODUC	CTION TECHNIQUES EMISSIONS IMPACT
Management Ma	iterials Activities/ Ground Waste Energy Accidents Noise Monitoring Closure Installation
inanagonioni ir	abatement water matter charge house house monitoring clocal a issues
Incoming	- Evelo and treatment chemicale should be stared in tenks or siles. Liquid fuels and chemicals, if
waste handling	 Puers and treatment chemicals should be stored in tanks of slips. Liquid idels and chemicals, if not supplied in drums, should be stored in tanks provided with closed loop vapour systems and/or scrubbers. Silos for powders or finely divided particulate matter require fabric filters.
<u> </u>	Contamination of rainwater should be minimised through the provision of roofing and drainage
	segregation. Provision must be made for dealing with all forms of potential water release or run-off. Storage capacity should be provided for contaminated rainwater to allow for sampling and testing prior to release (see also WID article 8(7)).
	See control of fugitive emissions to water (Section 2.3.11) for further details.
	2.3.1.3 Municipal waste management
	Legislative requirements for the delivery and reception of wastes at MWIs will arise from WID and are considered to represent BAT. There are specific requirements in respect of checking of statutory documentation, and a general duty to "take all necessary precautions". This is therefore a site and operation specific, risk based judgement. Operators must therefore adopt procedures and a management system to enable the identification and management of the risks associated with the receipt of the wastes. Procedures should take account of the matters noted in this guidance and the duties arising from the Directive (see Section 2.3.1.1 - legislative obligations).
	Incoming municipal waste should be:
	 in covered vehicles or containers.
	 unloaded into enclosed reception bunkers or sorting areas with odour control (see below).
	 Designs and handling procedures should avoid any dispersal of litter.
	 Techniques should be used to improve the homogeneity of waste fed to the incinerator.
	 Inspection procedures should be used to ensure any "problem" wastes are removed and placed in
	a designated storage area pending removal.
	• Waste may then transferred to the feed chute e.g. by a grab crane.
	 Where the waste is not pre-treated or sorted, smaller grab sizes should be used on a more frequent basis to allow for greater waste inspection
	To minimise odour:
	 doors, normally self closing, should be provided for any potentially odorous indoor areas;
	 bunkers should be ventilated with the extracted air being used as a source of furnace combustion air;
	 during shut-down, particularly where there is only one furnace, doors will limit odour spread while still allowing vehicle access and air should be extracted via a separate system.
	 extracted air which is not incinerated should be treated (see Section 2.3.13 - Odour);
	 where there is a recycling facility before the incinerator, the volumes of odorous air which need to be extracted may well exceed the furnace requirements if attention is not given to building sealing arrangements at the design stage;
	 bunker management procedures (mixing and periodic emptying and cleaning) should be employed to avoid the development of anaerobic conditions;
	 wastes shall be removed on a first in, first out basis so as not to exceed a specified maximum storage time (e.g. 4 days or less if problems arise);
	• During shut downs waste shall be diverted away from the site if odour management is not effective.
	 Generally, multiple stream plants are preferred to large single stream plants to provide continuity of odour control and waste movement.
	 the quantity of incoming waste being stored should be limited to the agreed design limit, and must be confined to the designated areas.
	Where dust emission needs controlling, low volume water fog sprays should operate above the storage bunkers. Liquid run-off and wash down from the storage and handling areas should be minimised and be used in the process, such as in the ash quench, wherever possible.
	Pre-treatment of municipal waste:
	Preferred practice for raw municipal waste may include the removal of large bulky items, followed by the extraction of recyclables and the shredding the remaining waste, although this may be carried out prior to delivery to the installation. Such pre-treatment can help ensure a more consistent feed to the furnace thereby aiding good process control and preventing emissions.
	A particle size reduction system is accordial for fluidiand had incincutors. However, there have

A particle size reduction system is essential for **fluidised bed incinerators**. However, there have been significant operational problems (including high maintenance costs and fires) with front-end **materials recycling facilities** (MRF) handling raw mixed municipal waste.

INTRODUC	CTION TEC	CHNIQUES	EMISSIONS	IMPACT							
Management Ma	terials Activities/	Ground Waste	Energy Accidents Noise	Monitoring Closure Installation							
in	puts abatement	water	Energy Accidents Noise	issues							
Incoming waste handling	Dirty MRFs may pro where recyclables a provided emission li justified. Where was improve consistency Applicants must cl	duce lower grade reare already being rem mit values are guara ste feed pre-treatment and this should be a early demonstrate	cyclables than those segreg noved by means of doorstep nteed, the additional cost of nt is not practicable the max defined in operating procedu why waste shredding / MR	ated at source. Furthermore, collection or other schemes, a front end MRF may not be imum degree of waste mixing will ires. Where not adopted, F techniques are not BAT.							
	Decisions regarding take account of the the waste delivered. catchment area (as further front end MR statutory emission li by the costs.	the need for and ext wider waste strategy For example, where demonstrated by rec F or shredding may mit values can be se	tent of waste treatment at m adopted in the locality as th e removal of recyclables is b cycling performance or facilit be justifiable as BAT for the cured and the additional environment	unicipal waste incinerators should is will influence the composition of being carried out within the waste y provision) incineration without remaining waste, provided <i>v</i> ironmental gains are outweighed							
	2.3.1.4 Chemical waste management										
	Legislative requirem eventually WID). The documentation, but site and operation s management system receipt of hazardous guidance and the du	tents for the delivery ne legislation sets so also include a genera pecific, risk based juc m to enable the ider s wastes. These pro uties arising from the	and reception of wastes at I me specific requirements in al duty to "take all necessary dgement. Operators must th ntification and management cedures should take accoun Directives (see Section 2.3.	HWIs arise from HWID (and respect of checking of statutory y precautions". This is therefore a herefore adopt procedures and a of the risks associated with the t of the matters noted in this 1.1 - legislative obligations).							
	In particular, sufficie	nt information should	d be obtained prior to receip	t of the waste onto the site to							
	 safely offloaded for safe storage and further characterisation prior to incineration or pretreatment or incinerated without need for further characterisation (i.e. the description is already sufficiently detailed given the nature and source of the waste and the intended treatment) 										
	This should include consideration of odour risks (e.g. mercaptans, thiols, amines) – with high odour risk wastes offloaded, stored and transferred along dedicated "high odour" routes with closed loop or abated vents.										
	Whether liquids or sludges are delivered by tanker, in drums or by pipeline, the waste should be held in a buffer store pending suitable chemical analysis prior to blending and feeding to the incinerator. This should include checks on its compatibility with waste already in any bulk storage tank(s) where it will be stored.										
	For merchant incin	erators, systems sh	ould be in place to:								
	 ensure that wast 	e arrives with inform	ation covering:								
	 its physical a 	nd chemical compos	sition;								
	- any other info	ormation necessary t	to assess its suitability for in	cineration;							
	 Its nazard cn substances v 	aracteristics; vith which it cannot h	e mixed [,] and								
	 handling pred 	cautions.									
	 confirm the information 	mation by:									
	 checking that 	t the quantity is as de	eclared by the consignor;								
	- documentatio	on checks; and									
	 sampling who Small scale of receiving tan undesirable of 	ere appropriate. Sar compatibility tests are k to ensure that there consequences.	nples should be kept for at l e normally carried out with a e are no reactions which lea	east 1 month after incineration. sample of the contents of the d to heat release, gassing or other							
	For in-house incine particular, where no burned.	erators procedures s n hazardous waste is	should be in place to give the s being burned, to ensure th	e same level of protection, and, in at only the agreed wastes are							
	Pre-treatment of ha	azardous waste:									
	The highly heteroge to be gained from w	neous nature of haza aste blending and / c	ardous wastes means that th or pre-treatment.	nere may be significant advantages							

A procedure should be in place to ensure that waste is treated / blended to give the most constant combustion conditions possible.

INTRODUC	TION	TE	CHNIQ	UES	E	VISSIO	NS		MPAC	Т	
Management Mat	terials Act	tivities/	Ground	Waste	Enerav	Accidents	Noise	Monitoring	Closure	Installation	
In	puts aba	itement	water							Issues	
Incoming waste	Liquid wa combustio	stes may on.	require scr	eening or fi	Itering to	remove sc	lids and f	thereby ensi	ure steady	'	
handling	Storage of	of hazard	ous waste	s:							
	Bulk stora	age tanks	should:								
	 be site 	ed above g	ground;								
	 be built 	nded ;									
	 be fitte 	ed with pro	otection ag	ainst overfi	lling;						
	 have level gauges visible from the filling point which should be clearly labelled; and have releases from vents controlled by back venting (particularly for high vapour pressure fluids), by feeding to the incinerator, by carbon filters or by condensation possibly using an appropriate refrigerant. The use of floating roof tanks can avoid venting but introduces the problems of seals. Scrubbing is acceptable if the VOCs will not subsequently come out of solution or if the liquor can be adequately treated or can be fed to the incinerator. (There may be safety implications with both venting to the incinerator and with the use of carbon filters. The Operator should consult HSE on details of the specific design.) 										
	Drummed waste should:										
	 be stored in areas protected from heat and direct sunlight, preferably under cover; 										
	 be sto 	ored on an	impervious	s surface w	hich has	an adequa	te fall to	a collection	sump;		
	 not be 	e stored m	ore than 2	rows high;							
	 be analysed as soon as practicable, and then either emptied into bulk storage tanks or passed to the incinerator for destruction. The authorisation should limit both the maximum number of drum which may be held on site and the maximum periods that drummed waste may be held after receipt; and 										
	 be submitted 	bject to ro	utine proce	dures for c	hecking	the conditio	n of drun	ns and palle	ts.		
	Air from d methods disposed	lrum stora should be of without	ge opening employed giving rise	g/transfer po to ensure t to emissio	oints sho hat conta ns to atn	uld be trea ainers which nosphere a	ted as for have be nd odours	tank vapou en emptied s.	rs above. should be	Similarly, stored and	
	Pipework flanges. N and a me	should p Where it n ans provid	referably be nust be und ded to dete	e overgrou derground i ct any leaka	nd and w t should age into	velded wher be double-o that outer c	re possibl contained ontainme	e to minimis , for exampl ent.	e the num e by a sle	nber of eve or duct,	
	Maintena	ince:									
	The site p equipmen	preventati nt to preve	i ve mainte Int fugitive o	nance prog odour (or of	gramme ther) rele	should incl ases; e.g. s	ude asse seals on p	essment of a oumps, valve	II waste ha	andling nges etc.	
	2.3.1.5	Clinica	l waste m	anageme	nt						
	All installa be design latest. So	ations sho led to mee ome faciliti	uld conside et the Direc ies may alr	er the implic tive standa eady be rec	cation of rds and quired to	the waste i existing one meet the re	ncineratio es upgrac equireme	on directive. led by 28 De nts of HWID	New facil ecember 2	ities should 2005 at the	
	The Agen version 1. managen referred to	ncy has iss .2 - Feb 20 nent licens o for full d	sued guidar 001) in resp sing. Its req etails if req	nce (Ref: To bect of the s quirements uired.	echnical storage a have be	guidance o and handlin en taken in	n clinical g of clinic to accour	waste mana al waste at nt in this sec	agement fa sites subje tion, but it	acilities - ect to waste should be	
	A control Modern s be checke	system sh ystems us ed. Waste	nould be in se bar codir e descriptio	place that f ng and com ons should b	t racks th puter log pe suffici	ne waste fro gging. Stati ent to ident	om the po utory was ify the na	bint of arising te transfer of ture of the v	g to the ind locumenta vaste and	cinerator. ation should its source.	
	The opening of containers to check waste meets relevant descriptions is not advised. This being the case, Operators of incineration facilities are responsible for providing waste producers and carriers of sufficient information to enable them to adequately segregate waste at the point of production, and for auditing these systems to ensure they are carried out. This is to prevent the inclusion of wastes with particular environmental or regulatory significance (e.g. hazardous waste which may require the installation to meet the requirements of HWID)										
	Clinical w be able to can vary g arrives it o variability	aste is so identify (greatly de can upset by provid	metimes pr without ope pending on the inciner ing means	re-sorted by ening any b the source ator operat to secure a	/ a separ ags) the e of the w ing cond a more co	ate waste h nature of th aste and if itions. Insta onsistent fe	nandling one waste a large c allations s ed. This	company. T received, sii onsignment should acco may include	the Operation nce the cation of a partion unt for this selective	tor needs to lorific value cular type potential loading /	

Г

INTRODU	CTION TEC	CHNIQ	UES	E	MISSIO	NS		MPAC	Т			
Management Ma	aterials Activities/	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation			
il	nputs abatement	water	maete	Lineigy		110100	monitoring	clocalo	issues			
Incoming waste	management of ide facilitate mixing.	entified wa	aste types	to blend	the feed ar	nd/or larç	ge pre-chargi	ing feed h	noppers to			
handling	Waste is ideally har immediately prior to elsewhere, and min (sealed) can be use individual plastic ba health and safety le	ndled in lock incineratio imises spill d for transp gs of clinica gislation fro	ked, readil n, as this ages and oorting and al waste sl om 1 Janu	y cleanal provides the risk to d feeding hould be ary 2002	ole, wheeled assurance for Operators clinical was minimised. and are con	d contair that the . Alternaste into t Rigid c nsidered	ners which an waste does r atively, dispo he incinerato ontainers an BAT.	re only op not get div osable cor or. Handli re a requi	pened verted ntainers ing of irement of			
	Plastic waste contai chlorine loadings.	iners that w Operators s	/ill be incin hould adv	erated s l ise suppl	h ould not b iers of this r	be made requirem	of PVC in o ent and reinf	rder to re force it as	duce required.			
	Waste awaiting incineration should be held in a secure storage facility designed and constructer as to contain spillages and prevent access by unauthorised personnel. This should be covered we the containers are not reliably weatherproof. New installations should provide storage within the building that houses the incinerator and dust ventilation air to the incinerator wherever possible. Existing facilities should consider this option, but physical restriction may prevent this. Sharps containers should be stored securely prior to incineration.											
	A container wash a conveyors etc shoul water to be fed back be high. With even may not be complet methods for minimis biodegradable both in some, large-scale	and disinfe Id be design to the inci the best se ely broken sing the dis in the STW e situations	ectant are ned to faci nerator (e ewage trea down and charge an / and in th	a should ilitate effe .g. ash qu timent wo could pa d should e receivir	be provided ective disinfe uench) it is prks, the str ss to the re provide jus ng water. O	d. All sto ection. \ accepted ong disir ceiving \ tification on site ef	brage areas, While it is pred that the wa offectants diso water. Opera that the disin fluent treatm	containe eferable for shdown v charged t ators sho nfectants ent may	rs, loaders, or process volumes can o sewer uld consider used are be required			
	Storage times for conspecific wastes, with known risk of contain autoclaving). As a g refrigerated storage (< 0 Celsius). Contain	linical wast n the highes mination wi general guid (at <5 Cels ingency pla	te should I st risk was th human de, waste sius) is pro ins should	be minim stes proce pathoger should n ovided. S be made	ised and se essed as so as should be ot routinely Storage in e e to deal wit	lected a on as po e treated be store xcess of h breakd	ccording to th ossible. Clini I prior to its c d for longer to one week sh downs.	he nature ical waste collection than 48 h nould be i	of the e with a (e.g. ours unless in freezers			
	2.3.1.6 Animal	waste ma	nageme	nt								
	Animal carcass in	cineration										
	Installations burning subject to BAT and amended to include	only anima legislation controls sp	al carcass made to in pecific to in	es are ex nplement ncineratio	empt from Directive 9 on (e.g. air e	the requ 0/667/El emission	irements of V EC – this Dir limits).	VID, but i ective ma	remains ay be			
	Carcasses awaiting or in separate, refrig	incineratio gerated stor	n should b rage. This	e stored s should:	either in the	e refrigei	ated trailers	in which	they arrive			
	be totally enclos	ed with a s	elf-closing	door;								
	 be lockable; be bird, insect. 	and rodent	-proof [.]									
	 be blid-, insect- be cleaned and 	disinfected	each wee	k: and								
	 have effective m 	eans of od	our contro	I.								
	Odours can be treat are strongly preferre dragging them acro- contain the odours.	ted in a var ed to maski ss the floor	iety of way ing. Carca . Smaller	/s - <mark>see S</mark> asses sho carcasse	Section 2.3. buld, at all ti s are best h	13 - Odo mes, be nandled	our. Contain transported in plastic who	ment and so as to p eeled bin	treatment prevent s with lids to			
	All vehicles, contain collection, transfer a disinfection, be cons	ers, trailers and handlin structed of	s, storage g of carca imperviou	areas, loa sses sho s materia	aders, conv uld be desig Is and be ke	eyors ar gned for ept clear	nd equipment easy and eff า.	t used for fective cle	⁻ the eansing and			
	Floors should have materials and shoul	a chemical d be sloped	resistant f d to a hold	finish to p ing pit.	prevent atta	ck by the	e cleaning ar	nd disinfe	cting			
	Washdown water m adequate on-site eff checked and mainta should therefore be	ay contain fluent facilit ained. Hype considerec	pathogens ies are pro ochlorite d d carefully.	s and sho ovided. S lisinfectar	ould be inject Sumps and t nt may give	cted into transfer rise to a	the furnace equipment ir idditional chlo	except whether the second seco	here hould be ding and			
	PVC packaging sho	uld be avoi	ided to pre	event add	itional chlor	ine load	ing on the in	cinerator.				

INTROD	UC	TION	TEC		UES	EI	MISSIO	NS		IMPACT				
Management	Mat in	erials Act	vities/	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation			
			lement	Water							loodoo			
Incoming waste		Other Ani	mal Wast	te Incinera	ation (incl	uding MI	BM)							
handling		Installation the guidar Section 2.	ns that bui nce given a 3.1.1 in re	rn any anin above) mu espect of in	nal wastes st therefor coming wa	other the e comply aste hand	an carcasse with the re lling, delive	es are sul quiremer ry and re	bject to WID nts of the Dir ception.	and (in a ective ou	addition to tlined in			
		Similar sta standards of the insta	andards to may be a allation.	those outl ppropriate	ined abov in some c	e for carc ases dep	ending upo	ation sho n the nat	ould be adop ture of the wa	ted. Clin aste and	ical waste the location			
		Security o	f stored w	astes is pa	irticularly i	mportant	in respect of	of meat a	arising from c	culls to pr	event theft.			
		MBM hand to prevent vents to th static.	dling shou agglomer ie delivery	ild take acc ration. Silo / tanker. E	ntially dusty nature. It should be stored in agitated silos e filtered and loading operations should use closed loop required during loading to prevent dust explosions due to									
		Where MBM (or other wastes) have been stored for considerable periods it is likely that compacti have taken place. The lumps may therefore require size reduction prior to incineration – this will particularly the case where fluidised bed combustors are used. Longer-term storage also increas risk that the waste will contain pyrolysed fat that may self-combust. Adequate fire detection and o should be provided.												
		Transport pneumatic	of MBM fr means.	rom the sild Screw con	o to the co veyors ma	mbustor ay assist v	may be ach with breakin	ieved usi Ig up lum	ing belt or so ips.	crew conv	/eyors, or by			
		All storage means of combustor	e and trans containme r or by the	sfer should ent of all pla e use of (we	l be carrie ant within ell-maintai	d out suc a building ned) seal	h that odou g with auton ed storage	r is conta natic doo and trans	ained. This r rs and extra- sfer systems	nay be pı cted air p	rovided by assed to the			
		2.3.1.7	Sewage	sludge r	nanagen	nent								
		Sewage s content to as autoge energy ad	Sewage sludge should be de-watered to produce a sludge cake of sufficiently low water moisture content to be incinerated without the use of supplementary fuel (except for start-up). This is referred to as autogenous or autothermic combustion. Further drying will not necessarily provide any overall energy advantage, although it would minimise problems of visible plume (see also Section 2.3.5).											
		Sewage sludge cannot be effectively de-watered without prior conditioning. This is normally by the addition of polyelectrolyte but can be achieved by thermal conditioning, the addition of ash, or the addition of chemicals such as calcium hydroxide, ferrous sulphate, ferric chloride, or aluminium chlorohydrate.												
		De-waterin Where the a sludge o photograp	ng system de-water lryer, to re h shows a	is include p red sludge educe the v a belt press	blate press is not auto vater conte s de-water	es, rotary ogenous, ent and m ing plant.	y vacuum fil further dryii ninimise the	lters, cen ng is norr supplem	trifuges and mally accom nentary fuel r	filter pres plished, t equireme	sses. by means of ents. This			
		The efflue would be the sewage tra- controlled	nt from de o return it eatment p waters.	e-watering to the sew lant will rer	cannot be /age treatr main able	discharg nent worl to meet it	ed directly v ks. The Op ts statutory	without tr erator sh obligatioi	eatment. The nould ensure ns in respect	ne usual p that in do t of disch	practice bing so, the arges to			
		Where tre by lamella alternative of odour c	atment is settlemer ly by filtra ontrol add	required so nt or simila ition. Solid litives may	olids remo r (see Teo ls should b be require	val may b hnical Gu be returne ed prior to	be achieved uidance Not ed to the inco preturn to the	using flo e on efflu inerator f ne STW.	occulents and uent treatme feed. Neutra	d efficien nt Ref. 1 alisation a	t settlement 1) or and the use			
	Odour is caused mainly by long dwell times in the sewerage systems before the sewage arrives a treatment plant. It arises in tank areas, from sludge de-watering; from filtrate treatment, and from conveyor systems and should be controlled by:													
		 self-clo all han 	osing door dling and ferably be	s should b de-waterin	e provideo Ig areas, a s a source	to all ac and conve	cess points eying syster	; ns shoul ion air:	ld be ventilat	ed with t	ne extracted			
		 during separa 	shut-dow	n, particula n; and	arly where	there is c	only one fur	nace, air	should be e	xtracted	∕ia a			
		 extract 	ed air, wh	nich is not i	ncinerated	l, should	be treated (see Sect	tion 2.3.13 -	Odour).				

INTRODUC	CTION TEC	CHNIQUES	F	MISSIO	NS		MPAC	T			
Management Ma	terials Activities/	Ground Waste	Energy		Noise	Monitoring	Closure	Installation			
in	puts abatement	water	Energy	Accidents	NOISC	worntoring	Closure	issues			
Incoming waste handling	2.3.1.8 Drum in Drums should be un the delivery note and received as part of r chemicals being tran of the drum supplier system is being rigo	ncineration man nloaded to an inter d the number of dr regular consignme nsported in the dru r. However, Opera prously operated by	agement mediate re- ums and th nts from m ms, it is no tors should tors should the drum	ception area neir contents ajor chemica ormal practic d carry out s suppliers.	where the secorde al compa se to plac pot chec	he consignm d. Because nies with littl e reliance or ks to verify tl	ent is ver most dru e varianc n the qual nat the de	ified against ms are e in the lity system slivery note			
	Where the furnace cannot handle halogenated substances these must be identified and rejected. Where the furnace has that capability it may be advantageous, economically and from the point of view of minimising the use of energy, to operate the furnace at the higher temperatures (see Section 2.3.4 - furnace requirements) only when burning chlorinated wastes. Under these circumstances a strict waste segregation scheme will be necessary.										
	Where contents cannot be verified the drums must be segregated and analysed to ensure that the plant has the capability and is authorised to burn them.										
	Authorisations shou they can be held prie	Ild limit both the quion to incineration.	antity of dr	ums to be h	eld and t	he maximun	n period fo	or which			
	 Residual chemicals will spill out of the drums if care is not taken. Procedures should be rigorously observed to ensure that: caps and lids are kept securely in place; drums are preferably stored vertically; and where stored horizontally the fill points are towards the top. 										
	Attention should be and water treatment	given to ensuring t arrangements are	he impervie adequate	ious nature (of the sto	orage areas a	and that tl	ne drainage			
	Drums have their lids removed prior to incineration. New incinerators should preferably be designed to enable the drums to be up-ended on to the conveyor and for all of the contents which drain out to of the drums to be carried into the incinerator. This has distinct advantages of requiring less fuel and minimises the need to dispose of drainings to landfill. The design must take into account the drainage apron details, the burner positions and the residence time in both primary and secondary chambers.										
	Where the incinerator is not designed to accommodate the drainings, the drums should be drained of excess residual content and the drainings should be transferred to a purpose-built chemical incinerator. The Operator should justify, to the satisfaction of the Regulator, any other form of disposal. A fixed drainage station should be provided; a loose container may be accidentally knocked over and discharge into the drains. Ensuring good drainage from the drums avoids chemicals dripping on to the floor at the entry to the furnace where conditions for combustion are not ideal.										
	To control odour and release of VOCs, drums should be opened and de-headed in an enclosed area with extraction to the furnace or to other odour control devices - (see Section 2.3.13 - Odour). The tim between opening and incineration should be kept to the minimum consistent with ensuring that the drums are adequately drained.										
	2.3.1.9 Wasten	nanagement - p	yrolysis a	and gasific	cation i	nstallation	s				
	The handling require appropriate waste ty	ed will depend upc ype specific guidar	n the type	of waste tha d in this Sec	at is being tion <mark>(2.3</mark> .	g treated and 1).	l should f	ollow the			
	Pyrolysis and gasific covered by the WID (see 2.3.1.1).	cation installations and will therefore	that subse also be ree	equently burn quired to me	n the pro-	ducts of thes andards outli	e procest ned in the	ses are e directive			
	Those processes the specification. It is the careful selection and	at are operating han nerefore likely that d/or waste pre-trea	ave genera heterogen itment prio	lly required eous waste r to charging	waste to streams g to such	fall within a (e.g. municip systems.	well-defin al waste)	ed) will need			
	2.3.1.10 Wasten	management at	refuse de	erived fuel	installa	tions					
	This sub-section is targeted at dedicated RDF installations. The handling required will depend the type of RDF and should follow the appropriate waste type specific guidance outlined in this (2.3.1) above (e.g. SLF has similar hazards to some chemical wastes, municipal RDF has sim hazards to municipal waste). Unless specifically exempted, all RDF plants will be required to standards required by WID (see Section 2.3.1.1).										
	Specific guidance in poultry litter, wood, s	n respect of issues straw) can be foun	arising froi d in earlier	m the handli Agency gui	ng of ind dance (S	ividual types 2 1.05) and	of RDF (should be	e.g. tyres, e consulted.			

INTRODU	JCTION TECHNIQUES EMISSIONS IMPACT										
Management	VaterialsActivities/ abatementGround waterWasteEnergyAccidentsNoiseMonitoringClosureInstallation issues										
Incoming waste handling	It will however be necessary to ensure that the techniques selected comply with WID and in particular the following basic principles: • Sufficient information should be obtained prior to receipt of the waste onto the site to ensure that it										
	 can be either: safely offloaded for safe storage and further characterisation prior to incineration or pre- treatment or incinerated without need for further characterisation (i.e. the description is already sufficiently) 										
	 detailed given the nature and source of the waste and the intended treatment) the RDF should be stored and transferred such that it cannot escape from control or give rise to potentially polluting emissions or contamination. This should take in to account the physical and chemical nature of the RDF concerned. 										
	 It will generally be BAT to store RDF under cover, or in purpose built containment in areas of contained or controlled drainage, with suitable fire protection systems. 										
	 It will generally be BAT to handle and transfer the RDF using sealed systems to prevent its escape. Odour control may be important for some RDF types. Guidance on odour control for municipal waste installations should be taken into account, (Section 2.3.1.3). 										
	RDF management techniques at co-incineration plants whose prime purpose is the production of energy or material products (e.g. cement works or power stations that are defined as co-incinerators under WID) should take account of the guidance given above and that found in the relevant sector guidance (e.g. cement and lime). The aim should be to handle RDF (or other wastes) at these co-incineration installations in a manner which fulfils the requirements of the WID and compliments the BAT for general raw materials handling in that sector as outlined in the relevant sector guidance.										
	2.3.1.11 Waste management at Co-incinerators										
	Specific techniques for waste management at co-incineration plant are not detailed in this guidance. Guidance on handling of incoming raw materials and fuels detailed in the sector specific guidance for the particular industry concerned should be consulted (e.g. cement and lime sector guidance). It will be necessary for the techniques selected to satisfy the requirements of HWID and WID – the requirements of these directives are discussed in Section 2.3.1.1.										
	As the directive requirements of co-incineration plants may be different to incineration plants (and there is a need to ensure the installation is correctly defined) the Directives themselves and relevant Agency guidance on their application to co-incineration should be consulted and discussed with the Regulator at an early stage.										

INTRODUC	CTION	TEC	HNIQ	UES	EI	MISSIO	NS		MPAC	T			
Management Ma	terials Act	tivities/	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation			
			Water							135003			
waste charging	2.3.2	Waste	chargir	ng									
	2.3.2.1	Legislat	ive oblig	ations ar	nd gene	ric requir	ements	for waste	chargin	g			
	The follow represent	/ing paragi BAT at oth	raphs are a ner plants:	applicable	to all inst	allations co	vered by	WID /HWID) and is al	so likely to			
	The HWIE) and WID	have sligh	ntly differer	nt wordin	g but essen	tially say	that:					
	"Incinerati feed:	ion and co	-incineratio	on plants s	hall have	and operat	te an aut	omatic syste	em to prev	ent waste			
	(a) at sta	nrt-up, until	the [requi	red] tempe	erature	.has been r	eached;						
	(b) wher	never the [irequired] t	emperatur	e is no	ot maintaine	ed;						
	(c) when distur	ever the co bances or	ontinuous failures of	monitors . the purific	show ti ation dev	hat any emi vices."	ssion lim	it value is ex	ceeded c	lue to			
	Waste cha place whe open or w	Vaste charging must therefore be interlocked with furnace conditions so that charging cannot take lace when the temperatures and air-flows are inadequate or when any flue gas cleaning bypasses are open or where the continuous monitors show that the emission limit values are being exceeded.											
	Designs s system sh escape of	hould endo hould be ca fumes or o	eavour to r apable of re excess air	make the c esponding flows.	charging to chang	operation as jes in furnac	s airtight ce pressi	as possible ure during ch	and the fanarging, to	an control avoid			
	In systems that loads directives.	s that use the chute	a waste fill should be	led chargir interlocke	ng chute d to prev	or hopper to ent loading	o achieve under th	e an airtight s e conditions	seal, the r outlined l	nechanism ɔy the			
	Where con other than Operator s restored.	ntinuously disturban should act This shou	monitored ces or failu to correct ld include	emission ures of the the situation considerat	limit valu purificati on as soc ion of wa	es are exce ion devices on as praction iste chargin	eded for e.g. pool cable unf g feed ra	any reason r combustior til normal ope tes.	(including control), erations c	g reasons the an be			
	Chargin	g Rates											
	Charging performar declared i paid to the Operators should alto whilst ens	rates outsince. The c n the appli e procedur s should re er mass th uring that	de the inst apacity wil cation and es that are cord through roughput r waste resi	tallation de Il vary acco I a firing dia designed ghput rates rates in orco dence in th	esign cap ording to agram in to ensur s and not der to ensur the chamb	acity seriou the CV of th cluded. At the that the d the exceed that sure optimu our is sufficie	sly unde ne waste all install esign cha at declare m combu ent to se	rmine enviro fed. The de ations close arging rate is ed in the app ustion conditi cure ash bur	nmental esign shou attention s not exce lication. ions are a mout requ	uld be should be æded. Operators achieved, uirements.			
	2.3.2.2	Chemica	al waste										
Chemical waste charging	Most liquid to ensure within the	d chemical good mixir flash point	ls are intro ng and ato t of the liqu	duced to tl misation. ıid.	he furnac Heating	e by conve to control vi	ntional lio scosity s	quid fuel bur hould be lim	ners. It is ited to be	s essential ing well			
	Where wh handling r shredding the key to	ole drums nechanism is an optic satisfacto	of waste a and must on it is mor ry operatio	are fed dire t be design re usually f on. Particu	ectly, the ned to wit found tha ilar attent	incinerator hstand any it careful pa tion should l	will have resulting ckaging be paid t	to be fitted increase in and schedul o this aspect	with a sui pressure ing of the t.	table . While charges is			
	A well ma pallets) to which may	naged stor be fed to t y be difficu	rage area, the inciner Ilt to incine	with detail ator such t rate (e.g. l	ed labell that they lodine).	ing and inve do not give	entories v rise to e	will assist in i xcessive loa	making u ds of sub	ວ loads (e.g. stances			
	Sealed de combustic necessary	elivery char on escapin / to preven	mbers sho g from the it reactions	uld be use feed mech in the fee	d where nanism. der syste	there is a ris Positive pre em. Purge (sk of eith ssure ine gases sh	er waste or ert gas blank ould be fed t	products teting maging to the inci	of y be nerator.			
	2.3.2.3	Municip	al waste	and clini	ical was	ste							
Municipal and clinical waste charging	Normal feasions for the should be include a l	ed mechar engineere low-level a	nisms for s ed to preve llarm in the	olid waste nt back flo e feed hop	s include w of com per.	e ram, gravit ibustion pro	y and ho ducts th	opper feeds. rough the wa	The arra	ngements and should			

INTRODUC	TION T	ECHNIQ	UES	E	MISSIO	NS		MPAC	т
Management Mai in	terials <mark>Activitie</mark> puts <mark>abateme</mark>	s/ Ground nt water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Waste charging	Doors isolate th should be made such as double	e furnace from for preventing doors and/or a	n the feed c g the ignitio a cooling sy	hute to p on of the vstem. W	prevent the fi waste which Vater-cooled	ire burnir is in cor chutes a	ng back up t ntact with the are currently	he chute. e outside [,] in use.	Provision of the door,
	The provision of the grate steadi the control of th	f consistent fee ly draws the w e combustion	ed is crucia aste on to i is needed (Il to ensu it are pre see Sec	ring steady ferred. Whe tion 2.3.4 - F	combust ere feed ^T urnace I	ion condition is intermitter requirement	ns. Syste nt, particu <mark>s)</mark> .	ms in which lar care in
	With moving gra of the furnace w maintain good c and may allow f	ate systems it i vill be prevente combustion con or manual inte	is particular ed. There s nditions. T ervention by	rly impor should be his shoul / trained	tant that ope an automa Id be control operators if	erating pr tic mean led by m required	ocedures sh s to vary the easured cor	now how o waste fe nbustion	overloading ed rate to parameters
	2.3.2.4 Sew	age sludge							
Sewage sludge waste charging	Because of the problems. The from plate press combustion con Very wet sludge used in the sew gas loading.	homogeneous degree of pre- ses may give r ditions. The w s will have a le age works ma	a nature of s dewatering ise to hand vater conte ow CV and y contain c	sludge, it y will imp ling and nt of the may lea hlorine a	s injection to act upon the charging dif sludge may d to combus nd sulphur r	the furr feed me ficulties t assist in tion prot nolecule	ace usually echanism us hat lead to l suppression plems. The s that will ex	causes fe ed. Very ess stable n of NOx settlemer ert an ad	ew dry cakes e production. It chemicals ditional acid
	2.3.2.5 Drui	n incineratio	on						
Drum charging	Drums should b value fed to the	e fed to the fui	rnace in se as constan	quence, t as poss	according to sible.	their co	ntents, to er	nsure that	the calorific
	Inevitably there the furnace entr to prevent the b	will be some c y. The apron uild up of com	lrips from th section sho bustible ma	he drums ould effec aterial, (s	s as the feed ctively conta see also Sec	l convey in liquid a tion 2.3.	or takes thei and should I 1 - Waste ha	m across pe regular andling).	the apron to ly cleaned
	Because the fur achieve the requ operating at hig prevent fugitive drum loading. V exit; they also k drums.	nace must be uired operating h excess air le emissions. Do Vater curtains eep dust and a	open to all g temperatu evels. Arran oors and in are effectiv ash burning	ow the d ures whil ngement terlockin ve at min g around	rums to pase st coping wit s for feeding g are not pra imising "puff the entry to	s in (and th consid the drur acticable fing" rele a minimu	out) the inci erable ingre ns should be because of ases from th um and prov	nerator h ess of colo e enginee the high t ne feed er ride a que	as to I air and red to frequency of ntry and the ench for the
	There should be in the furnace d	e no drums in t o not meet tho	the furnace ose given in	on start Section	-up and feed 2.3.4 - Furn	l should ace requ	be stopped v lirements.	when the	conditions
	2.3.2.6 Anin	nal carcasse	es						
Animal carcass charging	Feed is intermiti carcasses; as a as above, is of o usually limited, a the doors should	tent, by front lo result air will l critical importa and also becau d be interlocke	bader vehic be drawn ir nce. Beca use regulat ed with the	ele, ram f nto the fu use of th ion (by th draught.	eed or manu rnace when is and becau ne Regulator	ual. The waste is use the p r) of feed	doors have charged an rotection of control is d	to open to d control abateme ifficult, the	o admit the of the fans, nt plant is e opening of
	2.3.2.7 Pyrc	olysis and ga	asificatio	n plants	5				
	Pyrolysis and ga Packed rams ar maintain a good mechanisms mu corrosion that c	asification plar nd screw feede l seal, as may ust be of appro ould lead to es	nts require of ers are suita shredding opriate desi scape of the	careful ca able for u waste to gn and n e gases p	ontrol of air i use. Feed th prevent men naterials to r produced.	ngress d nroats tha chanical resist the	luring waste at reduce in blockages. reactor bac	charging diameter Charging k pressur	operations. may help to es and
	WID (HWID) red standards may a the starting poir	quirements mu also represent it for site speci	ist be fulfille BAT for in ific determi	ed at inst stallation nation of	allations cov s not falling BAT.	vered by within th	the directive e directives	e. Directiv and shou	/e ld represent
	2.3.2.8 Refi	ise derived i	fuels						
	It will generally l control of air fec	be BAT to cha I to the furnace	rge the RD e.	F using	sealed syste	ms to pr	event its eso	ape and	allow
	WID (HWID) red	quirements mu	ist be fulfille	ed.					

INTRODUCTION TEC			CHNIQ	UES	EI	VISSIO	NS		MPAC	T
Management	Material inputs	S Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Waste charging Co-incinerator charging	2.3 Spe det cen req 2.3	.2.9 Co-inc ecific technique ailed in the sec nent and lime s uirements of HV 1.1.	<i>inerators</i> s for chargi tor specific ector guida WID and W	ng at co-in guidance f nce). It wil ID – the re	cineratior or the pai Il be nece quiremen	n plant are r rticular indu ssary for th ts of these	not detaile stry conc e techniq directives	ed in this gui erned shoul ues selecte are discuss	idance. (d be cons d to satist sed in Se	Guidance sulted (e.g. fy the ction

INTROD	N TE	CHNIQ	UES	EMISSIONS			IMPACT			
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Furnace 2.3.3 Furnace types

2.3.3.1 Introduction

"Furnaces" has been divided into 2 sections; **this section describes the furnace designs available** while Section 2.3.4 deals with the design and performance requirements for BAT. More detailed descriptions of furnace types may be found in the literature.

RDF and Co-incineration

Refuse derived fuel does not have its own sub-section here because it is a *waste type* rather than a specific furnace design. RDF may be burned in a variety of plant providing that plant is designed to receive fuels of similar physical, chemical and combustion characteristics e.g. where a high CV liquid RDF replaces a high CV liquid primary fuel.

Co-incineration furnace design is also not included here - refer to sector specific guidance.

Table 2-3 - Summary of Combustion Technology Application

Combustion Technologies			Waste	е Туре		
 known to be used in the UK: UK suitable or likely to be: S 	Chemical	Clinical	RDF (note 4)	Municipal	Sewage sludge	Animal carcass
Fixed hearth						UK
Fixed stepped hearth		UK		UK		
Moving hearth (normally sloping and stepped)		UK	UK	UK		
Pulsed hearth	S (Note 1)	UK		S		S
Rotary kiln	UK	UK	S?	S	S	S
Fluidised bed (note 2)			UK	S	UK	
Liquid injection	UK		S?		S	
Semi pyrolytic	UK	UK	S	S		S
Gasification (note 2)	S	S	S	S	S	S
Pyrolysis (note 2)	S	S	UK	UK	S	S
Cyclonic combustors						
Gas incinerators	UK					
Drum incinerators	UK					

Notes: 1. For mainly solid chemical wastes.

2. May only be suitable for selected / pre-treated waste fractions

3. RDF suitability will depend on the individual nature of the RDF concerned.

2.3.3.2 Fixed hearth incinerators

While these have been used for clinical waste and even chemical waste they would now normally only be acceptable for the incineration of consistent wastes whose combustion has a low relative pollution potential. They are in use for animal carcass incineration where the containment offered by the fixed hearth may assist in ensuring unburned liquids (e.g. fat) do not leak out.

The primary chamber waste is normally over-fired with primary air. Support burners are often required. Proper mixing of waste on the hearth is difficult and requires careful adjustment of the feed and ash removal rates. Achieving consistent burnout is difficult. The skill and training of the Operator are particularly important. Such designs may have difficulty in meeting WID standards, mainly due to the semi-batch nature of the waste travel on the grate and de-ashing operations.

A secondary chamber with injection of supplementary fuel and secondary air is essential.

INTROD	N TEC	CHNIQ	UES	EMISSIONS			IMPACT			
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Furnace	2.3.3.3 Fixed hearth stepped furnace incinerators
types	Lead in particular for CIAU these comprises a series of stars (hypically, 2) with embedded primery sin

Used, in particular, for *CWI* these comprise a series of steps (typically 3) with embedded primary air channels, down which the waste is moved by a series of rams. The first step is a drying stage, with sub stoichiometric oxygen conditions, during which most volatile compounds are released and burn above the grate in the combustion chamber. The remaining, less volatile material is pushed onto the next step where the main combustion takes place. The third step is the burnout stage before the ash is discharged into a final ash burnout chamber. The material can typically spend 8 hours on the hearths and a further 8 hours in the burnout chamber to achieve burnout performance of less than 1% carbon in ash. Throughput rates should be altered to ensure good burnout.

The steps between hearths provide agitation as the waste tumbles down the step; however, this can also produce surges of unburned particulate and hydrocarbons and consequently the provision for good secondary combustion and residence time becomes more important.

Gas combustion takes place in the primary chamber and in a subsequent secondary combustion chamber.

2.3.3.4 Moving grate incinerators

Municipal waste is the main application for these incinerators which can be designed to handle large volumes of waste. Older grates such as horizontal (Continuous Travelling Grate Stoker), W-Grate or oscillating grate have now been largely replaced by either:

- The roller grate, which consists of five to eight adjacent drum or roller grates, located in a stepped formation to form an inclined surface with the drums rotating in the direction of waste movement. Combustion control is enhanced by independent air feed to each roller;
- The stepped inclined grate which uses moving bars, rockers or vibration to move the waste down each of the grates (normally three). A separate drive and air supply is provided to each grate to provide the different conditions required for drying, combustion and burnout as described for the fixed stepped hearth above. In some cases the air supply is further subdivided to give better control.
- **Inclined counter-rotating grates** in which the grate bars rotate backwards provide good waste agitation whilst preventing waste from tumbling down the forward inclined grate.
- Some designs incorporate a rotary section to provide additional agitation and enhance burnout.

In larger furnaces it is possible for the required residence time to be achieved in a single chamber but this may be more difficult to verify (see Section 2.3.4.4 - Validation of combustion conditions).

2.3.3.5 Pulsed hearth incinerators

The pulsed hearth incinerator uses the pulsed movement of one or more refractory hearths to move the waste and ash through the incinerator. The hearths, which are stepped at each side to form a U shape, are suspended from four external supports. The pulsing action of the hearth is achieved pneumatically. They have been used for municipal, clinical, animal carcasses and other solid wastes.

The smooth hearth can handle difficult wastes with reduced risk of jamming or loss of liquid wastes. There are no moving mechanical parts exposed to burning material or hot gases. The main difficulties, have been in achieving effective and reliable burnout of the solid wastes. For clinical waste it should be noted that the burnout time on the hearth has been short in installations to date when compared with stepped hearth incinerators, (see Section 2.5 for details on ash handling).

2.3.3.6 Rotary kiln incinerators

Rotary kilns have wide application and can be of the complete rotation or partial rotation type. They have the benefit of good waste agitation and achieve good burnout provided waste residence time in the furnace is adequate. They can be used in combination with other designs to provide additional ash burnout.

Incineration in a rotary kiln is normally a two-stage process consisting of a kiln and a separate secondary combustion chamber. The kiln itself, which is the primary combustion chamber, is a cylindrical shell lined with a refractory. The kiln is inclined downwards from the feed end and rotates slowly about its cylindrical axis. The rotation moves the waste through the kiln with a tumbling action so as to expose fresh surfaces to heat and oxygen. Structures within the kiln may be added to increase turbulence and to slow the passage of liquid wastes. Residence time of the solids going through the kiln may be changed by adjusting the kiln's rotational speed.Due to the absence of exposed metal surfaces, rotary kilns are normally able to operate at high temperatures and often operate in a slagging mode. Slagging kilns operate at a temperature high enough to melt inorganic waste and produce a fused glassy slag which is low in organics and has a low leaching rate. This has

INTRODUCTION			CHNIQ	UES	EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Furnace types

made them particularly suitable for *hazardous waste* incineration where whole drums and solid wastes can be completely destroyed. The slagging temperature may be reduced and the slag's viscosity controlled by additives. Molten slag is drained from the kiln and solidified by quenching in a water bath. In other cases the ash is collected dry.

Careful attention needs to be paid to the seals between the rotating kiln and the end plates to prevent leakage of gases and unburnt waste. Tumbling of the waste may generate fine particles requiring secondary combustion and good particulate abatement.

2.3.3.7 Fuidised bed incinerators

Fluidised beds are only suitable for reasonably homogeneous materials and are therefore the main designs for the incineration of *sewage sludge*. Fluidised bed technology may also be used for any waste that has been sufficiently treated, including treated *municipal waste* and refuse derived fuels.

Most fluidised beds fall into 2 categories - circulating (CFB) and bubbling (BFB). A fluidised bed incinerator is normally a single stage process. It consists of a refractory-lined shell. The chamber contains a granular bed consisting of an inert material such as sand or limestone. This bed is supported on a distribution plate and fluidised by air or other gas being blown up through the plate. Ancillary equipment includes a fuel burner, a waste feed mechanism and possibly an afterburner chamber to provide adequate residence time..

The CFB is based on the same principle as BFBs. The difference is the fluidisation velocity. In a CFB the high airflow creates greater mixing of the air and fuel. Particles are carried out of the vertical combustion chamber by the flue gas and are removed in an integral cyclone. The solids from the cyclone, the vast majority of which are sand, are returned to the fluid bed.

The high fluidisation velocities may result in carry over of fine particulate matter However, modern dust collection equipment can be expected to handle this.

The advantages of a fluidised bed incinerator include:

- combustion efficiency is high and temperatures are uniform, making residence time calculations more reliable;
- lower temperature leads to low NO_X;
- simple furnace no moving parts; and
- the sand provides continuous attrition of the burning material removing the layer of char as it forms and exposing fresh material for combustion. This assists with both the rate of combustion and burnout.

An *MWI* fluidised bed is operational in the UK. The waste is pre-sized by means of a crusher / shredder. There have been operational problems with the waste pre-treatment stages that have lead to significant down time.

2.3.3.8 Starved air (semi-pyrolytic) incinerators

See also gasification (Section 2.3.3.13). More a method of control than a specific configuration, the concept can be applied to various designs. The primary chamber operates at sub-stoichiometric air levels to evolve a gas that is combusted in a secondary zone operating under excess air conditions. A supplementary fuel burner is required in the secondary zone to ensure the required combustion conditions are maintained at all times. The design of the secondary combustion zone and support burner will need to consider the full range of characteristics of the gas evolved to ensure that unburned gas is not released.

The advantages can be a more controlled burn leading to lower releases of NOx, VOCs and CO. The relatively low combustion airflow results in low entrainment of particulate in the flue gas.

2.3.3.9 Cyclonic combustion

Cyclonic combustion techniques, for the destruction of organic wastes, were reported upon as an emerging technique in earlier Agency guidance on incineration (ref.: S2 5.01). The technique was reported to be characterised by reasonably high temperatures and very good mixing, achieving high destruction rates at short residence times e.g. 99.99999% destruction at 1200 °C for 0.25 seconds. The technique was reported to produce a vitrified slag. It is not known whether commercial applications have been developed.

INTROD	N TEC	CHNIQ	UES	EMISSIONS			IMPACT			
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Furnace	2.3.3.10 Liquid injection incinerators
types	Liquid intertion incidentation and covered by WID
	Liquid injection incinerators are covered by WID.

Liquid injection incinerators, which are usually refractory lined cylindrical chambers, are used for the incineration of chemical waste. The furnace is often divided into several chambers or zones. In the first, highly combustible liquids such as waste oils and solvents are burned. Aqueous mixtures containing combustible solids and low calorific value wastes may be injected into the secondary zones.

A secondary combustion chamber is sometimes used, but is not always necessary if the combustion conditions in the primary chamber provide the required temperatures and residence time.

2.3.3.11 Gas incinerators

The incineration of waste gases (unless the gas arises from the thermal treatment of waste) are not covered by WID. Because of the homogeneous and generally very clean nature of the feedstock it is likely that very low emission levels can be achieved using BAT. Emissions should be below those outlined in WID.

Whether considered to be a piece of abatement plant or an incinerator in its own right, these devices fall into the categories of:

- conventional flame incineration at around 850-950 °C;
- catalytic incineration at around 350-400 °C; and
- flameless (gas passes through a hot granular bed) at around 850-1200 °C.

Devices are available to cover airflows from 1 to $30,000 \text{ m}^3$ /h and have been banked together in units up to $400,000 \text{ m}^3$ /h. VOC destruction rates can be typically 99.99% but can be as high as 99.9999%, where required, by choice of the appropriate system. Such destruction rates can be achieved with residence times less than 0.2 s and with virtually no NO_X production. The devices can be heated electrically or by injection of fuel gas to the feed stream but are normally self-supporting, as long as the VOC content of the feed is at least 0.5 to 1.0 g/m^3 . While it is generally beneficial to recuperate the heat in the bed, systems which reverse flow through the bed to recover thermal energy, will not normally achieve the very high destruction rates because small quantities of gas bypass when the changeover valves are operated. Flameless devices are not generally suitable where there is a significant particulate burden.

2.3.3.12 Drum incinerators

Drum incinerators are all of the same basic design, comprising a conveyor system which takes the inverted drum through a long narrow furnace where the burners, normally gas fired, burn out the residual contents and burn off, or at least loosen, the paint. The gases pass to a secondary chamber where further burners ensure effective combustion.

2.3.3.13 Gasification installations

This section provides a basic summary of the gasification process. More detailed description of individual types of gasification plants may be found in the BAT report on waste pyrolysis and gasification activities.

Gasification is the conversion of a solid or liquid feedstock into a gas by partial oxidation under the application of heat. Partial oxidation is achieved by restricting the supply of oxidant, normally air. For organic based feedstock, such as most wastes, the resultant gas is typically a mixture of carbon monoxide, carbon dioxide, hydrogen, methane, water, nitrogen and small amounts of higher hydrocarbons. The gas has a relatively low CV, typically 4 to 10 MJ/Nm3, and may be highly corrosive and toxic owing to the partially reduced species present. Particular attention must therefore be paid to ensuring the gas produced in the reactor and passed to the combustion stage is contained. This should include attention to ensure:

- Consistent waste feed characteristics are obtained (pre-treatment is likely to be required for heterogeneous wastes) to ensure even reaction rates and internal plant pressures
- The plant must be sealed to contain the gases produced
- The materials of construction must be able to withstand the highly corrosive environments which they will be subjected to.

Air is the normal oxidant, although oxygen enriched air or oxygen can be used. The gas CV will be higher (10 to 15 MJ/Nm3) when air is not used due to the absence of nitrogen.

For most waste feedstock, the gas produced will contain tars and particulate. Depending upon the combustion technology selected the gas may require cleaning before combustion.

INTRODUCTION TE		N TE	CHNIQ	UES	EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Furnace types

Where the gas (or solid or liquid products) produced are subsequently combusted, the installation will be required to meet the standards of the WID.

There are four main categories of gasification plant:

- 1. *Rotating kiln* similar to rotary kin incinerators except that they operate under sub-stoichiometric conditions.
- 2. **Fixed bed reactors** waste is loaded through the top of a shaft chamber. Oxidant is added through a grate or tuyeres at the base of the shaft. In *updraft* versions the fuel gas is removed from the top of the shaft, in *downdraft* versions from the base.
- 3. *Fluidised bed reactors* similar to incineration but at sub-stoichiometric conditions. Such systems can be operated at high pressure as part of a gasification combined cycle to supply fuel gas to a gas turbine.
- 4. **Other systems** are reported to be available but there is little information on design and performance.

2.3.3.14 Pyrolysis installations

This section provides a basic summary of the pyrolysis process. More detailed descriptions of individual types of pyrolysis plants may be found in the BAT report on waste pyrolysis and gasification activities.

Pyrolysis is the thermal degradation of a material in the complete absence of an oxidising agent. Most pyrolysis plants have an externally heated chamber that is sealed to prevent air ingress. In practice, complete elimination of air is very difficult and some oxidation may occur.

The process generally occurs in the range of 400 to 1000 Celsius. The heat breaks complex molecules into simpler ones which form gas, liquid or solid (char). The relative proportions depend upon the temperature, the nature of the waste feed and the time it is exposed to that temperature. Long exposures to lower temperatures (400 to 500 Celsius) maximise char productions (e.g. charcoal production). Short exposures (<1 second) to high temperatures (500 to 1000 °C) are known as "flash pyrolysis" and increase the proportion of gas or liquid.

There are four main types of pyrolysis installations:

- 1. **Externally heated rotating kiln** the waste is degraded in an externally heated rotating drum. The heat is obtained from combustion of a proportion of the fuel gas. The gas produced is cleaned using scrubbers or filters before combustion.
- 2. *Heated tube systems* similar to the rotating kiln except that the waste is pushed through a static heated tube by a screw feed or ram.
- 3. *Fluidised bed system* have higher heat transfer rates so can be used for flash pyrolysis giving high liquid yields and often used for wood or similar shredded materials. Heat may be added by external heat transfer or directly from a combustion process.
- 4. Other systems these are generally based on externally heated static kilns

If gas is the principle product, it is likely to have a CV range of 15 to 30 MJ/Nm³ (cf. CV on natural gas of approx. 39 MJ/Nm³). This gas can be burned in boilers, engines or gas turbines. The raw gas will contain highly toxic and corrosive reduced species. Similar considerations to those outlined above for gasification plant apply.

If a liquid (pyrolysis oil) is required it is necessary to stop the reactions that would otherwise go on to produce a gas, typically by condensation. Pyrolysis oil is a complex mixture of organic compounds. Some of these oils may have value as a product when certain selected feedstock are pyrolysed. Where waste feedstock are concerned, the main uses are likely to be as a liquid fuel

Pyrolysis processes produce two main types of solid residues:

- Ash from inert solid material present in the waste e.g. glass, stones etc
- Carbon char which may be used as a product (e.g. carbon black) or burned as a fuel or disposed of as a waste residue.

Where the char is disposed of (rather than burned as a fuel or used as a product) the WID 3% TOC limit will apply. Although the char may consist of primarily elemental carbon (which is not included in the TOC test), it is still possible that the 3% TOC level will be exceeded. Where this is the case BAT will require further processing of the char such that the residues ultimately produced for disposal meet the 3% TOC standard. This additional processing may involve the use of a water gas reactor or combustion.

INTRODUCTION		N TEO	CHNIQ	UES	EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Furnace types This subsequent combustion of a product of pyrolysis (whether the solid, liquid or gaseous product) will mean that the WID will apply to the installation. Guidance on its application can be found throughout this document, but it is advised that early discussion with the Regulator takes place to clarify the application of the directive to the individual pyrolysis installation.

If the waste feed to the pyrolysis installation is (or can be treated to be) sufficiently heterogeneous, and the gas produced either is (or can be treated to be) of good quality, it is theoretically possible that high levels of electricity generation may be achieved through the use of gas engines or gas turbines. At present there is mixed evidence regarding the ability of such systems to meet the stringent requirements of WID in respect of emissions to air. However, Operators proposing pyrolysis should justify their selected power generation technology and explore opportunities to produce clean, high quality fuel gases that may be burned in higher energy efficiency plant (i.e. those that do not depend upon a steam cycle) and can comply with WID emission limit values.

INTRODUCTION		N TE	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

Furnace	
requirements	

2.3.4 Furnace requirements

Good furnace design and combustion control are the most important factors for ensuring wastes are effectively destroyed and that the production of potentially harmful emissions are minimised. This section provides guidance on the techniques that represent BAT at each stage of the process.

2.3.4.1 Legislative obligations

Obligations

European directives already set <u>minimum</u> standards for operating conditions of municipal and hazardous waste incinerators. The new Waste Incineration Directive (2000/76/EC) extends the scope of these requirements to the majority of waste incinerators and other co-incinerators that burn wastes or fuels derived from wastes.

The following summarises the main combustion related requirements arising from European legislation:

- The gases resulting from the combustion of non-hazardous wastes must be maintained at above 850 °C for at least 2 seconds.
- The gases resulting from the combustion of hazardous wastes with a halogen content greater than 1% (as chlorine) must be maintained at above 1100 °C for at least 2 seconds.
- There should be at least 6% oxygen at installations subject to MWID and HWID (this is required of MWIs and HWIs only and only until 28 December 2005 (or from 28 December 2002 for new installations) when they become regulated under WID. WID does not specify oxygen concentrations it should however be noted that BAT will require sufficiently oxidising conditions at the final combustion stage to provide for good combustion. In many situations BAT may equate to the 6% oxygen requirement unless the operator can demonstrate otherwise.
- Incinerators must be provided with auxiliary burners to achieve and maintain the required temperatures. This does not apply to co-incineration unless BAT requires them.
- The combustion temperature and residence time, and the oxygen content of the stack gases, must be <u>validated</u> at least once, and under the most unfavourable operational conditions (see Section 2.3.4.4 Validation of combustion conditions).
- Residues (ash) shall be minimised in their amount and harmfulness.
- Incinerator slag and bottom ashes shall not exceed 3% TOC or 5% LOI (dry weight). This does not
 apply to co-incineration where BAT for the sector will apply (see sector specific guidance).
- Installations should not give rise to significant ground level air pollution.

Whilst it is recognised that these requirements (along with the emission limit values noted elsewhere in this guidance) set high technological standards, Operators will still need to consider the use of techniques that may further reduce releases and demonstrate that their installation and its operation is the BAT.

2.3.4.2 Grates and primary air



- Primary air supply should be controllable, and where there are separate grates should be
- separately controllable between the grates.

Cont.

BAT for grates and primary air

INTRODUCTION TECHNIQUES						E	MISSIO	NS	IMPACT			
Management	Mat in	erials: puts	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	
Eurnace												
requirements		4.	For most desig controlled both entrainment of controlling thes	For most designs of furnace (fluidised beds may be an exception), primary air should be controlled both to minimise NO _X (Section 2.3.8.3) production and minimise velocities and the entrainment of particulate (Section 2.3.8.3). Starved air systems can be very effective in controlling these while maintaining low levels of CO (Section 2.3.8.3).								
		5.	Heavy metal of compounds. F inorganic mate	xides and Proper dist rial volatili	alkali meta ribution of a sed.	Il salts in air and fu	the fly ash lel, avoiding	arise fror 9 hot zon	n the volatilis es, will reduc	sation of i ce the am	norganic ount of	

6. Higher primary airflow through grates may be required to reduce temperatures. The use of watercooled grates may minimise the airflow requirements

2.3.4.3 Combustion chambers, secondary air system designs and supplementary burners

Application Form Question 2.3 (cont.) BAT for combustion chambers, secondary air system designs and supplementary burners is as follows:

With the Application the Operator should:

1. supply the general Application requirements for Section 2.3 on page 39 for this aspect of the activities.

Indicative BAT Requirements

- Combustion chambers, casings, ducts and ancillary equipment should be made, and maintained, as gas-tight as practicable. They should be maintained under slight reduced pressure and designed to prevent both the release of gases and disturbance of combustion conditions during waste charging. Control of the induced draft fan, primary air and the feed rate should be balanced.
- 2. Gas temperature in the primary zone and at the point of exit from the secondary combustion chamber should be continuously monitored and recorded, and audible and visual alarms should be triggered when the temperature falls below the minimum specified. The charging system (see also Section 2.3.2 Waste charging) should be interlocked with the validated combustion temperature to automatically prevent additional waste feed:
 - At start up, until the combustion temperature is reached
 - Whenever the relevant combustion temperature is not maintained
 - Whenever the continuos emission monitors show exceedences of the emission limit values (over the appropriate averaging period).
- 3. All incinerators should be fitted with a burner which automatically switches on if the temperature falls below the relevant temperature at any time when unburned waste is in the incinerator and to ensure that the temperature conditions are met prior to waste being admitted on start-up.

Supplementary burners and fuels

- 4. Supplementary burners must be provided at all incineration installations in order to secure and maintain the required combustion temperatures. Co-incineration plants are not required to include supplementary burners under EC legislation but they may be required by BAT for the particular sector (see relevant sector guidance)
 - the burners must be capable of supporting the combustion temperature under all conditions when there is waste in the furnace;
 - the burners may be used for initial start-up, temperature maintenance and final shut-down
 - the application should state the start up and shut down sequence, including the temperatures at which the waste will be introduced, and prevented, and at what temperature the supplementary burners will trigger
 - automated systems should be used trigger the supplementary burners and to prevent additional waste feed until the required temperature is re-established

Cont.

BAT for combustion chambers etc.

INTRODU	CTIC	ON TEC	HNIQ	UES	E	MISSIO	NS		MPAC	T		
Management M	aterials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Furnace requirements	5.	Supplementary levels no worse may be dictate	/ fuels use e than thos d by BAT.	d in such b se from bui	ourners s rning gas	hould be of s oil, as defi	a type w ned by D	/hich, in effec Directive 75/7	t, produc 16/EEC;	e release lower levels		
		 Supplementary They do no unburned w Combustion means that temperature any other full 	v fuels that t give rise vaste comp n temperat waste der es above t uels propo	are waste to emissio position ag tures are g rived fuels he minimu sed should	s or was ns worse ainst tha reater th cannot b m. I be the s	te derived n e than gas o t of gas oil a an those ou e used for s subject of tri	nay only iil (this m and cons itlined in start-up, als.	be used as s ay be judged sidering the e the table bel but may be u	support fu I by consi ffect on e ow – this sed for m	el if: dering the missions) effectively aaintaining		
BAT for combustion chambers etc. (cont.)	6.	 The criteria which govern efficient combustion of furnace gases are: adequate oxygen content to ensure complete combustion; sufficient temperature to promote combustion; sufficient time to complete the combustion reactions; and turbulence to promote mixing. 										
		Whether the at even tertiary ch legislation mus	oove criter nambers, is t, however	ia can be r s for the de r, be comp	net in a s esigner to lied with.	single cham o decide an . These are	bered fu d justify. summa	rnace, or will The require rised at point	require s ments of 7 below.	econdary or European		
	7.	All incineration plant should be equipped and operated in such a way that the temperature of the combustion gas is raised to that specified in Table 2.4, after the last injection of air, in a contro and homogeneous fashion and even under the most unfavourable conditions anticipated, for a least <i>two seconds.</i>										
		These <i>tempera</i> and for as long	These temperatures should be maintained during start-up and at the end of an incineration cycl and for as long as combustible waste is in the combustion chamber.									
		Oxygen levels stated and just the chosen oxy ensure oxidativ stage, (see Sec Other operatin	Oxygen levels should generally be in excess of 6%v/v or, if not the reason why not should be stated and justified (including consideration of the advantages and disadvantages of operating at the chosen oxygen level). It must still be demonstrated that oxygen levels will be adequate to ensure oxidative combustion and hence destruction of organic species at the final combustion stage, (see Section 2.3.4.5 regarding oxygen levels).									
		the satisfaction better techniqu	of the Re e.	gulator tha	t the rele	ease limits v	vill be me	et and the pro	ocess rep	resents a		
Table 2-4 -				Process				Minimum t	emperati	ure °C		
Furnace gas		Chemical was	ste					850	(1100)			
temperatures		Clinical waste	;					1000) (1100)			
		Municipal was	ste						850			
		Animal carcas	sses						850			
		Sewage Slud	ge						850			
		Gasification (combustio	n of produ	cts of ga	sification)			850			
		Pyrolysis (cor	nbustion o	of products	of pyroly	/sis)			850			
	1	Co-incineratio	n					850	(1100)			
	1	Refuse Derive	ed Fuels					850	(1100)			
		Drum recover	у					850	(1100)			
		Note 1: Figures (exprese Note 2: Minimu	s in bracket ssed as chlo um tempera	s apply whe orine) or wh ture is norm	re wastes ere other ally that n	containing n thermally res neasured nea	nore than istant sub ar the inne	1% halogenate stances are to er wall of the co	ed organic be inciner	substances ated. chamber.		

INTRODU		DN TE	CHNIQ	UES	E	MISSIO	IMPACT					
Management N	laterials	Activities/	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Furnace requirements	2.3.	4.4 Validat	ion of Co	mbustio	n Condi	tions						
	Euro incin serv	European legislation requires that combustion temperature and residence time (incinerators and co- ncinerators) are subjected to appropriate validation at least once when the plant is brought into service, under the most unfavourable operating conditions.										
	Bec ensi pos	Because it can be prohibitively expensive to retrofit existing plant it is vital that designers ensure that the residence time and temperature requirements are considered at the earliest possible stage.										
	Deta valid the B	Detailed consideration of the techniques that fulfil this requirement are contained in the BAT report on validation of combustion conditions. This guidance summarises some of the key issues and highlights the BAT.										
	App Que	blication Form estion 2.3 (con	t.)	BAT for	validatior	n of combus	tion con	ditions:				
	Wit	th the Appli	cation th	ne Opera	tor sho	ould:						
	1.	Supply the ge activities.	eneral Appli	cation req	uirements	s for Sectior	1 2.3 on p	bage 39 for t	his aspec	t of the		
	2.	Demonstrate guidance (Se demonstrate	that the de ction 2.3.4. this in the a	sign select 1) and in r application	ted can m elevant E is given i	neet the com Suropean leg n this sectio	nbustion gislation. on below.	requirement Guidance c	s outlined on how to	I in this		
	3.	Explain how t and the exhain	he applicar ust gas oxy	nt intends t gen conce	o validate	e the combu vhen the ins	stion res tallation	idence time is brought ir	and temp nto service	perature, e.		
	4.	Confirm the w be carried out	vorst case o t.	combustior	n conditio	ns under wł	nich the v	alidations o	utlined in	3 above will		
	Ind	icative BA1	Require	ements o	combus	stion vali	dation					
BAT for	1.	At the design	stage and	I for the ap	plication	Operators s	hould:					
assessing residence time and		 Use CFD to demonstrate that the residence time and temperature requirements will be met in the chosen design and to identify the ideal (or best practicable) locations for temperature monitoring for the purposes of validation measurements. 										
turbulence		Outline the representation	e assumption ative of the	ons and in chosen de	puts useo esign	I in the CFD) modelli	ng and expla	ain how th	iese are		
		Identify th residence	e qualifying time and te	g zone ove emperature	r which th e requirer	ne residence ments.	e time an	d temperatu	re will me	et the		
		 Use a model that is representative of the real flow situation in the qualifying zone (this is most likely to be a combination of plug flow and stirred reactor flow rather than one extreme) 										
	 Taking account of this guidance and the BAT report, confirm the details of the n be used to validate temperature and residence time modelling, including identifi worst case conditions under which the test(s) will be carried out. (Note: At a we plant the worst case conditions selected may equate to those at the extremes o firing diagram). 									hod that will tion of the operated ne plant		
		Identify th	e locations	for the ten	nperature	measurem	ents requ	uired to unde	ertake val	idation.		
	2.	Unless not pr should:	acticable (s	see 3 below	v) at the o	operational	stage th	ne validation	techniqu	es used		
		 Measure v report) 	worst case	gas reside	nce time	using a time	e of flight	t method (as	specified	l in the BAT		
		Use multip temperatu	ole traverse ire location	e measurer at, or shor	ments of g	gas tempera the qualifyin	ature to i	dentify (or co dary combus	onfirm) the stion zone	e lowest gas e.		
		Confirm the identified is minimum for the second seco	hat 95% of t lowest temp temperatur	the one-mi perature lo e requirem	nute mea cation ov ient.	an temperati er a period	ures (cor of at leas	ntinuously m st one hour)	onitored a exceed th	at the ne stated		
		Use suction thermocol 2.10 - More thermology in the success of the success o	on pyromet uples may on nitoring of p	ers to mea only be use process va	sure tem ed if calib riables).	peratures (a rated agains	acoustic st suction	oyrometers on pyrometers	or shielde s, <mark>(see als</mark>	d so Section		

				IES			NS			Т		
	aterials	Activities/	Ground		_			• • • •		Installation		
Management ir	nputs	abatement	water	Waste	Energy	Accidents	Noise	Monitoring	Closure	issues		
Furness	_											
requirements	3.	At some exist in 6 above. Th practicality of o justifying a site • Compliance to air may be methods. • The siting of	6 above. The following matters are considered to be valid when assessing the costs and acticality of carrying out temperature / residence validation. They should be considered when stifying a site specific approach at existing installations: Compliance with combustion related emission limit values (e.g. VOC, NOx, CO) for releases to air may be taken as indicative of adequate combustion and allow for reduced cost validation methods. The siting of new monitoring access holes may be problematic at existing installations.									
	4.	The " qualifyin is defined as fo	" <i>qualifying zone</i> " over which the temperature and residence time shall be required to comply efined as follows:									
		Should not combustion	Should not include areas where primary combustion occurs but relate to the <i>completion</i> of combustion									
		Should be	referred to	as the qua	alifying se	econdary co	mbustio	n zone or Q	SCZ.			
		Should cor therefore g	nmence at enerally ex	a location	after the dence tin	last injection ne achieved	on of seco I in the p	ondary (or o rimary comb	ver fire) a ustion un	ir and will it or zone.		
		Does not re above the i	equire rese equired lev	t where su vel. vhere tertis	ipport bu	rners are io	cated pro	ovided they i	naintain t	emperature		
	L.				ary air is c							
	5.	 Carried out 	itions for v over a ran	alidation r	neasurer rational c	nents shoul onditions in	d be: cluding t	he "most un	favourabl	e" and		
		 The "most of: 	unfavourat	ole" conditi	on is con	sidered to a	arise as a	a consequen	ice of a c	ombination		
		- waste t related	ype being a	at the bour s (e.q. CV	ndary of t	he design e e)	envelope	in respect o	f it combu	ustion		
		- the pro- diagran	cess opera	ting at the	limits of	its operation	nal range	e as defined	by the pla	ant firing		
		Each condThe monitor	tion should pring within	l be testec each test	l twice du period sh	iring the val nould last at	idation p least on	rogramme ie hour				
	For i com	more detailed g bustion conditio	uidance on ns.	validation	methodo	ologies refe	r to the E	3AT report o	n validatio	on of		
	2.3.	4.5 Measur	ing oxyge	en levels								
	App Que	olication Form estion 2.3 (cont.)	BAT for	measurir	ng oxygen le	evels is:					
	Wit	h the Applic	ation the	e Opera	tor sho	uld:						
	1.	supply the gen activities.	eral Applic	ation requ	irements	for Section	2.3 on p	age 39 for th	nis aspect	t of the		
	Ind	icative BAT	Require	ments								
BAT for measuring oxygen levels	1.	The EC Direct Zirconia based levels. The dif expressed as a	ves do not l technique ference be a wet meas	state whe s measure tween the surement o	ther com the wet dry and t or 10% dr	bustion oxy level where the wet figu 'y.	gen leve as extra res may	els are mease ctive system be 20%, e.g	ured wet o s measur . 8% oxyg	or dry. e the dry gen		
	2.	 expressed as a wet measurement or 10% dry. 2. This emphasises the need to set the oxygen control point at a level for a particular plant and waste which takes account of the speed with which the control system can introduce more secondary air in response to fluctuations in the rate of combustion on the grate. The larger the fluctuations and slower the rate of response of the control system, (see Section 2.3.4.6 - Combustion control) the larger the margin of excess oxygen must be. Some operations are running with 12% excess oxygen whereas a <i>ChWI</i> burning a consistent liquid feed may only need 2% to achieve effective combustion. During short-term perturbations the level will therefore fall below these values. 										

INTRODU	CTIC	ON	TEC		UES	E	MISSIO	NS		IMPA	CT	
Management Ma	aterials	s Ac	tivities/ atement	Ground water	Waste	Energy	Accidents	Noise	Monitorin	g Closure	Installation issues	
Furnace requirements	2.3	.4.6	Combus	stion con	trol	I						
	Ap Qu	Application Form Question 2.3 (cont.) BAT for combustion control is as follows:										
	Wi	ith the Application the Operator should:										
	1.	supply the general Application requirements for Section 2.3 on page 39 for this aspect of the activities.										
	Ind	dicative BAT Requirements										
BAT for combustion control	1.	Careful consideration must be given to maintaining optimum control of the combustion process at any instant, especially when burning wastes of very variable moisture content and calorific value and those which cannot be readily charged to the furnace at a steady rate.										
	2.	Con tum	trol will be pling of a n	better whe nass of wa	ere the larg ste in an N	gest pertu ∕IWI) is s	urbation (be mall compai	it a sing red with	le drum to a total mass	a ChWI, or being burr	the ned.	
	3.	Beca espe burn	ause waste ecially durin er operatio	e feed rate ng stoking, on.	is a relativ need to b	vely slow e control	acting conti led by prima	rol parar ary and s	neter, shor secondary a	er term flu air flow rat	ictuations, es and	
	4.	The hom (suc and	shortest-te ogeneous h as CO o unburned	erm fluctua wastes an r oxygen so hydrocarbo	tions are u d take pla ensors) m ons) before	usually ca ce in the ust be us e the con	aused by su order of sec ed to avoid trol system	dden co conds. F short ter reacts.	nflagrations Fast respon rm releases	s of the nor se measur s (particula	n- ring systems rly of CO	
	5.	Оре	rators sho	uld demon	strate how	their cor	ntrol system	will dea	I with both:			
		• t	he largest he shortes	normal per t duration i	rturbation; perturbatic	and on which	is significan	t in the r	particular p	ocess		
	6	Pote	ntially the	response f			r svetome m	av be bi	ought dow	to the mi	crosecond	
	0.	leve usin plan effic valu	I. Alternati g acoustic ts, this alo iency and able for im	wely rapid (which car ne has sho can save fu proving pe	response n be exper own signific uel where prformance	can be o nsive) or cant redu burners a on exist	btained by to optical/infra- ictions in CC are regularly ing plants.	aking m -red tem D release y employ	easuremen perature m es. Better o ved. Such t	ts just abo onitoring. control also echniques	ve the bed, On some o improves can be	
	7.	To b whic	e effective h can also	, rapid mo respond r	nitoring ne apidly. Te	eds to be chniques	e combined s include:	with a s	econdary a	ir supply a	rrangement	
		• k c v	eeping se an be mor vindows;	condary jet nitored by s	ts clear of simple air	slag and flow or pi	operational ressure instr	l - partici rumenta	ularly in MV tion, backe	VIs. Jet pe d up with v	erformance viewing	
		• (5	xygen inje significant i	ection, via l reductions	ances, has in the nun	s been us nber of hi	sed for mero igh CO ever	chant Ch nts	WI in the L	IS and Ge	rmany with	
		• a t t i c	although th he air supp peing by da ncrease in obtained by	ere are no oly ductwor amper oper air flow thr controlling	known ex rk upstrear ning so tha rough the j g the fan s	amples i m of the j at, on ope jets. This speed.	t would appe jets and higl ening of the s may provid	ear to be her pres damper de a mue	e possible to sure fans, v s, there will ch faster re	o provide o vith contro be an imr sponse tha	capacity in I to the jets nediate an that	
	8.	Star NO _x useo	ved air sys is normall I with sepa	tems redu y formed. arate cham	ce both the They can bers, (see	e oxygen combine Section	good NO _X a 2.3.8).	d the ter and goo	nperature i d CO perfo	n the area mance pa	where the rticularly	
	9.	Whe	ere support	burners a	re used (s	uppleme	ntary firing)	the use	of low NO _X	burners is	BAT.	
	10.	Meth in wh the s form	nane (natu hich the ga secondary ed back to	ral gas) ad as is either combustio o N ₂ .	dition is al injected ir n area (tel	n emergi ito the be rmed reb	ng technique ed where it c urn) where i	e, althou can supp it can ree	igh not yet press the fo duce the N	commercia rmation of D _X which h	ally proven, NO _X or into nas already	
	-											

INTRODUC	CTION TECHNIQUES EMISSIONS IMPACT									
Management Ma	terials Activities/ Ground Waste Energy Accidents Noise Monitoring Closure Installation issues									
Furnace requirements	2.3.4.7 Combined incineration of different waste types This section relates to the combustion of different types of wastes within the same incinerator. It does									
	not deal with the combustion of wastes with, or in the place of, other fuels at installations whose primary purpose is the generation of energy or the production of materials. The techniques that represent BAT in these circumstances should be determined by reference to the appropriate sector guidance.									
	Application Form Question 2.3 (cont.) BAT for combined incineration of differing waste types is as follows:									
	With the Application the Operator should:									
	1. supply the general Application requirements for Section 2.3 on page 39 for this aspect of the activities.									
BAT for	Indicative BAT Requirements									
co-incineration of differing waste types	Because BAT for the incineration a particular waste will be dependent upon the characteristics of that waste a dedicated furnace, abatement system and monitoring train is likely to be BAT where anything other than very low proportions of alternative waste types are proposed.									
	Where combined abatement is used at existing plant, the design should address the potential for loss of efflux velocity at the stack top when either incineration line is not operating and monitor at locations that may allow calculation of emission limit values for each line.									
	It will not be acceptable to use combined flue dilution to meet emission limits.									
	Clinical waste in MWIs									
	 Waste which would normally render clinical waste hazardous, i.e. sharps, pharmaceuticals, infectious waste and body tissue (beyond incidental amounts) should be incinerated in dedicated CWI plant. The incineration of some types of clinical waste may require the plant to meet the requirements of HWID. 									
	 WID requires that infectious clinical waste is placed straight in the furnace without first being mixed with other waste, and without direct handling. 									
	 Generally, the more innocuous waste, i.e. Class E waste as defined by the Health and Safety Commission in Safe Disposal of Clinical Wastes (1992), which is suitable for landfill, may be burned in a well run MWI, but only if: 									
	the incinerator meets the modern standards expected for municipal waste;									
	 a strict code of quality control is exercised on the source of the waste and its handling into the incinerator and the procedures for these are regularly audited; 									
	• the CW is burned within 24 hrs and records are kept of temperature and quantity of waste fed;									
	 that procedures are in place to divert and transfer waste already held, should the incinerator be out of action; 									
	 a mass throughput limit is applied which corresponds to a small fraction of the total waste burned - say < 10%. 									
	Clinical waste in ACIs									
	 Animal carcass incinerators are unlikely to be able to meet the standards required (WID) to enable them to incinerate other types of wastes. 									
	Sewage sludge in MWIs									
	5. There are no current applications for this arrangement. Any proposals for this or any other co- incineration project will be dealt with as they arise.									


Chimneys & vents

2.3.5 Chimneys and vents

2.3.5.1 General

Requirements in respect of the assessment of environmental impact of releases to air are outlined in Section 4.1.

With other factors the same (e.g. plume temperature, efflux velocity) higher release points result in lower ground level pollutant concentrations. Operators should justify the selected release height selected and include their assessment in the application.

The assessment should include graphs showing:

- The change in unit ground level pollution concentration against stack height
- stack height against additional cost
- the change in unit ground level pollution against cost

The assessment should also take account of other factors such as the need to further reduce a particular pollutant concentration below a specific significance threshold or planning restrictions. The costs associated with increasing stack height should generally be restricted to construction and maintenance costs. Costs associated with gaining or amending planning permission should be detailed separately but may be taken into account (but should not be over weighted) when justifying the selected stack height.

2.3.5.2 Wet plumes

Wet plumes may not disperse well and can ground easily. Additionally there can be local visual amenity issues and, in severe cases, loss of light issues. Therefore, unless it is agreed that these issues are not locally significant, the gas should be discharged at conditions of temperature and moisture content that avoid saturation under a wide range of weather conditions. The requirement could be specified as a maximum permissible length of plume or as no visible plume for a given percentage of the year.

The normal option to reduce plume visibility is to add heat, but **the use of energy should be balanced against the benefits gained**. Plumes must be abated sufficiently to ensure good pollutant dispersion but a limited visible plume may be acceptable in some conditions. Eliminating plume visibility under all meteorological conditions may not be possible and could result in excessive energy use, to a point that it would not be considered to represent BAT.

Alternatively, moisture can be removed by cooling and condensation, followed by reheat. However, where this is not an inherent part of a wet scrubbing process (where temperature is typically reduced to around 70 °C for scrubbing purposes), the disadvantage would be the generation of a significant liquid effluent stream.

As a further option, to minimise expense and energy use, Operators may wish to guarantee to reduce load under extreme weather conditions rather than to over-design a plume abatement system.

2.3.5.3 Dump stacks and bypasses

Dump stacks should only be included where they are essential for safety reasons. In general in should be possible for dump stacks to be ducted to the main stack, thus forming a bypass and improving dispersion with the additional height and allowing monitoring equipment to quantify the release.

Systems must be designed so that the dump stack is not normally expected to operate. Operational frequencies greater than once per year are unlikely to acceptable. When a dump stacks or emergency bypass operates this will be considered to be a period of "abnormal operation" and the process should be reduced or closed down (Ref. WID Article 13).

Start-up and shutdown should normally be achieved without any releases from the dump stack. An abatement system bypass, linking to the main stack may be operated on start-up where this has been authorised and is necessary to prevent damage to abatement systems.

Electric heating is an available option for new bag filters to avoid the need for bypass on start-up. Failure of the flue gas cleaning plant should not normally lead to operation of the dump stack. The reliability of heat removal systems, in particular feed pumps and dump condensers, should be demonstrated to be adequate.

Further guidance in respect of *abnormal operating conditions* may be found in Section 2.8 - Accidents and their consequences.

INTRODUCTION		N TEC	CHNIQUES		EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Cooling towers	2.3.6 Cooling systems (local pollution prevention aspects)
	This section deals with the local pollution separate of sociling systems. Energy officiency appeats are

This section deals with the local pollution aspects of cooling systems. Energy efficiency aspects are discussion in greater detail in Section 2.7, and in the cross sector BAT reference note on industrial cooling systems.

2.3.6.1 Cooling systems types

Cooling systems are mainly required at incineration installations for:

- Condensing boiler water for re-circulation after a steam turbine (the major use);
- Cooling scrubber waters to reduce scrubber water evaporative losses;
- Cooling quench water;;
- Cooling of mechanical operations (e.g. pumps etc);
- Condenser chilling.

The main cooling systems in use at waste incineration installations are where electricity is generated using steam turbines. The need to retain (expensive) boiler water means that they will be closed circuit (i.e. the boiler water is retained within the system for re-circulation).

The main differences arise in the design of the heat exchanger and the source and fate of the cooling medium. In this sector the cooling medium is usually supplied by:

- Once through sea water or river water;
- Evaporative cooling tower;
- Forced draft air cooling.

Cooling System Type	Advantages	Disadvantages
Once through	 Greater cooling efficiency may improve energy recovery Low noise impact Low visual impact 	 Possible fish kill Possible thermal release effect in water course Bio-fouling Biocide discharges
Evaporative cooling	Good cooling efficiencySmall plot possible	 high visual impact water consumption chemical treatments for bio- hazard control
Air cooling	No water intake or dischargeUnobtrusive designNo water consumption	 Possible noise impacts Lower cooling efficiency Power supply costs

Application Form Question 2.3 (cont.)

BAT for cooling systems is as follows:

With the Application the Operator should:

- 1. supply the general Application requirements for Section 2.3 on page 39 for this aspect of the activities.
- 2. Justify their chosen technique with regard to the criteria and guidance outlined in this section.

Indicative BAT Requirements

2.3.6.2 Discharge of cooling tower water

BAT for cooling systems

Table 2-5-Cooling system type advantages and

disadvantages

1. Where evaporative cooling towers are used, biocides lead to prescribed releases to both air and water and their use should be minimised (commensurate with meeting health and safety requirements) by optimising the dosing regime (e.g. intermittent shock dosing or only dosing at critical times of the year). The use of automatic mechanical cleaning systems for main condensers minimises the use of biocides.

INTRODU	СТІС	N TE	CHNIQ	UES	E	VISSIO	NS		MPAC	Т
Management H	aterials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
	_									
Cooling towers	2.	The engineeri released to the coupled with a vessels and a	ng of the b e environm automatic c dequate op	iocide systement. This operation of operation of operation of operating products of the systemetry of	em shoul would inv f the final ocedures	d prevent a olve monito discharge	occidenta pring of le valves, a	l overdoses evels in the o s well as bur	of biocide utgoing v nding of s	e being vater storage
Cooling	23	6.3 Cooline	n water in	takes						
systems	2.0.	0.0 000000		lanco						
(cont.)	3.	Once through energy efficien • There is ac	cooling syn ncy of the i dequate pr	stems can nstallation. ovision of v	be more Such sy vater (e.g	efficient an /stems will (g. coastal si	d may the only be s tes);	erefore impro uitable where	ove the o e:	verall
		CHP or dis Eish (and (othor agua	ig cannot p	by the w	y use the w	aste nea		nd will no	at ho
		significant	;		Jy line w			1 assesseu a		
		Thermal a	nd biocide	dispersion	are such	that enviro	nmental	impacts are	not signif	ficant;
		The energ alternative	y and any technolog	other envir ical solution	onmental ns (e.g. a	benefits ca ir condense	an be der ers).	nonstrated to	o outweig	lh
	2.3.	• 6.4 Cooling	g tower p	lumes						
	4.	Large conden ground level c roads. Such e	sed plumes an contain effects sho	s, such as harmful su uld be avoi	those from Ibstances ded.	m evaporat s and cause	ive coolir e loss of l	ng towers, wl ight, poor vis	nich com sibility and	e down to d icing of
	5.	If required, plu required will d confirm that th areas of habits	ume abater epend upo ne visible (o ation at a h	ment should in local fact condensed neight that i	d prefera ors. Plui) plume v will cause	bly use reje me modellir vill not grou e significant	ect heat. ng should nd beyon loss of li	The degree I be employe Ind the bound ght.	of abater ed by an (ary fence	nent Operator to e nor reach
	6.	Eliminating plu result in excess alternative, an when conditio (e.g. use air co	ume visibili ssive energ o Operator ns worse tl ooling) to e	ty under al gy use, to a may desigr han those a ensure that	l meteoro point than for parti are exper the requi	ological con at it would n icular condi ienced, red irements ar	ditions m ot be cor tions of te luce load e met.	ay not be ponsidered to re emperature a or take othe	essible an epresent and humi r appropi	ld could BAT. As an dity and, riate action
	2.3.	6.5 Release	es to land	1						
	7.	Timber used in dichromate, an life, but initial treated timber	n cooling to rsenic pent surface res should inc	owers is us toxide), mo sidues can clude the re	ually trea st of whic lead to si quiremer	ated with CC ch remains gnificant le nt for contro	CA (copp well bour vels in the olled was	er sulphate, nd to the timl e purge wate hing at the tr	potassiur ber over i er. Speci eatment	n its operating fications for site.
	8.	On final dispo out if it has be	sal, inciner en specific	ation of the ally author	e cooling ised.	tower timbe	er in the i	nstallation m	ay only b	be carried

INTRODU	CTION TECHNIQUES EMISSIONS IMPACT											
Management M	terials Activities/ Ground Waste Energy Accidents Noise Monitoring Closure Installatio											
Boiler	227 Boiler design											
design	This section deals with boiler design as it relates to the minimisation of local pollution.											
	Energy efficiency matters are dealt with in Section 2.7. However, it should be noted at this stage that WID requires <i>"any heat generated by the incineration or the co-incineration process shall be recovere as far as practicable"</i> . The design of boilers clearly has a crucial role in ensuring hat this requirement is fulfilled at many installations, but this must not be at the expense of unacceptable additional local pollution.											
	Application Form Question 2.3 (cont.) BAT for pollution prevention is as follows:											
	With the Application the Operator should:											
	1 supply the general Application requirements for Section 2.3 on page 39 for this aspect of the activities.											
	2 Explain how their boiler design takes account of the guidance in this section in respect of designing the boiler to prevent emissions.											
	Indicative BAT Requirements											
BAT for flue gas	Minimising dioxin production by boiler design and operation:											
recirculation	1. Slow rates of combustion gas cooling should be avoided may increase the scope for <i>the de-nov</i> formation of dioxins and furans.											
	The primary temperature zone of concern is between 450 and 200 °C, however dioxins will still the formed outside this range at a decreasing rate as the temperature moves further away from this core range, (see Section 2.3.8 - Dioxins).											
	It should be stressed that the philosophy for dioxin control has its emphasis on preventing formation (rather than subsequent abatement) and, as one of the primary sites for formation, the design and operation of the waste heat boiler is important. The main techniques are:											
	2. Maximum rate of decrease of gas temperature which is achieved by:											
	 ensuring that the steam/metal heat transfer surface temperature is a minimum (around 170 °C) where the flue gas is in the <i>de novo</i> synthesis temperature range, subject to acid dew point considerations; 											
	 CFD is used to confirm that there are no pockets of stagnant or low velocity gas; 											
	 boiler passes are progressively decreased in volume so that the gas velocity increases through the boiler; and 											
	 boundary layers of slow moving gas are prevented along the boiler surfaces. 											
	A balance must be maintained, to ensure that these design measures are not made at the expense of a major effect on boiler efficiency.											
	3. <i>Minimising boiler deposits</i> (which contain substances which catalytically enhance dioxin formation) is a problem with most wastes. Municipal waste, in particular, leads to deposits of sodium and potassium sulphates, and to a lesser extent chlorides. Fly ash can then adhere to these deposits to compound the problem. In the initial stages the material is easily removed by sootblower, on-line. As the fouling increases the deposits become fused and can only be removed off-line. Control methods include:											
	 design features to maintain critical surface temperatures below the sticking temperature. Thi includes not only the arrangement of cooling surfaces, but also avoiding peak combustion temperatures by good waste mixing (where relevant) (see Section 2.3.1 - Incoming waste handling), uniform waste feed (see Section 2.3.2 - Waste charging) and good primary and secondary air control (see Section 2.3.4 - Furnace requirements); 											
	 additives to prevent sodium and potassium depositing (mixed success); and on-line cleaning by: 											
	 boiler tube rapping, by striking the tubes (limited success) or lifting and dropping whole banks of tubes (limited experience); 											

IPPC

INTROD	UC	TIO	N TE	CHNIQ	UES	El	MISSIO	NS		MPAC	Т
Management	Mat in	erials puts	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
	_										
design			- continu econor	iously allow	ving steel s ons);	shot to fal	ll through th	e tubes	(applied succ	cessfully	to
			- steam	or compres	sed air so	ot blowing	g; and				
			- off-line	cleaning.							
		4.	NOx reduction Dioxins, Section	n technique on 2.3.4.8 -	s may also FGR and	help to i Section 2	minimise di 2.3.8 - SNC	oxin emis	ssions- <u>(see</u> CR).	Section 2	2.3.8 -
		Mini	mising releas	es to wate	r from boi	lers:					
			Boiler blow-do phosphates w and de-aeratio	own contain ith possibly on.	s small an small am	nounts of ounts of a	solids plus alkalis, hydr	water tre azine an	eatment cher d ammonia ι	nicals - n used for p	nainly oH control
			Water treatme streams which content will de sulphates are exchange mat	ent and de-i n are mixed epend on th also likely t terial. The	onisation together a e original to be prese presence	olant efflu and pH ad water sup ent from t of salts in	uent usually djusted for o oply, be it to the use of s the release	compris discharge wns wate ulphuric e should	es separate e. Soluble al er, river or es acid for rege be considere	acid and nd suspe stuary wa neration ed.	alkali nded solids iter. Soluble of the ion
			Wash water a phosphates, in during mainte	nd cleaning ron oxides i nance. Coi	g solutions in suspens mplex toxi	, containi ion, hydr c corrosic	ng for exam ochloric or l on inhibitors	nple citric hydrofluc may be	acid, sodiur pric acids, ma present in th	n hydroxi ay be gen lese liquo	ide, alkali nerated ors.
		5.	All these liquo before dischar	rs should b rge or dispo	e neutralis	ed or tre censed w	ated on- or aste dispos	off-site to al facility	o produce ar	accepta	ble waste

INTRODUCTION		N TEC	CHNIQ	UES	E	MISSIO	NS	IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Abatement	2.3.8	Abatement of point source emissions to air	
to air			

2.3.8.1 Legislative requirements

Following the use of techniques to reduce the production of pollutants (including consideration of waste pre-treatment options) it will be necessary for abatement techniques to be employed that will meet the emission limit value requirements of European Directives, *as a minimum.* The Operator will be required to demonstrate in their application that BAT has been employed. It will not be sufficient to simply design a plant so as to meet the emission limit values, without considering how further emission reductions may be obtained through the use of BAT.

Installations in this sector that are exempted from the requirements of WID will be expected to meet the standards of the directive in all cases other than where the Operator can clearly demonstrate that the costs of meeting the particular requirements outweigh the benefits. New installations are less likely to be able to justify departures from the directive standards than existing installations.

2.3.8.2 Nature of the emissions to air

Sources

The nature of the emissions to air from the combustion of wastes is relatively well characterised. The potential releases will depend on the nature of the waste. In general emissions will comprise:

- · Particulate matter
- acid gases e.g. HCl, HF, SO_X , NO_X
- heavy metals
- volatile organic compounds, carbon monoxide, dioxins and furans
- carbon dioxide and water

2.3.8.3 Control of point source emissions to air

Guidance Abatement technology should be selected such that the emission limit requirements of European legislation are complied with **as a minimum**. Operators will not only be required to demonstrate that their techniques will meet these standards but will also need to demonstrate why further emission reductions cannot be made through the use of BAT.

There may be benefits accrued from using a number of the techniques described in this section in combination. Furthermore, the selection of one particular abatement system or a particular combustion design (e.g. fluidised bed) may, for valid engineering and environmental reasons, exclude the use of, or undermine the performance of, an alternative abatement system. It will therefore be appropriate for operators to justify their individual equipment selections by reference to the performance of the installation as a whole i.e. operators should set out a number of alternative installation designs and compare the overall performance. Guidance on the assessment of alternative installation designs has been produced by the Agency (H1) and should be used to justify the chosen option.

The nature and source of the emissions expected from each activity is given in previous sections and will be confirmed in detail in the Operator's response to Section 3.1.

Cross-sectoral guidance on abatement techniques for point source emissions to air can be found in Ref. 10.

Application Form Question 2.3 (cont.)

Control of Point Source Emissions to Air

With the Application the Operator should:

- 1. supply the general Application requirements for Section 2.3 on page 39 for control and abatement equipment; and in addition
- 2. describe the measures and procedures in place and proposed to prevent or reduce point source emissions to air. This should include, but is not limited to, the general measures described below and justify where any of the measures are not employed.

INTROD	UCT	ON	TEC	HNIQ	UES	EMISSIONS				IMPACT			
Management	Materia input	als <mark>Activi</mark> s <mark>abate</mark> r	ties/ 0 ment	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Abotant													
to air	3.	provide require	the follo d should	wing witl be estat	h the applic blished in p	ation as re-applic	appropriate ation discus	. If there sions:	e is doubt, th	e degree	of detail		
		• a de	escription	n of the a	batement e	equipmer	nt for the act	tivity;			• •		
		 the of V 	identifica 'OCs) an	d assess	ne main che sment of th	emical co	these chem	icals in the	issions (part ne environm	ent;	r mixtures		
		• mea	asures to		e the secur	ity with w	hich the rec	quired pe	rformance is	delivered	1;		
		 measures to ensure that there is adequate dispersion of the emission(s) to exceedences of local ground level pollution thresholds and limit national an pollution impacts, based on the most sensitive receptor, be it human health ecosystems; 											
		 dam 	hage to h	ealth or	soil or terre	estrial eco	osystems.						
	4.	demons Guidan support	strate that ce is give ted by mo	at an app en in Tec ore detai	ropriate as chnical Gui led dispers	sessmen dance No ion mode	t of vent an ote D1 (<mark>Ref</mark> . elling as des	d chimne 12); for f scribed in	ey heights ha this sector th 1 Section 4.1	as been m nis will nee	ade. ed to be		
	5.	where a likely be levels o probabi any sig avoideo houses should preferre The imp	appropria wer shor wer shor nificant ri d. It may in order be ducte ed but co pact of fu	ate, also Proces t periods currence isk to hea be poss to preve d to the sts shou ugitive en	recognise t s upsets of s should be e, the heigh alth. Wher ible to desi nt dew poin main stack Id be consi nissions ca	he chimr equipme assesse t of the c ever pose gn out th ht probler to ensure dered in n also be	ney or vent a ent failure g d. Even if th himney or v sible, the us eir routine u ms. At new e maximum relation to th e assessed i	as an em iving rise he Applic ent shou se of abai use for st plant an dispersion he likely in many o	ergency em to abnorma cant can den ld neverthele tement bypar art-up by pro y essential n on. At existi impacts and cases	ission poil lly high er nonstrate ess be se sses shou oviding he najor bypa ng plant ti frequenc	nt and the nission a very low t to avoid uld be ated bag asses nis is also y of use.		
	In	Indicative BAT Requirements											
	1.	 The Operator should complete any detailed studies required into abatement or control options (see item 3 in Section 2.3) as an improvement condition to a timescale to be agreed with the Regulator but in any case within the timescale given in Section 1.1; 											
	P	Particulate matter											
BAT for particulate	2.	Filters i use as	mpregna an additi	ated with onal con	catalysts tl trol should	nat destro be consi	oy dioxins a dered.	nd furans	s have been	develope	d and their		
matter	3.	Fabric of partic cannot (incorre reagent	filters an culate ma be used ect tempe ts).	re prover atter to b at high to eratures i	n and when elow 5mg/r emperature may also le	correctly m ³ and and es (over a ead to a re	y operated a re likely to b approx. 250 eduction in a	and main be BAT fo °C) as th acid gas	tained provio or many appl nis may give absorption b	de reliable lications. rise to fire by some a	e abatement They e risk Ikaline		
	4.	The fab individu maintai limits de	oric filter s lal bag fa ned whe uring on	should ha ailures. 1 n filter ba stream n	ave multipl There shou ags fail, i.e. naintenanc	e compai ld be suff design s e.	rtments, whi ficient of the should incor	ich can b se to allo porate ca	e individuall ow adequate apacity for m	y isolated performa leeting en	in case of ince to be hission		
	5.	There s when th better c	should be nis happe control of	e bag bur ens. The emissior	rst detector ese may be ns than sim	s on eac of the di ple obse	h compartm fferential pre rvation of e	ent to incessure ty mitted pa	dicate the ne pe. This typ articulate leve	eed for ma be of syste els.	iintenance m provides		
	6.	Where may rec practica	wet scrul quire reh able) to p	bbing is eat (usin prevent d	used in cor g indirect h ew point pr	nbination leat exch oblems.	with fabric ange from a	filters (e. an otherw	.g. HWI), the vise waste h	cool and eat source	wet gases e where		
	7.	Ceram i use has plant ca "blindin	ic filters s general apacity re g" and ar	provide Ily been I equired) a re therefo	an alternat limited to si at higher te ore genera	ive where maller pla mperatu lly consid	e high tempo ant owing th res. Fabric lered BAT.	erature fi e larger (filters ter	Itration is rea gas volumes nd to be less	quired, alt (and hen susceptil	hough their ce filtration ble to		
											Cont.		

INTRODUC	CTIC	ON TEC	HNIQ	JES	E	MISSIO	NS		MPAC ⁻	Г
Management Ma	iterials	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure ^{II}	nstallation issues
Abatement to air BAT for particulate matter (cont.)	8.	Electrostatic p same extent as have the advan derived from in loading on bag be minimal whe to the overall p themselves and	brecipitato fabric filte tage of lov duced drat filters and ere reagen ressure dra	ors (EPs), ers and are w pressure ft fans. Th hence red ts are dos op is itself	either we e not con e gradien ney may f duced en ed onto l relatively aver form	et or dry, are sidered ade ts which ma therefore pro ergy consur barrier filters y minor in co red	e not cap quate or y result i ovide a r nption. I a as the c ompariso	able of abat their own. in lower para neans of rec However, th contribution in to that cre	ting particul EPs do hov asitic energ ducing partic is energy sa of the partic ated by the	ate to the wever y loads culate aving will culate load filters
		themselves and	a the reage			icu.				
	9.	Wet scrubbers not capable of advantages in represent the E give rise to liqu This has implic	s are not comeeting the respect of BAT in comination of the second id effluent, ations whe	onsidered e same ei the contro bination v which, if en conside	to be BA mission le I of other vith barrie not recyce ring the l	AT for particulation of the pa	ulate aba er techni s (e.g. sc echnique process,	atement on t ques. They luble acid g es as mentio , requires tre	heir own, as can howev ases) and n ned above. eatment and	s they are er offer nay They I disposal.
		Plume visibility employed – the of the installatio imported energ	is likely to source of on with oth y sources	be increa this heat er waste h unlikely to	sed when will have neat bein o represe	re wet scrub implications g an accept nt the BAT.	bers are s in relati able sou	employed u on to the ov rce, but with	unless plum erall energy the use of	e reheat is / efficiency additional
	Ох	ides of nitrog	gen							
BAT for NOx		The following to discharges to the following term of	echniques he atmosp	may repre here.	esent the	BAT for the	reductio	on of oxides	of nitrogen	
		Primary NO	x measu	ires						
	10.	<i>Fuel Selection</i> which are nitrog	There are gen rich (e ion outline	e not likely .g. sewag d below.	y to be op e sludge	oportunities) will need to	to reduc pay pai	e fuel NOx i rticular atten	n this secto ition to the t	r. Wastes echniques
	11.	Combustion C combustion ten other designs a wastes of a cor feeds (e.g. raw proved problem and fires occur therefore be we	Chamber L nperatures and are con hisistent an municipal hatic for so ring. The p eighed aga	Design - F than other mmonly us d small pa waste) if f ome waste potential N inst these	luidised l er system sed for se article siz they are streams NOx redu potentia	bed combus is. They car ewage sludg e but are not not pre-treat (e.g. mixed ctions of cor Il difficulties	tors (FB therefo is incine t suited t suited rad. Was raw mu mbining for heter	C) operate a re produce 1 ration. They to large or h ste feed pre hicipal waste FBC and fee ogeneous w	at relatively less therma / are well su eterogeneo paration sta e) with brea ed preparati vaste types.	lower I NOx than uited to us waste uges have kdowns on must
		Where the emis need for secon (or no) reagent being given. T levels, and pro- furnace temper The primary co	ssion limit dary abate injection r his, and th vide optima ratures me nsideration	values sta ement (e.g nay repres e ability of al reagent ans that th n should th	ted in Eu . reagent sent the I f other no reaction nere is cu nerefore	uropean Dire t injection), a BAT. Howey on-FBC tech conditions (urrently little remain that	ectives cand the v ver, such niques to see SNC to choos of waste	an be guara vaste is suita o guarantees o meet the r CR below) a se between f characteris	nteed witho able, FBC w s are not ge equired em t slightly hig these techn tics.	ut the vith limited nerally ission yher ologies.
	12.	Air Control – µ NOx productior maintained unc gas releases).	o rimary ar n. All equij ler slight n	n d second pment sho egative pr	dary - Hig buld there essure to	gh excess ai efore be sea o allow contr	r at the o led to pro ol of air	combustion event fugitiv input (and to	stage can ir e air ingres o prevent co	ncrease s and ombustion
		Primary and se chamber secur not being exces	condary a e oxidative ssive whicl	ir feed sho e combust h would re	ould be o ion of ga sult in hi	ptimised so ses (and hei gher NOx pr	that con- nce dest oductior	ditions in the ruction of or 1.	e combustio ganic speci	n e), while
		At new plant, o	r those une	dertaking	upgrade	of the comb	ustion ch	namber,		
		 CFD should alternative (allow for in 	l be used t (multiple) a service on	o select o air injection timisation	ptimal pr n ports ai	imary and so nd directiona	econdary al injectio	y air input re on nozzles s	gimes; hould be pr	ovided to
			op							Cont.

INTROD	JCTIC	DN TEO	CHNIQ	JES	E	VISSIO	NS	I	MPAC	;T
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Abatement to air BAT for NOx (cont.)		Existing munic MWID/HWID r December 200 meet the stand minimum exce the level of ex requirements appropriate to oxygen conce	cipal and harequirement 5. After the dards of the ess oxygen cess oxyge of achieving include in tertation to version	izardous w ts that oxy is date the new Dire concentra n they inte oxidative he applica /ary either	vaste inci gen shou se plants ctive (200 tion. This end to op combus tion data side of th	nerators and ild be prese s (and other D/76/EC). T s being the erate at, and tion whilst m to show ho ne selected	e require ent in at lo existing his new case ope d justify l ninimisin ow emiss optimal	d (respective east 6% exce incinerators Directive doo erators will be how this will g NOx produ ions would c point.	IV) to con ss until 2) will be re s not spe e required meet the lotion. It w hange we	nply with 8 equired to cify a 1 to state would be ere the
		Pyrolysis and operate the ini exclusion of or promote gas e are well contro stated in this s are relevant to stages.	a gasificati itial waste ci xygen and sevolution. It blled to prev section rega the subsection	on plants lestruction semi-pyrol is importa vent gas e arding bala quent com	are a spe stage at ytic and g ant that th scape an ncing the <i>bustion</i> c	ecial case in reduced ox gasification nese "reaction d to create e need for o of the produ	that the tygen lev plant use on" stage optimal of xidative cts that r	y are specific rels. Pyrolys e sub-stoiching so are sealed conditions. T combustion a result from th	cally designs is itself recommetric le commetric le l, and that the considerand NOx presented and NOx presented to the configuration of	ined to quires the vels to t air flows derations prevention reaction"
		Technical guid engines or gas that their subs	lance for th s turbines is equent con	e combust provided nbustion w	tion of pro in other ill be req	oducts of th Agency guid uired to cor	ese proc dance. H nply with	esses in inte owever, it is WID standa	rnal comb important rds	oustion to note
	13.	Temperature greater than 8 halogenated o be that which must be met a	Control To 50 °C or 11 organics (as relates to a it all times v	emperatur 00 °C whe chlorine). chieving th vhen wast	es must i ere hazar The loc ne require e is being	meet the re- dous waste ation at whi ed residence g burned inc	quiremer contains ch this te e time of cluding si	nts of the releasing s greater that emperature s 2 seconds. tart up and s	evant Dire n 1% w/w hould be These rea hut down.	ctives i.e. met should quirements
		Excessive or u Water cooled	uneven tem grates may	peratures assist with	should h	owever be a ature contro	avoided a	as this may le	∋ad to hig	her NOx.
		As with 12 abo subsequent co stage itself.	ove, for pyro ombustion c	olysis and of the prod	gasificati ucts of th	ion plants the reaction s	nese con stage, rai	sideration re ther than to t	late to the he initial r	eaction
	14.	Flue Gas Rec of secondary a consumption of energy recover inclusion in re- higher CO leve strict requirem considered like plants may pro- be necessary reduction – re- other factors w give rise to ex emissions and	Eirculation air with re-c of reagents ery by retain cent applica els are not ely that FGI ove expens for existing venue cost when detern cessively re I corrosion	provides a irculated fi used for s ing heat fr ations indic as significa eve 6% ex R represer ive or impu plants to i savings fru- nining the educing co – they sho	In effective lue gases econdary form stack cates that ant as hat cess oxy ints BAT for actical do nclude co om reduce site spect nditions, uld there	ve means of s. It has the v NOx contr c gases. Th t concerns r d been exp gen may fa- or all new p ue primarily onsideratior ced reagent ific BAT. R incomplete fore be rega	f NOx pre addition ol (see b regarding ected. F cilitate th lant in th to the sp of FGR use sho ce-circula combus arded with	evention by r nal benefit of elow) and m ercial use of g additional c urthermore t is technique is sector. Re pace require when asses uld be consid tion rates gra tion, elevated th caution.	eplacing reducing ay increas this techn orrosion a he remov It is ther etrofits at d for ducti sing BAT dered alor ater than d CO and	10 to 20% the se overall ique and its arising from al of the refore existing ing. It will for NOx ng with 20% may VOC
		Secondary	NO _x me	asures						
		Secondary me measures out measures out will serve to re secondary trea	easures sho lined above lined above educe the pl atment stag	ould be cor . The use (including roduction o les.	nsidered of secon FGR) is of NOx, w	after the ap idary measi unlikely to which in turr	oplication ures with represen will redu	of primary N out applying t BAT as the uce reagent o	IOx reduct the prima primary t consumpt	tion Iry echniques ion during
	15.	Selective nor Injection of NH to nitrogen and BAT. When d potent greenh and is easier t retention time must therefore plant). (Cont.)	a catalytic i I ₂ -X comp d water. Ar osing is option ouse gas) h o handle. S must be pro- be optimis	reduction ounds into mmonia ar timised am nowever un SNCR relie ovided for ted (CFD r	(SNCR) the furn ad urea ir monia te rea may b rea may b the inject nodelling	ace reduce njection are nds to give be effective optimum ten ted agents to may be us	s NO _X er suitable rise to lo over a s mperatur to react v eful and	missions by o and either m ower nitrous lightly wider re around 90 with NO. Por is likely to be	chemically ay repres oxide forn temperatu 0 °C, and t injection e essentia	/ reducing it ent the nation (a ure window sufficient n locations il for all new

INTRODUC	INTRODUCTION		HNIQ	UES	E	MISSIO	NS	IMPACT		
Management Mat	erials puts	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Abatement to air		Poorly optimise should be mon ammonia slipp	ed reagent hitored and age.	injection n the additio	nay give on of rea	rise to eleva gent closely	ated emi controlle	ssions of am ed to minimis	monia. N se the pos	IO _X levels sibility of
BAT for NOx (cont.)		SNCR should SNCR will be r average NOx s average releas further NOx red beyond WID co large plant (over additional NOx	be used wi required to standards i ses are typ ductions ca ompliance er 250K te c reduction	here NO _X e ensure WI reagent injo ically in the an be achie therefore a /yr. waste f	emission ID standa ection ra e range c eved but appear u throughp	s are above ards are me te set points of 150 to 180 only with in nlikely to re ut) or where	WID rele t. In ordes are usu 0mg/m ³ . creasing present E e local er	ease levels. er to comply ally set such At higher re cost – reduc 3AT. This m avironmental	It is prob with WID that long agent dos ctions sigr ay not be condition	able that daily er term sing rates hificantly the case at s justify
	16.	Selective cata	alytic redu	ction (SCI	R)					
		SCR reduces I about 300-400	NO and NO °C. SCR t	D ₂ to N ₂ wit echnology	th the ad can also	dition of NH o reduce VC	l₃ and a o Cs, CO a	catalyst at a a a a a a a	temperatu missions	ure range of
		SCR is a prove where NOx em UK disposal ga possible that th	en technolo nissions of ate fees) cu nis situation	ogy in the v below 70m urrently ma n will chang	waste inc ng/m ³ are ike this to ge.	ineration se achieved. echnique ur	ector, par The cos nlikely to	ticularly in co ts of this tec be BAT for t	ontinental hnique (a he UK. H	Europe nd lower lowever, it is
		The additional required tempe gases to be rel the particulate significantly be	costs of S erature ran latively cle and acid g elow that re	CR are der ge. To ave an. This m jas filtration equired for	rived ma oid cataly neans tha n plant. ' SCR, the	inly from the yst poisonin at the SCR (With the exi e combustio	e energy g the SC unit must t tempera n gases	requirement R unit is req be positione ature from th must be reh	s of achie uires com ed down s le filtratior eated.	eving the abustion atream of a unit being
		All applications SCR and justif against the alte efficiency with	s must ther y if it is not ernatives (SCR owin	efore inclu employed e.g. SNCR g to gas re	ide in the l. Simila) focussi -heat.	ir cost bene dy the adop ng particula	efit asses tion of th irly upon	sment consi e technique the potential	deration o must also for reduc	of the use of be justified ced energy
	17.	NOx control:	cost/bene	fit study						
		Operators show demonstrate the comparison withe asset lives	uld provide ne relative Il show the and typica	e a cost be merits of p cost per to l discount	nefit stuc rimary m onne of I rates giv	ly using the leasures, Sl NO _X abated en in that de	methodo NCR and over the ocument	blogy in H1 (I SCR for the projected lif	Ref. 5), to e installati e of the p	on. The lant using
	Aci	d gases and	l haloge	ns						
BAT for acid gases and halogens		Techniques that below. The tech which provides waste stream t	at may rep chnique the a solution to give rise	resent BA ⁻ at represer for anothe to acid ga	T to minii hts BAT i er. This s emissi	mise acid ga n one incine will generall ons, their qu	as and ha eration su y relate t uantity ar	alogen releas ub-sector ma o the potenti nd variability.	ses are su ay be diffe al of the p	ummarised erent to that particular
		Primary ac	id gas m	ieasures	;					
	18.	Fuel Selection	n							
		Start up and su commonly ava down or as sup Art 1(1) of Dire combustion ter effectively prev	upport fuel ilable. The oport fuels ective 75/7 mperature vents the u	s should be waste inc which can 16/EEC), li to be main se of waste	e low in s cineratior cause h quefied g tained a es as a s	sulphur. Su n directive p igher emiss gas or natur t all times w start up fuel	lphur cor revents t ions thar al gas. I hen was – regard	ntents of belo he use of fue n those of ga n requiring th te is being bo less of speci	ow 0.2%w els at star s oil (as c ne relevar urned, WI fication.	/w are t up, shut lefined by nt D also
		Owing to the p cases be few of the installation design envelop create particula of this is large waste at munic expected to tal	rimary pur opportunitie should be oe below). ar difficultie quantities cipal waste ke whateve	pose of inc es to influe designed However, es at the in of PVC pla incinerato er steps ar	tineration nce relea to cope v it may b stallation stallatior stics or p ors. Whe e necess	a being the of ases throug with the type e the case t a or that the blaster boar re such pro ary to ensu	disposal h waste s e of waste hat a par waste st d where blems oc re compl	of waste, the selection. It e it is to rece ticular waste ream has ch they are not ccur, the Ope iance.	ere may, ii is fundam vive (see a e stream i anged. A well mixe erator will	n many hental that abatement s known to An example ed with other be

INTRODUC	CTIC	DN TE	CHNIQ	UES	E	MISSIO	I	IMPACT				
Management Ma	terials	Activities/	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation		
- 11	iputs	apatement	water							135065		
Abatement to air BAT for acid gases and halogens (cont.)		This may inclu Up stream Use of from Abatemen Abatemen These options the particular problematic fr	ude: n waste mar nt end wast it plant open it plant rede s are discus waste strea ractions.	nagement i re treatmer ration trimr esign and r ssed furthe am and dec	to prever nt technic ning; ebuild. r below. cisions re	nt the inclusi jues; The choser garding the	ion of pro	oblem waste will depend o reliably seg	s; upon the iregate the	nature of e		
	19.	Waste Treatr	nent									
		 Waste treatme Allowing th Homogeni This can assis Minimising Minimising 	ent techniq he removal ising the wa st by: g the quanti g the amour	ues may a of problem aste feed to ty of reage nt if waste	ssist with wastes; provide ent requir reagent r	for improve for improve ed to treat the equiring re-	releases ed proces he acid g circulatio	s of acid gas ss stability. gases; on or disposa	es by: al.			
		Waste treatme required wher	ent requirer	ments shou e exhibits s	uld be co such a he	nsidered at terogeneou	the insta s nature	Illation desig that:	n stage ai	nd will be		
	 Emission limit values are not be complied with, The abatement design envelope (see abatement design limits below) is required to be so wid that it would necessitate very large reagent consumption i.e. overdosing and waste production; Incineration without pre-treatment may be dangerous from an operational perspective. 											
	20.	Abatement D	esign Lim	its								
	Waste varies in terms of its physical and chemical nature depending upon its source and whet it has undergone any pre-segregation or treatment. All installations must therefore be clear at the types of wastes they intend to receive and their composition. Applications must very clear outline the composition of the types of waste that will be incinerated and demonstrate that the installation design takes the full range of likely compositions into account. Existing installation may be able to illustrate this with real data regarding waste types and emissions compliance.											
		In particular th variation in ra taken to ensu	ne abateme w flue acid re that shor	nt plant de gas conce t term fluc	esign env ntrations tuations a	elope must that will be are conside	be wide encount red.	enough to a ered. Partic	ccount for ular care i	⁻ the must be		
	The design of the acid gas abatement system must take full account of the flue gas loading and the reaction kinetics of the reagent selected in the conditions that will be encountered in the equipment. In-situ temperatures and moisture contents will have a key role in determining the residence time that is required to ensure effective acid gas neutralisation (and removal). Once the abatement plant design has been established, sufficient over-capacity should be provided to allow for maintenance or waste heterogeneity variables.											
		Applications n operation outs include consid	nust pay pa side the des deration of:	irticular att sign enveld	ention de ope and p	escribing ho possible exc	w waste eedence	will be mana of authorise	aged to proed limits.	event This shall		
	1	The bread	Ith of waste	compositi	on likely	to be encou	ntered ir	the waste t	ypes to be	e received;		
		 Identificati make refe 	ion of any p rence to an	articular w	astes wh	ich may cau	use high ulties wit	acid gas loa	ding – this cular sect	s should or:		
		Measures upstream	to be taker	to preven	t the inci	neration of t	the waste	es identified,	including	the		
		 Measures 	to treat or i	mix wastes	to ensu	re peaks are	e smooth	ned out;	,			
		Plant dete (see below	ction and c v).	ontrol mea	sures inc	cluded to de	al with s	hort term hig	h acid ga	s loading		
										Cont		

INTROD	UCTIO	N TECHN	QUES	E	MISSIC	NS		MPAC	Т
Management	Materials inputs	Activities/ Groun abatement wate	nd Pr Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Abatement to air BAT for acid gases and halogens (cont.)	21.	Secondary acid ga Abatement Type – v There are three main and dry scrubbing. E in different circumstar depending upon othe Generally wet scrubb values, particularly w conditions and low op species. Wet scrubb gas loading e.g.HWIs production of an efflu Dry and semi-dry sys outlined in Table 2-6 referring to the factor	as measure vet / dry / sen techniques fo ach has adva nces depende r decisions tal ing will provid here wastes a berating tempe ing is therefore s. Wet scrubb ent stream an tems are avai below. Opera s indicated.	s ni dry or the corn ntages a ent mainly ken in res e the gre irre of ver eratures, e well su ing does d of a we lable. Th ators will l	trol of acid nd disadvar upon the t spect of over atest secur varied cor enhances t ited to plant however ha t plume wh a advantag be required	gases, w ntages, a ype of wa erall proc ity in terr nposition he captu with exo ave disac ich may jes and c to justify	vet scrubbing nd each may aste being in ess design. ns of meeting n. This is bed re of the wat ceptionally hi dvantages re- require energ lisadvantage their selecte	, semi-dr / represer cinerated g air emis cause the er soluble gh or vari garding th gy input to s of these ed techno	y scrubbing nt the BAT , but also sion limit wet acid gas able acid ne o reheat it. are logy by
Table 2 6-		Abatement Type	Ac	dvantage	es		Disad	vantages	
Abatement type -		Wet	High re	eaction ra	ates		Large eff	luent disp	osal and

water consumption if not fully Good performance over treated for re-cycle range of loadings Effluent treatment plant Low reagent consumption required Low solid residues May result in wet plume production Energy required for effluent Reagent delivery may treatment and plume reheat bevaried by concentration Wet systems may and flow rate experience higher corrosion Condensation effect may assist with metals Pre-scrubbing particulate abatement removal may be required Dry Low water use Reaction rates low therefore • larger residence time Reagent consumption may required be reduced by recycling in Higher solid residue plant production Reagent delivery only by input rate Semi-dry Medium reaction rates Higher solid waste residues Medium water use In process reagent recycle not proven Reagent delivery may be varied by concentration and input rate

Guidance on the general suitability of each of these systems to different incineration sub-sectors is given in Section 2.3.9 below.

22. Alkaline Reagent Selection

Consistent low acid waste streams:

It may be possible for some waste streams of very consistent composition, that can be demonstrated to be reliably very low in halogens (e.g. well segregated non-halogenated waste solvent streams incinerated on the site of production) to be incinerated without alkaline scrubbing. Indeed, to do so where clearly not necessary is itself unlikely to be BAT owing to the unnecessary consumption of reagent. Water scrubbing only may be acceptable in these circumstances.

However, in general, provision for alkaline reagent injection will need to be made wherever the concentration of acid gases or acid gas forming materials in raw flue gases exceed the standardised flue gas emission limit concentrations outlined in Section 3.3.

Table 2 6-Abatement type advantages and disadvantages for acid gas control

INTRODUCTION		N TE	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

Abatement to air

BAT for acid gases and halogens (cont.)

Table 2 7-Reagent selection advantages and disadvantages for acid gas control

Other waste streams:

For the vast majority of waste streams the injection of alkaline reagents will be required to absorb acid gases and meet the required emission limit values. The most commonly used reagents are:

- Sodium Bicarbonate (NaHCO);
- Lime(Ca(OH)₂);
- Sodium Hydroxide (NaOH).

Although the basic stoichiometry of these reagents would indicate that lime, with two hydroxyl molecules per molecule of reagent would provide for more efficient removal of acid gases, sodium hydroxide is, in practice, the most efficient and lime the least. This is explained by the relative reaction rates achieved i.e. the lime / acid reaction takes longer to reach completion.

All of these reagents can be effectively used to secure the emission limit values outlined in Section 3.3, and may represent BAT in an individual situation. The advantages and disadvantages of these are outlined in Table 2-7 below. Operators will be required to justify their selected technology by referring to the factors indicated.

Reagent	Advantages	Disadvantages	Comments
Sodium Hydroxide	Highest removal rates Copes well with high acid load Low solid waste production	Effluent requires treatment Corrosive material ETP sludge for disposal	Suitable for HWIs and DIs
Lime	Very good removal rates Low leaching solid residue Copes well with medium acid loads Temperature of reaction well suited to use with bag filters Wet, dry and semi dry systems available	Corrosive material Some handling / pumping difficulties May give greater residue volume if no in-plant recycle	Wide range of uses
Sodium Bicarbonate	Good removal rates Easiest to handle Dry recycle systems proven	Effient temperature range may be at upper end for use with bag filters – ceramics required? Leachable solid residues Bicarbonate more expensive	Often used at CWIs Not proven at large plant

Guidance on the general suitability of each of these systems to different incineration sub-sectors is given in Section 2.3.9 below.

23. Alkaline Reagent Dosing Control

Optimisation of the alkaline reagent dosing system is the BAT. This is because a well optimised reagent doing control system will:

- · Control acid gas emissions within emission limit values;
- Reduce consumption of reagent;
- Reduce production of alkaline residues.

Optimisation in this context means delivering the right amount of reagent to absorb acid gases to meet emission limit values, without wasting reagent and producing excessive residues. The techniques that are considered BAT for securing this optimisation are:

- Trimming reagent dosing to acid load using fast response upstream HCI monitoring as a trigger;
- Ensuring reagent concentration can be rapidly changed through use of variable speed pumps
 / screw feeders and / or low volume intermediate silos (which will allow for more rapid
 concentration changes);
- · Small silo load cell systems provide close control on reagent delivery rates in dry systems;

INTRODUC	CTIC	ON TEC	HNIQ	UES	E	MISSIO	NS	I	MPAC	Т				
Management Ma	terials	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues				
						· · · · ·								
Abatement to air		 Good (prev Sufficient al when feed to 	entative) n bsorption t fails;	naintenano ouffer capa	ce of all r acity reta	eagent hand ined in abat	dling and ement sy	delivery equ /stem to mai	uipment; ntain abat	ement				
BAT for acid		multiple or I	back-up fe	ed system	ns on stai	ndby to mair	ntain rea	gent feed.						
halogens	24.	Acid gas cont	rol: Cost/	benefit st	udy									
		Operators shou demonstrate th installation with acid abated (as rates given in th	uld provide e relative i the altern G HCI) over hat docum	e a cost be merits of the atives out the projection ent.	nefit stud he select lined abo cted life o	ly using the ed primary a ove. The co of the plant u	methodo and seco mparisor using the	blogy in H1 (ndary meas n will include asset lives	Ref. 5), to ures for th the cost p and typica	e er tonne of I discount				
		As some techn overall viable ir regarding the re grounds over o	ological op Istallation easons wh verall inco	otions will design alte ny any app mpatibility	be mutua ernatives parently b	ally exclusive in relation t etter individ	e, it will b o that se ual proce	e acceptable lected, whils ess stages a	e to asses at providing re not sele	s the comment cted on				
BAT for other	Oth	Other releases												
releases	25.	Carbon oxides	s (CO ₂ , CO	D) and VO	Cs									
		Carbon Dioxic	le:											
	All measures that reduce fuel energy use also reduce the CO ₂ emissions. The selection, when possible, of raw materials with low organic matter content and fuels with low ratio of carbon content to calorific value reduces CO ₂ emissions. In this sector this is only relevant to the support fuels used. In general natural gas will be the preferred option. If not available low sulphur gas oil provides an alternative.													
		The global war releases arising wastes into (pri the following:	ming poter g from the marily) wa	ntial (GWF waste cor iter and Co	P) of the i nbustion O ₂ attent	nstallation w As it is the ion should n	vill be de purpose ot focus	rived mainly of an incine upon these	from the C erator to co releases b	CO ₂ onvert ut upon				
		 CO₂ equiva 10% of the SNCR reag 	lent releas GWP – the ent injectio	ses resultir ese may b on;	ng from N De minimi	I2O releases sed by appro	s – these opriate s	e can contrib election and	ute in the optimisati	order of on of				
		Improving in installations quoting the Section 2.7	nstallation 5. This ma net energy).	energy eff y be demo y productio	ficiency (onstrate l on per to	including rec by providing nne of waste	covery) v energy l e produc	vill prevent C balance (Sar ed, (see ene	CO ₂ releas nkey) diag ergy require	e by other rams and ements in				
		Carbon Mono	kide and \	/OCs:										
		Elevated CO el elevated releas comments mad	missions a ses of othe le in this s	re indicati r products ection are	ve of poor of poor therefore	orly controlle combustion e applicable	ed combu e.g. VOC to VOCs	istion and m Cs, NOx, Dic as for CO.	ay be indio oxins. The	cative of general				
		Carbon monox employed abat pulsed corona and would in an formation.	ide emissio ement tecl or re-burn ny event b	ons are no hniques. I technique e less pref	ot influence Reduction s but the ferable to	ced to any s ns in CO ma se are not k primary tec	ignifican iy be ach nown to chniques	t extent by th lieved using be used at a for the preve	ne convent catalytic o a commerc ention of C	ionally xidation, ial scale O				
		VOCs may be of solution.	removed to	o some ex	tent by n	neans of wet	t scrubbi	ng but they a	are liable t	o come out				
		Reductions in 0Ensuring th complied w	CO and VC e furnace ith, (see S	DC emission and comb ection 2.3.	ons may ustion re .4):	be achieved quirements	l by: outlined	earlier in this	s guidance	are				
		Securing consist 2.3.2 - Waste c	stent wast harging).	e feed cha	aracterist	cs (e.g. CV,	moistur	e) and feed	rates, <mark>(see</mark>	Section				

INTRODUC	ТІС	ON TECHNIQUES EMISSIONS IMPACT							
Management Mat	terials puts	Activities/ Ground Waste Energy Accidents Noise Monitoring Closure Installation issues							
Abotomont									
Abatement to air BAT for other releases		Starved air systems such as pyrolysis, semi-pyrolytic and gasification processes by their nature deliberately create combustible gases that will comprise high concentrations of CO and VOCs. These often highly noxious partially reduced gases will require subsequent oxidative combustion prior to release. This may be achieved in boilers, engines or gas turbines. Current evidence concerning the ability of these processes to meet the required standards is contradictory. In all cases Operators will therefore be required to demonstrate that the chosen combustion stage, either alone or in combination with a secondary combustion stage, will be capable of meeting the temperature and residence times outlined in this guidance, the WID and the relevant emission limit values.							
(cont.)		Starved air systems such as pyrolysis, semi-pyrolytic and gasification processes by their nature deliberately create combustible gases that will comprise high concentrations of CO and VOCs. These often highly noxious partially reduced gases will require subsequent oxidative combustion prior to release.							
		Current evidence concerning the ability of these processes to meet the required standards is contradictory. In all cases Operators will therefore be required to demonstrate that the chosen combustion stage, either alone or in combination with a secondary combustion stage, will be capable of meeting the temperature and residence times outlined in this guidance, the WID and the relevant emission limit values.							
	26.	Dioxins and furans							
		Although fitting carbon, or catalyst impregnated fabric filters, can abate the release of dioxins, the primary method of minimising releases is by careful control of combustion conditions. The gas residence times, temperatures and oxygen contents at the combustion stage must be such that any dioxins/furans should be efficiently destroyed, (see Section 2.3.4 - Furnace requirements).							
		Operators should also ensure that the conditions for <i>de novo</i> synthesis are avoided. This may be achieved by ensuring exit gas streams should be quickly cooled through the <i>de novo</i> temperature region between 450 °C and 200 °C. Where energy will be recovered, boiler design should consider this factor, (see Section 2.3.7 - Boiler design).							
		Dioxin/furan formation needs sources of organic materials and chlorine and thus the limiting of chlorine input may have some effect where this is possible. Where higher concentrations are unavoidable (e.g. HWIs) the prevention of dioxin releases will become a dominant factor in the plant design to an extent that the recovery of energy from the waste stream may be excluded in favour of rapid quench using water. Such quench systems must be designed to achieve a maximum exit temperature of 200 °C (in practice a temperature of approx. 70 °C is likely).							
		Dioxins tend to adhere to particulate matter and therefore <i>efficient particulate abatement</i> will remove dioxin/furans from the gas phase. Bag filters impregnated with catalyst specifically developed for the destruction of dioxins/furans are now commercially available and, where fabric filters are installed, should be used where the benchmarks in Section 3.3 cannot be otherwise achieved.							
		FGR, SNCR and SCR are all reported to assist in the prevention of dioxin formation and their destruction.							
		Carbon injection has a proven record of reducing dioxin emissions at a wide range of facilities for relatively little cost and is therefore BAT. The carbon is commonly injected into the gas stream with the acid gas abatement reagent, prior to retention upon filtration equipment.							
	27.	Metals							
		Operators are likely to have little control over the metal content of the wastes they receive. However, in the case of mercury (Hg) there is some scope for control at CWIs as the main sources would appear to be dental amalgum although some batteries may also contain Hg (and other cadmium). Up-stream waste segregation should be encouraged where releases approach emission limits.							
		Carbon injection gives reliable and effective mercury reductions (greater than 90% reduction) and should be incorporated wherever there is a risk that the waste may contain Hg. In practice it will be required for all wastes streams other than those from known reliable and consistent sources.							

Incineration

INTROD	UC	TION	TEC	HNIQ	JES	E	MISSIO	NS		IMPACT		
Management	Mate inp	erials / uts a	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	
Management Abatement to air BAT for other releases (cont.)	Mate	erials d uts a Fi A • • 28. M v arid of Ti ho at cc A H ty	Activities/ abatement or the majority ninimised. t HWIs addition Cold wet so Chemical co Sulphide ad ixed bed abso oke prior to we Vet plumes deally at condition f weather condition to wever the us bated sufficient old damp condi- liternatively me lowever, wher pically reduce	Ground water y of metals onal specia rubbing (v onversion Idition to g orbers are et scrubbir and, in se tions of ter ditions. ion is to re- se of energontly to ensu- ditions. oisture care e this is no ed to arour	Waste particulat al measure apour pha by addition as filtration reported to ng. rse well an vere case mperature educe wate y should b ure good c n be remove of an inher ad 70°C fo	Energy e abatem es may be se Hg is n of reage n system o prove e nd tend to s, loss of and mois er input to be balance lispersion ved by co ent part o	Accidents nent is the r e required f greatly redu- ents to wet s ffective for o ground ea light issues sture conter o the process ed against n but a limit poling and c of a wet scr ng purpose	Noise main mea or Hg aba uced betw scrubbers Hg remov sily. Add s. Theref nt that av ss. Alterr the bene ed visible condensat ubbing pr s), the dis	Monitoring ans of ensuring atement, inclue ween 200 an s to enhance val, as has the litionally there fore the gas roid saturation hatively heat fits gained. e plume may tion followed rocess (where sadvantage	Closure ng releas uding: d 100C) e capture ne additic e can be should be n under a can be a Plumes n be accep by rehea re temper would be	Installation issues es are on of lignite local visual e discharged a wide range dded, nust be otable in at. rature is the	
		ge A: re sy	eneration of a s a further opt educe load un ystem.	significan tion, to mir der extrem	t liquid effl nimise exp ne weathei	uent stre ense and conditio	am. I energy us ns rather th	e, Operat ian to ove	tors may wis er design a p	h to guar lume aba	antee to atement	

INTROD	N TE	CHNIQ	UES	EMISSIONS				IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Abatement	2.3.9 Review of commonly proposed techniques for control of
to air	releases to air – by plant type

This section provides a review of those techniques recently proposed or currently employed in each sub-sector of the incineration industry. They may represent BAT where emissions limits and other criteria are being achieved.

They are **provided as a guide only and DO NOT cover all aspects of each plant type**, only some particularly areas where they differ from one another or there are particular, sub-sector specific issues. **They should not be relied upon without referring to the rest of this guidance.**

Individual installations are still required to demonstrate:

- Site specific selection of BAT taking into account, for example, local environmental factors
- Ability to meet the minimum standards required by Directives, including emission limit values

2.3.9.1 Municipal waste incinerators

In order to prevent releases to air new UK plants are adopting:

- Waste selection by upstream removal of recyclable or provision of a waste treatment stage;
- Staged combustion with close combustion control techniques;
- Temperatures in the range 900 to 1000 °C and residence times of greater than 2 seconds;
- Flue gas re-circulation;
- Selective non-catalytic reduction (with either urea or ammonia injection);
- Rapid temperature reduction from 450 to 200 °C (for dioxin control);
- Feed forward control systems for alkaline reagent injection;
- Semi-dry lime injection systems;
- Carbon injection for Hg and dioxin absorption;
- Bag filters;
- Sufficient burnout techniques (including time) to maintain total organic carbon levels below 3% in bottom ash;
- Electricity generation using steam turbines with options for developing waste heat use.

2.3.9.2 Clinical waste incinerators

Modern CWIs although much smaller, are similar in design to MWIs. The common differences are:

- Waste pre-treatment is less used;
- · Waste management procedures must take account of potentially greater health and safety hazards;
- Combustion temperatures are higher and will be in excess of 1100 °C for hazardous waste;
- Dry acid scrubbing systems are sometimes adopted;
- Steam generation only may be favoured over electricity plus steam (usually where there is a high and dependable local steam demand).

2.3.9.3 Chemical waste incinerators

- · Waste pre-treatment and blending is often carried out;
- Waste management procedures, including handling and charging operations are often elaborate reflecting the hazardous nature of the waste;
- On site QA and laboratory operations are often provided;
- Combustion temperatures in excess of 1100 °C are used;
- Multi-stage wet or combined wet and dry scrubbing is the norm;
- Waste water treatment and re-circulation may not be maximised;
- Heat recovery is often limited to internal heat exchange and does not extend to full recovery.

2.3.9.4 Sewage sludge incinerators

• Fluidised beds are being adopted to assist in reducing NOx production – additional techniques may be required to achieve WID emission limit values owing to high waste feed nitrogen levels;

Candidate BAT for abatement of point sources to air by plant type

INTROD	UCTIO	N TEC	CHNIQ	UES	El	MISSIC	NS		MPAC	T			
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues			
Abatement to air	• C p fc	over dried sludg roduction. The prmation;	ges may be additiona	e auto-ther I water in v	mal and g wetter slug	generate sig dges may s	gnificant suppress	energy but c temperature	an also ir s reducin	ncrease NOx g NOx			
	• S	taged combus	tion, semi-	pyrolytic, g	gasificatio	n or fully py	rolytic m	nay be applie	d in this s	sub-sector;			
	• A	cid gas control	l issues are	e similar to	other sul	b-sectors b	ut genera	ally less of co	oncern.				
	2.3.9	9.5 Animal	carcass	and anin	nal rema	ins incine	erators						
	• N e	 Many of these plants will require substantial upgrades of abatement plant and monitoring equipment to meet the standards of WID; Sulphur levels in carcasses have lead to higher than expected acid gas releases meaning acid ga abatement is required; A rotary kiln plant already achieves below 200mg/m³ NOx; 											
	• S a												
	• A												
	 Carcass crushing using an auger to feed the furnace with the addition of limestone may help to avoid production of liquid slag which could leak from the furnace; Liquids are prone to leak from primary chambers – techniques should be used to prevent this or direct the liquids to the secondary combustion chamber (at one plant the secondary chamber is beneath the primary); The practice of using multiple smaller units of proven design may be beneficial com[pared with fewer, larger purpose built furnaces. 												
	2.3.9	9.6 Drum ir	ncinerato	ors									
	lssue requi	es are similar to irements.	o HWIs. T	he main co	oncern is	compliance	e with HW	ID emission	limits and	d monitoring			
	• L	ime injection a	nd bag filte	ers have be	een adopt	ted.							
	2.3.9	9.7 Pyrolys	sis and ga	asificatio	n plants	;							
	Pyrol techr that r	lysis and gasifi niques e.g. con may effect emis	cation plar nbustion ga ssions to a	nts utilise n as cleaning iir are:	nany of th g etc. The	e technique e key areas	es that a s where t	re common to here are add	common to waste combustion ere are additional techniques				
	• И	Vaste handling	g – more e	xtensive w	aste proc	essing may	y be requ	uired;					
	• V	Vaste charging	g – care re	quired to c	control or	eliminate a	ir ingress	3;					
	• R 0	Reactor contro	•I – waste j controllec	particle siz	e, reactor the gas,	temperatu liquid, solid	re and or I residue	xidant / fuel r split;	atio (gasi	fication			
	• F C(0	f uel gas clean ompounds (am f engines and g	i ng – pre-o imonia car gas turbine	combustior t be remov es;	red by we	to remove t acid scrub	bing) – r	ate (by filtration may be requi	on) and n red to pre	itrogen event fouling			
	• <i>E</i>	Elevated CO ar ombustion of fu	n d VOC le uel gas – re	<i>vels</i> are re esidence ti	eported wi	here engine also be a o	es and ga concern;	as turbines a	re used fo	or			
	• <i>L</i>	<i>iquid residue</i> ncountered. T	<i>treatment</i> his would l	t – is likely be less like	to be req ely where	uired where this produc	e elevate ct is put te	d levels of pl o some bene	henols an ficial use	d PAHs are ;			
	• S S 0	Solid residue t a Some plants inc f organic carbo	reatment - clude an inf on may be	 some spe tegral resid of concern 	ecificatior due vitrific in respec	ns of char m ation step - ct of compli	nay be pu – yielding ance wit	ut to bone fid g a highly sta h WID 3% T(e benefic ble residu DC limits.	ial use. .e. Levels			
	Furth Secti	ner details rega ions 2.3.3.13 a	nd 2.3.3.14	niques use 4, and in th	ed in the p ne <mark>BAT re</mark>	pyrolysis an port on was	d gasific ste pyrol	ation sub-seo ysis and gasi	ctor may ification is	be found in sues.			
	2.3.9	9.8 Refuse	derived a	fuel incin	erators								
	Tech many untre depe	niques are sim y of the feedsto eated wastes. and upon the pr	nilar to thos ock encour This should recise natu	e adopted ntered will t d mean tha nre of the F	at other i fall within at emissic RDF incine	incineration a more easons are more erated.	n installat sily define re easily	ions. As a re ed specificati controlled, al	efuse deri ion than r though th	ved fuel, nixed is will			
	220	9 Co-inci	nerators										

Specific abatement techniques for co-incinerators are not specified in this guidance. Information regarding the techniques that represent BAT for each sector in respect of the non-waste fuels use at these installations are provided in the sector specific guidance.

INTROD	UCTION	TEC	CHNIQ	UES	EMISSIONS				IMPACT		
Management	Materials Ac inputs ab	ctivities/ patement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	
 Abatement to air In considering candidate techniques for the co-incineration of wastes Operators should consider: The suitability of the abatement techniques outlined in this guidance to the particular industrial situation; 											
	 The emission limit value that must be met to comply with incineration directives (HWID and WID) – see emissions section for guidance on calculation of emission limits. Note that these represent the minimum standards and that BAT may require further emission reductions; 										
	 Waste composition and combustion characteristics in relation to existing fuels – in some cases wastes may reduce pollutant loading; 										

- Temperatures which the abatement equipment will be required to run at;
- Particulate sizes and pollutant loading.

INTRODUCTION T			CHNIQ	QUES EMISSIONS			IMPACT			
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Effluent	
treatment	

2.3.10 Abatement of <u>point source</u> emissions to surface water and sewer

The nature and source of the emissions expected from each activity will be confirmed in detail in the Operator's response to Section 3.1. In addition to the techniques below, guidance on cost-effective effluent treatment techniques can be found in ETBPP Guides (Ref. 8).

2.3.10.1 Legislative Requirements

European legislation (WID and HWID) set requirements in respect of:

- The design criteria for prevention of waste water releases (these are discussed in relation to overall plant design to minimise releases in Section 2.3.12 - Fugitive releases to water;
- Maximum emission limits for releases to water arising from air pollution control devices (see Section 3.4).

Following the use of techniques to reduce the production of pollutants it will be necessary for abatement techniques to be employed that will meet the emission limit value requirements of European Directives, *as a minimum.* An Operator will be required to demonstrate in their application that BAT has been employed. It will not be sufficient to simply design a plant so as to meet the emission limit values, without considering how further emission reductions may be obtained through the use of BAT. It is probable that the use of BAT will result in emissions considerably lower than those indicated by the legislation.

2.3.10.2 Nature of the effluent

Effluents may arise from incineration processes as follows:

- Air abatement equipment (e.g. wet scrubbers);
- Boiler blow-down;
- Cooling water discharges;
- Road drainage;
- Incoming waste handling areas;
- Raw material storage areas;
- Ash and other residue handling areas;
- On-site effluent treatment.

Good design should prevent the production of the majority of these effluents from being produced. Process and site infrastructure design should aim to prevent the contamination of rainwater by the effective segregation of site drainage from potentially contaminated areas.

Through the recycling of effluents produced to ash quench baths it is possible to eliminate any need for routine discharges of waste water (other than rain water) from the site. At the same time this method assists in preventing fugitive dust releases from ash storage and handling.

Techniques for Treatment of Scrubber Liquors

Whether scrubber liquors are to be re-used in the process or discharged, there is normally a need to separate out the pollutants captured. If this is not done, and the water is re-injected into the incinerator, the indestructible ones will simply build up in the circuit as they are repeatedly recycled. Treatment is typically as follows:

- basic treatment normally comprises neutralisation, flocculation, coagulation and settling. Settling is
 much more effective when techniques such as lammella plates are used (see Ref.11). Filtration
 may be necessary for separation of fine precipitates;
- for cadmium, mercury and other heavy metals, precipitating the metals either as hydroxides or sulphides followed by appropriate solids separation can remove up to 90% or more of most heavy metals but probably less than 70% of cadmium and nickel;
- the use of specialist complexing precipitation agents, such as TMT (trimercapto-s-triazine trisodium salt) can settle similar percentages and has the advantage of forming stronger bonds with the metals and therefore results in lower leaching. It is, however, more expensive;
- the settled solids should then be de-watered by filter, centrifuge or evaporation, to make them
 easier to handle and subsequently stabilise, prior to landfilling;

Contd/...

INTROD	UCTION	TE	CHNIQ	UES	E	MISSIC)NS		MPAC	Т
Management	Materials Act inputs aba	ivities/ tement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Effluent treatment	 organi preser destru remov not pre The lic water excha degree where revers 	ics, includ nt, are be ction. Th ed the tro event it; quor will, can supp nge resir e, the wa salts new e osmos	ding dioxins est removed his will also eated water however, s port this leve hs, microfiltr ter may be ed to be rer is are estab	, furans an by activate remove res- tiself could till contain s el of salinity ation and e recycled; noved, bec lished tech	d PAHs s ed carbor sidual me d be retur salts, in p v the trea evaporation ause of t niques	should be r n which car ercury. Alte rned to the particular cl ted water r on techniqu he nature o	neasured n be retur ernatively incinerato nlorides a nay be dis ues could of the reco	in the treate ned to the in- if the heavy or, where sal nd sulphates scharged. If be used. Or eiving water,	d effluent cinerator metals ha t concent s. If the re not, then nce treate evapora	t and, if for ave been ration does eceiving ion ed to this tion and
	Applicati Questior	on Form 1 2.3 (cor	nt.)	Effluent	Treatme	nt				
	With th	e Appli	ication th	e Opera	tor sho	ould:				
	1. supp sour	bly the ge	eneral Appli sions to wate	cation requ er and land	irements ; and in a	for Section addition	n 2.3 on p	age 39 to pr	event or r	educe point
	2. inclu syste	ide, wher em for th	re appropria e activity;	ite, off site	treatmen	t in the des	scription c	of the wastew	ater treat	tment
	3. prov it ca	ide, whe n be reus	re effluent is sed (e.g. by	s discharge ultrafiltratio	ed, a justi on where	fication for appropriat	not clean æ);	ing the efflue	ent to a le	vel at which
	4. desc perfo mea ensu treat cher 4.1 t appl	cribe mea ormance sures en ure that th ted efflue nicals in put need ies wheth	asures taken is delivered sure reliabil hey are con ent (including the environ to be under her treatme	n to increas I (there may lity?, heavy trolled all th g the make ment. The stood here nt is on- or	the the rel by be a bid metals a me time? -up of the se steps in order off-site;	iability with ological pla are measur etc.); ident e COD) and will be carr to demons	which the int suscepted only o ify the madd assessing ied out as trate that	e required cc otible bulking ccasionally – in chemical on nent of the fa s in response the controls	ontrol and or poisor - what teo constitue ate of the to Section are adeq	abatement ning – what chniques nts of the se ons 3.1 and uate. This
	5. iden guid impr	tify the to ance is a ovement	oxicity of the available, thi programme	e treated eff is should, u e;	fluent (se inless alr	e Section 2 eady in ha	2.10). Un nd, norma	til the Regula ally be carried	ator's toxi d out as p	city part of an
	6. when and	re there a the techr	are harmful niques prop	substances osed to red	s or level luce the p	s of residuation of residuation of residuation of the second second second second second second second second s	al toxicity pacts;	identify the	causes o	f the toxicity
	7. Dem resp oblig scru	nonstrate ect of the pation for bbers).	that the tec e preventior discharges	chniques se and contro to water) a	elected fu ol of disc arising fro	lfil the spe harges of v om air pollu	cific requi vater (see ition contr	rements of W e Section 3.4 rol equipmen	VID or HV - <mark>Standa</mark> t (e.g. we	VID in rds and t
	Indicati	ve BA1	T Require	ements						
BAT for effluent	1. The (see Regu	Operator item 3 ir ulator bu	r should cor n Section 2. t in any cas	nplete any 3) as an im e within the	detailed proveme e timesca	studies rec nt conditio le given in	uired inton to a tim Section 1	abatement escale to be 1.1;	or control agreed w	options ⁄ith the
	2. The • v • c • c • c • c • c s a s • s • s	following vater use contamina ultimately statutory a streatm should be and alkali streams, systems s	general pri should be ation risk of r, surplus wa and non-sta ent will be r used wher ine streams while dilutio should be e	inciples sho minimised process or ater is likely atutory obje nore efficie e possible . Also, biol n, by mixin ngineered t	ould be a and wast surface to need ctives). nt. How to avoid a ogical tre g stream to avoid e	pplied in se ewater reu water shou treatment Generally, ever, the pr adding furt eatment ca s, can assi effluent by-	equence t ised or re uld be mir to meet th effluent s roperties her chem n occasio ist treatme passing th	o control em cycled (see himised (see he requireme treams shoul of dissimilar icals, e.g. ne nally be inhit ent; he treatment	issions to Section 2 Section 2 ents of BA Id be kep waste stru- utralising bited by c plant.	o water: .2.3); 2.3.12); .T (and t separate eams waste acid oncentrated

Incineration

INTROD	UCTIC	N TEC	E	VISSIO	NS	IMPACT				
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Effluent treatment	3.	All emissions s (see Sections s reasonable cos this will be carr	should be c 3.2 and 4.1 st it should ried out in r	ontrolled,) but notir do so (se esponse t	as a mini ng that wh e Section to Sectior	mum, to av here BAT ca 1.1). Calc h 4.1.	roid a bre an delive ulations a	ach of water r prevention and/or mode	^r quality s or reduct lling to de	tandards ion at emonstrate
BAT for effluent (cont.)	4.	With regard to IPPC the preve be made at rea water, the adea substances mu found in Refere	BOD, the reaction or reasonable control of the second seco	hature of t eduction of ost should le plant to considere Ref. 11).	he receiv ⁵ BOD is a be carrie minimise d. Guida	ing water sl also subject ed out. Furt the emissio ince on trea	hould be t to BAT a thermore on of spe atment of	taken into ac and further r , irrespective cific persiste persistent s	ccount. Heductions of the re ent harmfu ubstance	However, in s which can ceiving ul s can be
	5.	 Where effluent particular demo- emission w substance the probabi- pumping st action plans activities su a suitable n the potentia event. 	is treated onstrating tent provide as treated to the rece ility of sewu ations, is a s in the evu uch as clea nonitoring al inhibitior	off-site at that: d at the se on-site, ba iving wate er bypass, icceptably ent of bypa ning or ev programm of any do	a sewage wage tre ased on r r; via storn low; ass, e.g. l ren shutti ren shutti e is in pla wnstrean	e treatment atment wor eduction of n/emergenc knowing wh ng down wh ace for emis n biological	works, th ks is as g load (no cy overflo nen bypa ssions to processo	ne above fac good as wou t concentrati ws or at inte ss is occurrir ss is occurrir sewer, takin es and action	tors appl Id be ach on) of ea rmediate ng, resche ng; g into coi ns plan fo	y in ieved if the ch sewage eduling nsideration or any such

INTROD	DUCTION TECHNIQUES EMISSIONS IMPACT												
Managamant	Materials Ac	ctivities/	Ground	Maata	Enormy	Accidente	Noico	Monitoring	Closuro	Installation			
Management	inputs <mark>ab</mark>	atement	water	Waste	Energy	Accidents	NOISE	wonitoning	Closule	issues			
Fugitives	2.3.11	Contro	ol of <u>fug</u>	<u>itive</u> en	nissio	n <mark>s to air</mark>							
	At many emissior sources	installation is. Details of fugitive e	s fugitive, o will be four emissions a	or diffuse, id in the ap are:	emission ppropriate	s may be m e sector gui	lore signi dance bu	ficant than p ut common e	oint sour xamples	ce of the			
	 open 	vessels (e	.g. the efflu	ent treatm	ent plant	:);							
	 stora 	ige areas (e	e.g. bays, s	tockpiles,	lagoons e	etc.);							
	 the log 	bading and	unloading	of transpo	rt contain	ers;							
	 trans 	ferring mate	erial from c	one vessel	to anoth	er (e.g. furn	ace, ladl	e, reactors, s	silos);				
	 conv 	eyor systen	ns;	. ,									
	 pipevent etc.); 	Nork and du	ictwork sys	stems (e.g.	. pumps,	valves, flan	ges, cato	chpots, drains	s, inspec	tion hatches			
	 poor 	building co	ntainment	and extrac	tion;								
	 poter 	ntial for byp	ass of aba	tement equ	uipment (to air or wa	ter);						
	 accio 	lental loss o	of containm	ent from f	ailed plar	nt and equip	ment.						
	Applica	tion Form		E uritiu									
	Questic	on 2.3 (cont	i.)	Fugitive	e emissio	ons to air							
	With th	he Applic	cation th	e Opera	tor sho	ould:							
	1 eur	only the ger	oral Applic	ation requ	iromonte	for Section	23000	age 30 for o	ontrol of f	uaitive			
	em	issions to a	ir; and in a	ddition,	liternents	IUI Section	2.3 011 p	age 39 101 C		uyilive			
	2. ide inc fug but	ntify, and w luding those itive release t need to be	/here possi e below, es es for each e understoo	ble quantif stimating th substance d here in c	fy, signific ne propor e; these s order to d	cant fugitive tion of total steps will be lemonstrate	emission emission carried that the	ns to air from ns which are out as in res controls are	n all relev attributat ponse to adequate	ant sources, ble to Section 3.1 e.			
BAT for	Indica	tive BAT	Require	ments									
fugitives	1. The (se Reg	e Operator : e item 3 in gulator but	should com Section 2.3 in any case	nplete any 3) as an im e within the	detailed proveme e timesca	studies requent condition	uired into to a time Section 1	abatement of escale to be	or control agreed w	options <i>i</i> ith the			
	2. Wh fug	ere there a itive emissi	re opportui ons to be s	nities for re ubmitted o	eductions on a regu	, the Permi lar basis.	t may rec	quire the upd	ated inve	ntory of			
	3. Du	st - The foll	lowing gen f skins and	eral techni	ques sho	ould be emp	loyed wh	iere appropri	ate:				
	•	avoidance	of outdoor	or uncove	red stock	piles (wher	e practic	able);					
	•	where una etc.;	voidable, u	se of spra	ys, binde	rs, stockpile	e manage	ement techni	ques, wir	ndbreaks			
	•	wheel and	road clean	ing (avoidi	ing transf	fer of pollution	on to wat	ter and wind	blow); minimis	ing drops:			
		mobile and	l stationarv	vacuum o	leaning:	(noting the	nighti ti	nergy needs	,	ing arops,			
	•	ventilation	and collect	tion in suita	able abat	ement equi	pment;						
	•	closed stor	age with a	utomatic h	andling s	system;	,						
	•	regular hou	usekeeping	J.	Ū								
	4. VO	Cs											
	•	When trans subsurface balance lin or an enclo	sferring vol filling via f les that tran osed syster	atile liquid filling pipes nsfer the van n with extr	s, e.g. Sl s extende apour fro action to	F, the follo ed to the bol m the conta suitable ab	wing tech ttom of th ainer beir atement	nniques shoune container, ng filled to the plant.	IId be em the use o e one bei	ployed – of vapour ng emptied,			
	•	Vent svste	ms should	be chosen	n to minim	nise breathi	ng emiss	ions (e.a. pre	essure/va	icuum			

 Vent systems should be chosen to minimise breathing emissions (e.g. pressure/vacuum valves) and, where relevant, should be fitted with knock-out pots and appropriate abatement equipment.

INTRODU	JCTIC	ON T	ECHNIQ	UES	E	MISSIO	NS		MPAC	Т
Management	Materials inputs	abateme	s/ Ground nt water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Fugitives	Od	our - See Se	ction 2.3.13							
	Par	ticulate rele t release. C	ases from ot Control meas	her proces ures to mi	ss opera nimise s	tions have such releas	the pote es are s	ential for sig ummarised	nificant below:	fugitive
	5.	Housekee	ping							
BAT for fugitives (cont.)		A high star paid to tho and unload standing. I drainage s	ndard of house se parts of the ling of vehicle Hard-standing ystem.	e-keeping s e plant whe s should or g areas sho	should be re proces nly be do uld be co	e maintained ss and wast ne in design onstructed v	d in all ar e materia nated are vith appro	eas with par als are store as provided opriate falls t	ticular att d or loade with prop o an ade	ention being ed. Loading per hard- quate
	6.	Materials	handling							
		The receip emission o	t, handling an f dust to the a	d storage c iir.	of all mate	erials shoul	d be carr	ied out so as	s to minim	ise the
		Stocks of c through su	lusty materials itable arrestm	s, should be ent equipm	e stored ient.	in silos or c	overed s	torage, vente	ed to air if	necessary
		Bulk ceme should be equipped v use of sucl	nt, clinker and vented to suita vith audible or n alarms shou	l quicklime able arrestr visual high Id be chech	should b nent plar n level ala ked regul	e stored in ht, for exam arms to war larly.	closed bi ple bag f n of over	uildings or si ilters. Storag filling. The c	los. Thes ge silos s correct op	e silos hould be peration and
		Seating of	pressure relie	f valves to	all silos s	should be c	hecked p	eriodically.		
		For emission loading etco recognised where prop	ons from ancil .), dust contai that in some perly designed	llary proces inment and cases, suc I, used and	ssing equ arrestme h as rem maintair	ipment (cru ent should b lote mineral ned, can be	shing, so be the pro conveyo an effec	creening, ble eferred optio ors, suppress tive alternativ	nding, pa n. Howe sion techr ve means	cking, ver, it is ìiques s of control.
		The transfe conveyor, g dusty mate minimise a capacity to for example at all times other mech	er of cement o gravity or pne rials, such as irborne dust e handle maxir e by fitting sid . Where dust nanical means	or lime shou umatic mea clinker, ins emissions. num loads e boards. y materials s) should be	IId be by ans or su side build Where c and shou Conveyo are conve e fully en	air slide, el itably enclo ings should onveyors ar uld be provi- r discharge veyed outsic closed and	evator, s sed belt be carrie re used, t ded with s should de buildir extracted	crew feeder, conveyor. T ed out so as they should I protection ag be arranged ngs, above g I to suitable	enclosec ransport of to prever be of suffi gainst wir to minim round con arrestmen	l chain of other nt or icient nd-whipping, nise free fall nveyors (or nt plant.
		Other mate aggregate, emissions	erials which m or coal shoul - for example	ay generat d be delive by damper	e airborn red, store ning or co	e dust emis ed and hand overing.	sions, fo dled so a	r example cr s to prevent	ushed ro or minimi	ck, coarse ise dust
		Road vehic emissions. control me	cles or rail wag If they are lo asures are pro	gons should aded with o ovided, for	d be load dry mater example	led in such rials, this sh extract ven	a way as ould be o tilation to	to minimise carried out w arrestment	airborne here loca plant.	dust il dust
		The packin fitted with e example ba pressure d	g of lime and extraction equ ag filters. Arre rop sensor an	cement inte ipment and estment pla id alarm.	o bags si I the disp ant should	hould be ca laced air du d be fitted w	rried out ucted to s <i>r</i> ith a failu	using purpo suitable arres ure warning o	se-desigr stment pla device, fo	ied plant ant, for or example a
		All spillage system. Pa support str	s should be cl articular atten uctures and re	leaned up p tion should oofs in orde	oromptly be paid er to mini	using, for e to preventir mise wind e	xample, ng and cle entrainme	a vacuum cle eaning up de ent of deposi	eaner or veposits of ted dust.	/acuum dust on
		In designin movement	g a new proce on site.	ess, consid	eration s	hould be giv	ven to a l	ayout, which	n minimise	es vehicle

INTRODU	ICTIC	ON TEC	CHNIQ	UES	Eľ	MISSIO	NS		MPAC	Т
Management N	/laterials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Fugitives	2.3	3.12 Contro ground	ol of <u>fu</u> dwater	<u>qitive</u> en	nissio	ns to su	rface v	water, se	wer an	d
	Ap Qu	plication Form estion 2.3 (cont.	.)	Fugitive	Emission	s to Water				
	Wi	th the Applic	ation th	he Opera	tor sho	ould:				
	1.	supply the ger emissions to w	eral Appli ater; and	cation requin addition,	irements	for Section	2.3 on p	bage 39 for c	ontrol of f	ugitive
	2.	identify, and q relevant sourc releases for ea need to be und	uantify wh es, estima ach substa derstood h	ere possibl ating the pro ance; these here in orde	e quantif oportion o steps wil r to demo	y, significar of total emis Il be carried onstrate tha	it fugitive sions wh l out as in it the cor	e emissions to nich are attrib n response to ntrols are ade	o water fr outable to o <mark>Section</mark> equate.	om all fugitive 3.1 but
	3.	Outline how th the prevention obligation for c	e techniqu and contr lischarges	ues selecte rol of fugitiv s to water).	d fulfil the e dischar	e general re ges of wate	equireme er (see S	nts of WID o ection 3.4 - S	r HWID ir Standards	n respect of and
	Inc	licative BAT	Require	ements						
	1.	Where there a fugitive emissi	re opportu ons to be	unities for re submitted o	eductions on a regu	, the Permi lar basis.	t may ree	quire the upd	lated inve	ntory of
	2.	Subsurface s	tructures	– the Ope	rator sho	ould:				
		establish a	nd record	the routing	g of all in:	stallation dr	ains and	subsurface	pipework	
		Identify all	subsurfac	e sumps ar	nd storag	e vessels; m nince etc	oro min	imiand and y	whore the	00.000
		 engineer s can be rea 	dily detect	ted, particu	larly whe	re hazardou	us (e.g. li	isted) substa	nces are	involved;
		 provide in pipework, statuti 	particular, sumps and	secondary d storage v	containn essels;	nent and/or	leakage	detection for	r such sul	osurface
		 establish a pressure te 	n inspectie sts, leak t	on and mai tests, mate	ntenance rial thickn	e programm less checks	e for all s or CCT	subsurface s V.	tructures,	e.g.
	3.	Surfacing – tl	he Operat	tor should:						
							e			

- describe the design(#), and condition of the surfacing of all operational areas;
 - have an inspection and maintenance programme of impervious surfaces and containment kerbs;
 - justify where operational areas have <u>not</u> been equipped with:
 - an impervious surface;
 - spill containment kerbs;
 - sealed construction joints;
 - connection to a sealed drainage system.

(# Relevant information may include as appropriate: capacities; thicknesses; falls; material; permeability; strength/reinforcement; resistance to chemical attack; inspection and maintenance procedures; and quality assurance procedures.)

4. Bunds

All tanks containing liquids whose spillage could be harmful to the environment should be bunded. For further information on bund sizing and design, see Ref. 11. Bunds should:

- be impermeable and resistant to the stored materials;
- have no outlet (i.e. no drains or taps) and drain to a blind collection point;
- have pipework routed within bunded areas with no penetration of contained surfaces;
- be designed to catch leaks from tanks or fittings;
- have a capacity which is the greater of 110% of the largest tank or 25% of the total tankage;
- be subject to regular visual inspection and any contents pumped out or otherwise removed under manual control after checking for contamination;
- where not frequently inspected, be fitted with a high-level probe and an alarm as appropriate;
- have fill points within the bund where possible or otherwise provide adequate containment;
- have a routine programmed inspection of bunds, (normally visual but extending to water testing where structural integrity is in doubt).



Odour

2.3.13 Odour

Application Form Odour control Question 2.3 (cont.) With the Application the Operator should: supply the general Application requirements for Section 2.3 on page 39 for odour control: and in 1 addition, where odour could potentially be a problem, the Operator should: 2. categorise the emissions as follows: a high level release which is expected to be acknowledged in the Permit - i.e. there will a. be an allowed release from the process (e.g. An odorous release from a stack or high level scrubber) and an element of BAT is adequate dispersion between source and receptor to prevent odour nuisance. The release will be allowed under the Permit but it is acknowledged that, under certain conditions, the plume may ground causing odour problems. Conditions in Permits are likely to be based on the actions to take when such events occur b. release should be preventable - i.e. releases can normally be contained within the site boundary by using BAT such as containment, good practice or odour abatement.

c. release is not preventable under all circumstances e.g. from a landfill or uncovered effluent treatment plant but potential problems are controlled by a programme of good practice measures;

- 3. for each relevant category, demonstrate that there will not be an odour problem from the emissions under normal conditions (see odour guidance).
- 4. for each relevant category, identify the actions to be taken in the event of abnormal events or conditions which might lead to odour, or potential odour problems (see odour guidance).
- 5. describe the current or proposed position with regard to any techniques given below or in Ref. 22.

Indicative BAT Requirements

- 1. The requirements for odour control will be sector specific and dependant upon the sources and nature of the potential odour. In general terms:
 - where odour can be contained, for example within buildings, the Operator should ensure that the maintenance of the containment and the management of the operations are such as to prevent its release at all times;
 - Where odour releases are permitted, (see examples above):
 - for new installations or significant changes, the releases should be modelled to demonstrate a low frequency of ground level concentrations above the odour threshold (or other threshold of acceptability). For occasions where weather conditions or other incidents are liable, in the view of the Regulator, to cause exceedences of the threshold of acceptability, the Operator should take appropriate and timely action, including shutting down the operations, to prevent further annoyance,
 - for existing installations, the same principle applies, except that where experience shows there to be no odour problem such modelling and actions will not be necessary.
- For complex installations, for example where there are a number of potential sources of odorous releases or where there is an extensive programme of improvements to bring odour under control, an odour management plan should be maintained. The Regulator may incorporate the odour management plan in the Permit.
- Substances present in emissions to air, which are known to be odorous, should be identified and quantified. Techniques should be employed to ensure that they are minimised to prevent them being noticeable outside the site boundary (Ref. 22).

INTRODUC	TIO	ION TECHNIQUES EMISSIONS IMPACT										
Management Mate	erials outs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Odour	4.	 Odour may an arrow materiand the stack flican occur, the stack; scrubber d 	rise from: al handling lue gas whe particularly lischarges i	, See the a ere there is where the	appropria s poor co ere is a c ar condit	ite box in Se mbustion co ondensing v	ection 2.3 ontrol or water vap	3.1 - Incomin poor dispers pour plume a	ig waste f ion. Poo at the disc	nandling; r dispersion charge from		
		 ash handlin 	ng.									
	5.	 Odour can be enclosing a enclosing a confining v ensuring th SSI); refrigeration ACI); regular cle design of a ensuring th ensuring g preventing MWI); chlorination drawing ain treating su 	e controlled odorous are odorous wa vaste to the nat putresci on of such w aning and (areas to fac nat the trans ood dispers anaerobic n of waters r from odor ch extracte ese techniqu	d by: eas (applic aste all the e designate ble waste waste which (for putreso illitate clear sport of wat sion at all t conditions being retu ous areas id air prior	able to a way to the ed areas is inciner h is to be cible was ning (all) aste and imes fror by aerat rned to s at a rate to releas I obviate	II); ne furnace ((aII); rated within e stored for stes) disinfer ; ash is in co m any relea tion, turning STW or in st which will e e to destroy the need fo	ACI, CW an agree longer th ction of w vered ve se points of waste torage (S ensure th v the odo r odour r	(I); ed timescale an an agreed vaste handlir hicles, where (all); e and short ti (SI); at odour is c urs - see bel nasking or c	(MWI, C\ d timesca ng areas (e appropr mescales aptured (ow. ounteract	WI, ACI, le (CWI, (all); iate (all); ; (SSI, all); and ;ants.		
	6.	Treatment of Using odorous which can be I the need to co Biofilters e.g. p availability. G been demonst gas treatment. Scrubbing for such as potas: hydroxide/hyd Carbon filters otherwise they disposed of. I fed to the furn and minimise The use of BA	captured of s air as furn handled this onsider prov peat or com as flow thro trated to be odour typic sium perma rogen pero are effectiv y can be ex f it cannot b ace, to des waste arisin	bodour hace air is a s way is ob vision for or hopost beds bugh the m very effect cally would anganate. xide and s e, especia pensive to be recovery troy the od ngs. revent offe given in Oc	an ideal w poviously I dour con can prov ledia nee tive in se use cour A 3-stag odium hy lly where run and ed then, lorous co	way of deali imited by the trol when the vide odour of eds to be slow and to be slow wage sludg ther current ge scrubbing vdroxide main the total qui lead to a sign preferably, preferably, ompounds, ro ours beyond	ng with. le needs le inciner control pr bw for the ge applica columns g sequen- ly be effe uantity of gnificant spent od recover the d the inst d Contro	The quantity of the furnation rator is not o rovided there em to be effe ations. Not s with acids of ce using sulf ective.	y of conta perating. e is suffici- ictive. The suited to of or oxidisin phuric aci needs to ent carbor ontent of t ndary.	minated air advantage is ent land ey have combustion g agents d, sodium s small, be should be he carbon		
	••	Industry (see I available Open the Regulator.	Ref. 22) alo rators shou	ong with inf Id use the	ormation above in	on dispers formation a	ion desig nd, if in c	in criteria. U loubt, discus	Intil this g s odour is	uidance is ssues with		

INTROD	UCTIO	N	TECHNIQUES			E	MISSIO	NS	IMPACT		
Management	Materials inputs	Activit abater	ties/ ment	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Groundwater protection legislation

Groundwater

2.4 Emissions to Groundwater

The Groundwater Regulations came into force on 1 April 1999 (see Appendix 2 for equivalent legislation in Scotland and Northern Ireland). An IPPC Permit will be subject to the following requirements under these Regulations.

- i. It shall not be granted at all if it would permit the direct discharge of a List I substance (Regulation 4(1)) (except in limited circumstances see note 1 below).
- ii. If the Permit allows the disposal of a List I substance or any other activity which might lead to an <u>indirect discharge</u> (see note 2 below) of a List I substance then **prior investigation** (as defined in Regulation 7) is required and the Permit shall not be granted if this reveals that indirect discharges of List I substances would occur and in any event conditions to secure prevention of such discharges must be imposed (Regulation 4(2) and (3)).
- iii. In the case of List II substances, Permits allowing direct discharges or possible indirect discharges cannot be granted unless there has been a prior investigation and conditions must be imposed to prevent groundwater pollution (Regulation 5).
- iv. The Regulations contain further detailed provisions covering surveillance of groundwater (Regulation 8); conditions required when direct discharges are permitted (Regulation 9); when indirect discharges are permitted (Regulation 10); and review periods and compliance (Regulation 11).

The principles, powers and responsibilities for groundwater protection in England and Wales, together with the Agency's policies in this regard, are outlined in the Environment Agency's document "*Policy and Practice for the Protection of Groundwater*" (PPPG) (see Ref. 23). This outlines the concepts of vulnerability and risk and the likely acceptability from the Agency's viewpoint of certain activities within groundwater protection zones.

- A **Prior investigation** of the potential effect on groundwater of on-site disposal activities or discharges to groundwater. Such investigations will vary from case to case, but the Regulator is likely to require a map of the proposed disposal area; a description of the underlying geology, hydrogeology and soil type, including the depth of saturated zone and quality of groundwater; the proximity of the site to any surface waters and abstraction points, and the relationship between ground and surface waters; the composition and volume of waste to be disposed of; and the rate of planned disposal.
- **B Surveillance** this will also vary from case to case, but will include monitoring of groundwater quality and ensuring the necessary precautions to prevent groundwater pollution are being undertaken.
- *Note 1* The Regulations state that, subject to certain conditions, the discharges of List I substances to groundwater may be authorised if the groundwater is "permanently unsuitable for other uses". Advice must be sought from the Regulator where this is being considered as a justification for such discharges.
- *Note* 2 List I and List II refer to the list in the Groundwater Regulations and should not be confused with the similar lists in the Dangerous Substances Directive.

Application Form Question 2.4 Identify if there may be a discharge of any List I or List II substances and if any are identified, explain how the requirements of the Groundwater Regulations 1998 have been addressed.

With the Application the Operator should:

- Meeting the requirements of the Groundwater Regulations
- 1. confirm that there are no direct or indirect emissions to groundwater of List I or List II substances from the installation, or
- 2. where there are such releases, provide the information and surveillance arrangements described in A and B above.

Under these Regulations the Permit may not be granted if the situation is not satisfactory, therefore, with the application, the Operator should supply information on list I and list II substances and if necessary, prior investigation and surveillance information:

INTROD	UCTIO	N T	FECHNIQ	UES	E	MISSIC	NS	I	MPAC	Т
Management	Materials inputs	Activities abateme	6/ Ground nt water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Groundwater	List	t I								
	1(1) Subied	t to sub-parag	raph (2) be	low. a su	bstance is i	n list I if i	t belonas to	one of th	e followina
		familie	s or groups of	substances	;- ;-			t bololige to		e lenethig
List I and List		(a) o	rganohalogen	compounds	s and sub	stances wh	nich may	form such co	ompound	s in the
II substances		а	quatic environ	ment;						
		(b) o	rganophospho	orus compou	unds;					
		(c) o	rganotin comp	ounds;						
		(d) s a o	ubstances whi quatic environ therwise be in	ch possess ment (incluo list II);	carcinog ding subs	jenic, mutag stances whi	genic or t ch have t	eratogenic p hose proper	roperties ties whicl	in or via the n would
		(e) n	nercury and its	compound	s;					
		(f) c	admium and it	s compound	ds;					
		(g) n	nineral oils and	l hydrocarb	ons;					
		(h) c	yanides.							
	2.	A subs on the	tance is not in basis of a low	list I if it ha risk of toxic	is been d city, persi	etermined I stence and	by the Ag bioaccur	ency to be ir nulation.	nappropri	ate to list I
	List	t II								
	1 -(1) A subs	tance is in list	II if it could	have a h	narmful effe	ct on aro	undwater an	d it helon	as to one of
		the far	nilies or groups	s of substar	nces:		ot on gro			go to one of
		(a) tł	ne following me	etalloids and	d metals	and their co	ompound	s:		
			Zinc	Tin			Copper			
			Barium	Nic	kel		Beryllium	I		
			Chromium	Boi	ron		Lead			
			Uranium	Sel	enium		Vanadiur	n		
			Arsenic	Col	balt		Antimony	/		
			Thallium	Mo	lybdenun	n	Tellurium	1		
				Silv	/er		P . (T			
		(b) b	locides and the	eir derivativ	es not ap	pearing in	list I;			1
		(c) s c it	ubstances whi ompounds liab unfit for huma	ch have a d ble to cause in consump	the form	ation of suc	the taste ch substa	or odour of (nces in such	water ar	ater, and nd to render
		(d) to	oxic or persiste ormation of suc	ent organic o ch compour	compoun nds in wa water inte	ds of silicor ter, excludi	n, and suing those	bstances wh which are bi	ich may o ologically	cause the harmless
		(e) ir	organic comp	ounds of ph	osphorus	s and elem	ental pho	sphorus:		
		(f) fl	uorides:	oundo or pr			onten prite	001101010,		
		(g) a	mmonia and n	itrites						
	(2)	A subs	tance is also i	n list II if-						
		(a) it	belongs to on	e of the fam	nilies or a	roups of su	bstances	set out in pa	araoraph	1(1) above:
		(b) it	has been dete	ermined by	the Agen	cy to be ina	appropria	te to list I un	der parag	graph 1(2);
		a	nd							
		(c) it p	has been dete ersistence and	ermined by d bioaccumu	the Agen ulation.	cy to be ap	propriate	to list II hav	ing regar	d to toxicity,
	3(*	1) The Se powers	ecretary of Sta s under paragr	te may revie aph 1(2) or	ew any d 2 (2).	ecision of tl	ne Agenc	y in relation	to the exe	ercise of its
	3(2	 The Separation 2) parage 	ecretary of Sta aph (1) above	te shall noti and it shall	fy the Ag be the d	ency of his uty of the A	decision gency to	following a r give effect to	eview un o that dec	der sub- cision.
	4	The Agent this Scenarios	gency shall from hedule in such ary available to	m time to tin manner as the public	me publis s it consic free of ch	sh a summa ders approp narge.	ary of the priate and	effect of its of shall make	determina copies of	ations under any such

INTROD	UCTIO	N TE	TECHNIQUES			VISSIO	NS	IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.5 Handling of Wastes Produced

This section relates to the management of wastes *produced at the installation*. This will mainly comprise ash. BAT for incoming waste (feedstock) management is covered in Section 2.3.1.

The normal nature and source of the waste from each activity is given in Section 2.3 and will be confirmed in detail in the Operator's response to Section 3.1 (emissions inventory).

In general the waste streams produced comprise:

- Bottom ash (approx. 25% by weight and 10% by volume of input for a modern MWI);
- Fly ash;
- Air pollution control residues (commonly combined with fly ash and then approx. 2.5% by weight of
 waste input for a modern MWI);
- Rejected feedstock wastes (chemical or physical incompatibility e.g. large objects);
- Recovered waste fractions e.g. steel and aluminium extracted from ash, or MRF recyclables

Application Form Question 2.5 Characterise and q proposed measure

Characterise and quantify each waste stream and describe the proposed measures for waste management storage and handling.

With the Application the Operator should:

- 1. identify and quantify the waste streams;
- 2. identify the current or proposed handling arrangements;
- 3. describe the current or proposed position with regard to the techniques below or any others which are pertinent to the installation;
- 4. demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures.

Indicative BAT Requirements

BAT for waste handling

- 1. A system should be maintained to record the quantity, nature, origin and where relevant, the destination, frequency of collection, mode of transport and treatment method of any waste which is disposed of or recovered.
- 2. Wherever practicable, waste should be segregated and the disposal route identified which should be as close to the point of production as possible.
- 3. Records should be maintained of any waste that is sent off-site (Duty of Care).
- 4. Storage areas should be located away from watercourses and sensitive boundaries e.g. adjacent to areas of public use and protected against vandalism.
- 5. Storage areas should be clearly marked and signed and containers should be clearly labelled.
- 6. The maximum storage capacity of storage areas should be stated and not exceeded. The maximum storage period for containers should be specified.
- 7. Appropriate storage facilities should be provided for special requirements such as for substances that are flammable, sensitive to heat or light etc., and incompatible waste types should be kept separate.
- 8. Containers should be stored with lids, caps and valves secured and in place. This also applies to emptied containers.
- 9. Storage containers, drums etc. should be regularly inspected.
- 10. Procedures should be in place to deal with damaged or leaking containers.
- 11. All appropriate steps to prevent emissions (e.g. liquids, dust, VOCs and odour) from storage or handling should be taken (see Sections 2.3.11, 2.3.12 and 2.3.13).

INTRODUC	CTIC	ON TE		UES	E	MISSIO	NS		MPAC	CT
Management Ma	terials puts	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
	Тес	chniques s	pecific to	o this sec	tor					
BAT for waste andling (cont.)	12.	Bottom ash become airbo suppression agglomerate use recovered	handling: orne. This sprays. Du the surface ed water wh	Where ash may be ach ust suppress of the ash here availab	n is handl nieved by sion spra without l le.	ed dry, the the quality ys should b eading to ru	method i of the co e limited un-off or	must ensure ontainment a to ensure th a leachate p	that dust nd/or by o ley moiste roblem, a	does not dust en and nd should
		Where handl drained befo the ash eithe quench tank conditions ar to the surfact	led wet, the re it is trans er during tra . Where ins re less impe e of land is	ash should aferred to sh nsport or a stallations h partive when permitted.	l be held kips or ot final dis ave an a re the asl	at an intern herwise lea posal. All w sh hopper, n is relativel	nediate p ves the s vater drai the wate y benign	point to ensur site, so that v ined should l r should be p , such as tha	re that it i vater will be returne oumped b at which c	s fully not drain off ed to the back. These lisposal on
		All ash trans	port contair	ners should	be cover	ed.				
		Adequate cle clean up pro should be fitt	eaning equi mptly any s ted with an	pment, suc pilled ash. absolute fill	h as a va With clin er. The	cuum clean ical waste a dry sweepir	ier, shou ash, in pa ng of spill	ld be provide articular, any lages is not a	ed and ma such vac acceptabl	aintained, to cuum cleane e.
	13.	Fly ash and produced as careful stora	APC resid a single str ge, handling	ues: Thes eam. Both g and trans	e two wa present portation	stes are cor potential ha , whether al	mmonly o zards the lone or ir	combined wi at may be mi n combination	thin the p inimised t n.	rocess and through
		Fly ash shou silo and cont Apart from th materials sho be handled it	Id be stored ainer filling ne minor us ould be kep n sealed co	d and trans , displaced e of dust su t dry to avo ntainers su	ported in air should ppressio id the for ch as tar	a manner ti d be ducted n sprays (us mation of le kers for lard	hat preve to suital sing reco achates	ents fugitive ble dust arres overed water . Dry residue ities or "big-t	dust relea stment eo where av es for disp pags" (1 r	ases Durin quipment. /ailable), dr posal shoul n ³) for

14. **Rejected Feedstock:** Efforts should be made to minimise the delivery of waste that cannot be processed at the facility (unless an appropriate authorisation is in place to permit the transfer of the waste). This will include up-stream waste management, provision of information regarding the types of waste acceptable and in some cases audit of waste suppliers procedures.

Despite these efforts some wastes will still be included with other wastes and delivered to the installation. Techniques should therefore be adopted for the inspection of the waste. These techniques should reflect

- the nature of the waste (including any potential additional hazards that might arise from waste inspection that may limit or prevent inspection)
- the history of the particular installation in respect of loads and sources of loads which may require special attention
- the ability of the installation to treat the waste and its operational design envelope (including any pre-treatment / waste mixing carried out)

Provision should be made for the safe storage of rejected loads in a designated area with contained drainage, preferably under cover. Procedures should be in place for dealing with such loads to ensure that they are safely stored and despatched for onward disposal. Storage times should be minimised.

Examples of loads which have caused difficulties at some plants have included:

- Large quantities of PVC window frames (high HCl loading)
- Large quantities of plaster board (high sulphur loading)

smaller installations.

- Large quantities of excessively wet waste (high moisture, low CV)
- Large quantities of iodine or mercury (particularly at HWIs)
- Some wastes containing wire which may jam loading systems or grates e.g. whole tyres, sprung mattresses or sofas
- · Large wastes that are not suited to incineration e.g. engine blocks
- 15. **Recovered Waste Fractions:** Provision should be made for the storage of all recovered fractions. The storage provided should take account of the general guidance given in this section.

INTROD	UCTIO	N TE	TECHNIQUES			MISSIO	NS	IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.6 Waste Recovery and Disposal

This section deals with the recovery and disposal of wastes *produced by* the installation. Section 2.2.2 summarises the main techniques for minimising waste arising at the installation (these are also outlined in Section 2.3 where applicable to particular activities)

Where waste production cannot (by the application of BAT) be avoided, the options will be for the recovery of the waste or its disposal.

The Regulations require the Regulator, in setting Permit conditions, to take account of certain general principles including that the installation in question should be operated in such a way that "waste production is avoided in accordance with Council Directive 75/442/EEC on waste; and where waste is produced it is recovered, or where this is technically or economically impossible it is disposed of, while avoiding or reducing the impact on the environment". The objectives of the National Waste Strategies should also be considered.

In order to meet this requirement the Regulator needs Operators to provide the information below.

Application Form Question 2.6 Describe how each waste stream is proposed to be recovered or disposed of; and if you propose any disposal, explain why recovery is technically and economically impossible and describe the measures planned to avoid or reduce any impact on the environment.

With the Application the Operator should:

- describe, in respect of each waste stream produced by the installation, whether the waste in question is to be recovered or disposed of, and if a disposal option is planned, to justify why recovery is "technically and economically impossible" together with "the measures planned to avoid or reduce any impact on the environment";
- 2. include in the description, the Operator's view as to whether waste disposal is likely to be restricted by the implementation of the Landfill Directive;
- 3. describe the current or proposed position with regard to the techniques below or any others which are pertinent to the installation;
- 4. demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures;

Indicative BAT Requirements – waste disposal/ recovery

- Unless agreed with the Regulator to be inappropriate, the Operator should provide a detailed assessment identifying the best practicable environmental options for dealing with the waste produced at the installation. For existing activities, this may be carried out as an improvement condition to a time scale to be approved by the Regulator.
- 2. The Operator should compare their proposed (or actual for existing installations) practice with that described in the BPEO assessment and justify any departures from the BPEO, whilst demonstrating that BAT for the prevention and treatment of wastes has been adopted.
- 3. Because ash will often be the major waste produced, in their BPEO the Operator should considering at least the following:
 - opportunities for bottom ash recycling e.g. bottom ash use as aggregate;
 - opportunities for fly ash re-use e.g. as a neutralising agent (care must be taken to avoid remobilisation of pollutants).
- 4. The Operator shall regularly audit the waste disposal /recovery routes to ensure their waste is being properly dealt with.

BAT for waste minimisation, recovery or disposal

INTRODUCTION		N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

2.7 Energy

BAT for energy efficiency under the PPC Regulations will be satisfied provided the Operator meets the following conditions:

either

the Operator meets the basic energy requirements in Sections 2.7.1 and 2.7.2 below and is a
participant to a Climate Change Agreement (CCA) or Trading Agreement with the government

or

• the Operator meets the basic energy requirements in Sections 2.7.1 and 2.7.2 below and the further sector-specific energy requirements in Section 2.7.3 below.

Note that even where a Climate Change Agreement or Trading Agreement is in place, this does not preclude the consideration of energy efficiency as part of an integrated assessment of Best Available Techniques in which it may be balanced against other emissions.

Further guidance is given in the Energy Efficiency Guidance Note (Ref. 13).

2.7.1 Basic energy requirements (1)

Application Form Question 2.7 (part 1) Provide a breakdown of the energy consumption and generation by source and the associated environmental emissions. The requirements of this section are basic, low cost, energy standards which apply whether or not a Climate Change Agreement or Trading Agreement is in force for the installation.

With the Application the Operator should:

1. provide the following Energy consumption information:

Energy consumption information should be provided in terms of delivered energy and also, in the case of electricity, converted to primary energy consumption. For the public electricity supply, a conversion factor of 2.6 should be used. Where applicable, the use of factors derived from onsite heat and/or power generation, or from direct (non-grid) suppliers should be used. In the latter cases, the Applicant shall provide details of such factors. Where energy is exported from the installation, the Applicant should also provide this information. An example of the format in which this information should be presented is given in Table 2-8 below. The Operator should also supplement this with energy flow information (e.g. "Sankey" diagrams or energy balances) showing how the energy is used throughout the process.

(Note that the Permit will require energy consumption information to be submitted annually)

	Energy consumption							
Energy source	Delivered, MWh	Primary, MWh	% of total					
Electricity*								
Gas								
Oil								
Other (Operator to specify)								

* specify source.

2. provide the following Specific Energy Consumption information

The Operator should define and calculate the specific energy consumption of the activity (or activities) based on primary energy consumption for the products or raw material inputs which most closely match the main purpose or production capacity of the installation. The Operator should provide a comparison of Specific Energy Consumption against any relevant benchmarks available for the sector.

provide associated environmental emissions
 This is dealt with in the Operator's response to Section 3.1.

Table 2.8 -Example breakdown of delivered and primary energy consumption

INTROD	N TEO	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.7.2 Basic energy requirements (2)



INTRODUCTION TEC			CHNIQUES		EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Table 2.9 -
Example Format
for Energy
Efficiency
Measures

Energy efficiency	NPV	CO ₂ saving	gs (tonnes)	NPV/CO ₂ saved	Priority* for implementation	
option	£k	annual	lifetime	£/tonne		
7MW CHP plant	1,372	13,500	135,000	10	high	
High efficiency motor	0.5	2	14	35	medium	
Compressed air	n/a	5	n/a	n/a	immediate	

* Indicative only, based on cost/benefit appraisal:

Where a Climate Change Agreement or Trading Agreement is in place, the Energy Efficiency Plan should be submitted as an improvement condition to a timescale to be agreed with the Regulator but in any case within the timescale given in Section 1.1.

5. **Energy management techniques** should be in place, according to the requirements of Section 2.1 noting, in particular, the need for monitoring of energy flows and targeting of areas for reductions.

Indicative BAT Requirements

- BAT for energy
- 1. **Operating, maintenance and housekeeping measures** should be in place, according to the checklists provided in Appendix 3 of the Energy Efficiency Guidance Note, in the following areas as applicable:
 - air conditioning, process refrigeration and cooling systems (leaks, seals, temperature control, evaporator/condenser maintenance);
 - operation of motors and drives;
 - compressed gas systems (leaks, procedures for use);
 - steam distribution systems (leaks, traps, insulation);
 - space heating and hot water systems;
 - lubrication to avoid high friction losses;
 - boiler maintenance e.g. optimising excess air;
 - other maintenance relevant to the activities within the installation.
- 2. **Basic, low cost, physical techniques** should be in place to avoid gross inefficiencies; to include insulation, containment methods, (e.g. seals and self-closing doors) and avoidance of unnecessary discharge of heated water or air (e.g. by fitting simple control systems).
- 3. **Building services** energy efficiency techniques should be in place to deliver the requirements of the Building Services Section of the Energy Efficiency Guidance Note. For energy-intensive industries these issues may be of minor impact and should not distract effort *from* the major energy issues. They should nonetheless find a place in the programme, particularly where they constitute more than 5% of the total energy consumption.
- 4. **Energy management techniques** should be in place, according to the requirements of Section 2.1 noting, in particular, the need for monitoring of energy flows and targeting of areas for reductions.
| INTROD | UCTIO | N TEC | TECHNIQUES | | | EMISSIONS | | | IMPACT | | | |
|------------|---------------------|--------------------------|-----------------|-------|--------|-----------|-------|------------|---------|------------------------|--|--|
| Management | Materials
inputs | Activities/
abatement | Ground
water | Waste | Energy | Accidents | Noise | Monitoring | Closure | Installation
issues | | |

2.7.3 Further energy efficiency requirements



INTRODUC	TIC	DN TE	CHNIQ	UES	EN	AISSIO	NS	IMPACT			
Management Mat	terials	Activities/	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation	
	puts	abatement	Water							135065	
BAT for energy (cont.)		 Effective Preventi Ensuring Use of id 	maintenar on of unco plant layo on exchang	nce of heantrolled a ut avoids ne in stean	at exchang ir ingress b pumping a d of high p	ers to main by providing and heavy t ressure me	tain high and ma ransfer v mbrane i	n heat transfe intaining sea vhere possib filtration for b	er Is Ie poiler (and	other water)	
	3.	Irrespective there are other	of whether her BAT co	a Climat	e Change ons involve	Agreement ed, such as:	or Tradi	ng Agreeme	nt is in pla	ce, where	
		 the choic where the conflicts 	ce of fuel in e potential with energ	npacts up minimisa y efficien	oon emission ation of was cy requirer	ons other th ste emission nentsl	an carbo ns by reo	on e.g. sulph covery of ene	ur in fuel; ergy from v	waste	
		 where the jeopardise dioxin george releases 	e nature of sed by add eneration – to air, and	f the was itional en see Sec 2.3.4 - F	te is such t ergy recov tion 2.3.7 - furnace rec	hat the prin ery (e.g. the Boiler designification puirements.	nary con e need fo gn, 2.3.8	cern of safe or rapid cooli 8.3 - Control o	waste disp ng to prev of point so	oosal may be ent <i>de novo</i> purce	
		the Operato	r should pr	ovide jus	tification th	at the prop	osed or o	current situat	ion repres	ents BAT.	
	Sub	-Sector sp	ecific iss	ues							
	4.	Municipal v	vaste incir	neration	.						
		Steam s	hould be g	enerated	for either o	lirect use of	electric	ity generation	n		
		• vvnere e 100,000	tonnes of a	annual wa	erated at le aste throug	east 910100 c Ihput	t electric	city should be	e recovere	a per	
		Waste h (this will explored	eat should require cos	be recov st justifica	ered unles ation). All o	s to do so c opportunitie	an be de s for CH	emonstrated P and distric	not to rep t heating s	resent BAT should be	
		The sitin recovery	g of plant r potential.	near to po Conside	otential or a ration shou	actual energ	y users to joint	will aid the m venture proje	naximisatio	on of ever possible.	
		 If waste upon, pa 	heat is not iss-out valv	recovere /es shoul	d, in order d be provid	to enable a led	ny future	e waste heat	use to be	capitalised	
	5.	Hazardous	waste inc	ineration	ı						
		 There ar heat via 	e likely to b exchange	be opport systems	unities for	internal ene	ergy savi	ng using con	nbustion g	enerated	
		The incir highly va grounds from suc polluting	neration of triable was that safe ir th wastes v emissions	higher co te stream ncineratio vill need t	oncentratio ns will e ab on is the pri to demonst	ns of haloge le to justify mary purpo rate that thi	enated o lower lev se. Inde s will no	r highly therr vels of energy eed proposal t give rise to	nally stabl y recovery s to recov higher lev	le wastes or on the er energy rels of	
		Low CV significa supporte	waste strea nt measure ed by high s	ams (e.g. es to reco suppleme	typically a ver energy entary fuel	queous or i on accoun firing.	norganic t of the lo	c wastes) are ow potential	also less recovery,	likely to merit unless	
		Consister maximis worth of	ent high CV ed. This m considerat	′ waste st lay involv ion where	treams offe re steam oi e high quai	r significan electricity entities are in	t energy generatio icinerate	recovery pot on and is par d.	ential that ticularly lil	t should be kely to be	
		 Where the possibility 	ne installati ies for prov	on is situ /ision of p	ated on or process ste	near other eam or heat	potential ing.	energy user	s there m	ay be	
	6.	Clinical wa	ste incinei	ration							
		Installati	ons will ge	nerally be	e expected	to generate	e steam t	for local use	or electric	ity generation	
		Lack of 2 be less f	24 hour ope avourable.	eration in	some case	es may mea	an the re	venues for e	xportation	schemes will	
		Where h for HWIs	azardous v should be	vastes ar consider	e incinerat red.	ed, the issu	es relati	ng to safe de	estruction	made above	
										Cont	

				-				IN		т			
INTROD		Activities/	Fround	-0		1133101	13		VIPAC	I Installation			
Management	inputs	abatement	water W	aste	Energy	Accidents	Noise	Monitoring	Closure	issues			
	_												
	7.	Sewage sludg	e incinerat	tion									
		There woul	d appear to	be co	nsiderable	scope for e	nergy rec	overy by me	ans of ele	ectricity			
		generation	and neat pr	OVISIOI a.a. (dri	n at sewag	je sludge inc	ineration	sites owing	(O)				
		- The like	libood in m	anv ca	eu) seway	e siuuyes, onsistent fee	detock:						
		- Hiah de	mand for el	ectrica	al power fo	r pumping ol	perations	at the treatn	nent work	s:			
		- Potentia	al use of hea	at for p	process he	ating;				,			
		- Land av be used waste h	vailability for l as a fuel, b eat).	integi balanc	rated syste ed with na	ems and CHI tural gas inte	⊃ (e.g. ga egrating v	as from anae with post stea	robic dige am turbine	esters may e incinerator			
		There may efficiency e	be energy r lectricity ge	ecove neratir	ery gains to ng equipm	be made fro ent can be u	om using tilised (a	pyrolysis or nd emission l	gasificatio imits met	on if high).			
	8.	Animal remain	ns and anin	nal ca	rcass inc	ineration							
		 Installations generation; 	s will genera	ally be	expected	to generate	steam fo	r local use or	electricit	y			
		Lack of 24 be less favo	hour operat ourable;	ion in :	some case	es may mear	n the reve	enues for exp	ortation s	chemes will			
		Rural locati	ons may ma	ake it i	more diffic	ult to find en	ergy outle	ets.					
	9.	Refuse derived fuel installations											
		Levels of enconsistent f	nergy efficie fuel should p	ency ai provide	re expecte e stable or	d to be the h perational co	high for th nditions;	ne sector bec	ause the	relatively			
		Electricity g	eneration is	s antic	ipated in a	Il cases (and	l / or stea	am raising);					
		CHP and d	istrict heatin	ng sho	uld be pro	vided (unless	s excessi	vely costly);					
		Integration	with other e	energy	users is e	xpected for a	all new in	stallations.					
	10.	Pyrolysis and	gasificatio	on inst	tallations								
		Installations engine gen	s may be ab eration tech	ole to ii inolog	ncrease el y (provideo	ectrical gene d emission lir	eration th mits can	rough the use be met);	e of gas t	urbine or			
		All products than a fuel)	s (solid char should hav	s, oils e their	and fuel g r energy p	as), which w otential maxi	vill not be mised ra	used as a pi ther than be	imary pro disposed	oduct (rather of:			
		- Char ma	ay be comb	usted	or its ener	gy released	using a v	vater gas rea	ctor;				
		- Oils ma	y be combu	isted;									
		- Fuel ga	s may be co	ombus	sted.								
	11.	Co-incineratio	on installati	ions									
		Refer to en	ergy aspect	s deta	iled in sec	tor specific g	guidance						

INTROD	UCTION	V TEC	HNIQU	JES	EMISSIONS			IMPACT			
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

2.8 Accidents and their Consequences

Guidance

IPPC requires as a general principle that necessary measures should be taken to prevent accidents which may have environmental consequences, and to limit those consequences. This section covers general areas of any installation operations which have the potential for accidental emission.

Some installations will also be subject to the Control of Major Accident Hazards Regulations 1999 (COMAH) (see Appendix 2 for equivalent legislation in Scotland and Northern Ireland). There is an element of overlap between IPPC and COMAH and it is recognised that some systems and information for both regimes may be interchangeable.

The COMAH regime applies to major hazards. For accident aspects covered by COMAH, reference should be made to any reports already held by the Regulator. However, the accident provisions under IPPC may fall beneath the threshold for major accident classification under COMAH and therefore consideration should be given to smaller accidents and incidents as well. Guidance (see Ref. 18), prepared in support of the COMAH Regulations may also be of help to IPPC Operators (whether or not they are covered by the COMAH regime), in considering ways to reduce the risks and consequences of accident.

General management requirements are covered in Section 2.1. For accident management, there are three particular components:

- identification of the hazards posed by the installation/activity;
- assessment of the risks (hazard x probability) of accidents and their possible consequences;
- implementation of **measures to reduce the risks** of accidents, and contingency plans for any accidents that occur.

Application Form Question 2.8 Describe your documented system that you proposed to be used to identify, assess and minimise the environmental risks and hazards of accidents and their consequences.

With the Application the Operator should:

- 1. provide the accident management plan described in the indicative BAT requirements below describing the current or proposed position with regard to the techniques listed below or any others which are pertinent to the installation;
- demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures;
- 3. identify any issues which may be critical.

Indicative BAT Requirements

- 1. A structured accident management plan should be submitted to the Regulator which should:
 - **a.** *identify the hazards* to the environment posed by the installation. Particular areas to consider may include, but should not be limited to, the following:
 - transfer of substances (e.g. loading or unloading from or to vessels);
 - overfilling of vessels;
 - failure of plant and/or equipment (e.g. over-pressure of vessels and pipework, blocked drains);
 - failure of containment (e.g. bund and/or overfilling of drainage sumps);
 - failure to contain firewater;
 - making the wrong connections in drains or other systems;
 - preventing incompatible substances coming into contact;
 - unwanted reactions and/or runaway reactions;
 - emission of an effluent before adequate checking of its composition has taken place;
 - steam main issues;
 - vandalism.

INTRODUC	CTION	TEC	HNIQ	UES	EN	/ISSION	S	11	MPAC	Т
Management Ma	terials A	ctivities/	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation
		atement	water							issues
BAT for control of accidents	b.	Assess viewed	s the risks as addres	s - having sing six t	identified	the hazards, tions:	the proce	ess of asses	sing the ri	isks can be
(cont.)		1. wha	t is the es	timated p	orobability	of their occuri	rence? (S	Source frequ	ency);	
		2. wha	t gets out	and how	much? (F	Risk evaluatio	n of the e	event);		
		3. whe rece	re does it ptors?);	get to? (I	Predictions	s for the emiss	sion – wł	hat are the p	athways a	and
		4. wha	t are the o	conseque	ences? (Co	onsequence a	assessme	ent – the effe	ects on the	e receptors);
		5. wha envi	t are the c ronment);	overall ris	ks? (Dete	rmination of t	he overa	II risk and its	significar	nce to the
		6. wha acci	t can prev dents and	vent or reduction	duce the ri e their env	sk? (Risk ma vironmental co	nagemer onsequer	nt – measure nces).	s to preve	ent
		The dep its locat	oth and ty ion. The	pe of ass main fact	essment w	ill depend on should be tak	the char en into a	acteristics of ccount are:	f the insta	llation and
		 the sactive 	scale and /ities;	nature of	f the accide	ent hazard pro	esented	by the install	ation and	the
		 the 	risks to ar	eas of po	pulation a	nd the enviror	nment (re	eceptors);		
		 the diffic 	nature of t culty in de	the install ciding an	lation and d justifying	complexity or the adequac	otherwis y of the i	e of the activitisk control te	vities and echniques	the relative
	с.	identify	v the tech	niques n	necessary	to reduce th	e risks i	ncluding:		
		c1. the	e following	g techniqu	ues are rel	evant to most	t installat	ions:		
		•	an inven which co forgotter damagin destroy i changes	tory shou build have in that man ing if they its ecosyst to the in	uld be mair environme ny apparer escape (e. stem). The ventory;	ntained of sub ental consequ ntly innocuous g. a tanker of Permit will re	ostances, iences if s substar milk spil equire the	present or li they escape nces can be led into a wa e Regulator	kely to be . It shoule environme atercourse to be notif	 present, d not be entally could fied of any
		•	procedu compatil contact;	res shoul bility with	d be in pla other subs	ce for checkin stances with v	ng raw m vhich the	aterials and y may accid	wastes to entally co	ensure me into
		•	adequat provided	e storage I;	e arrangem	ents for raw r	materials	, products ar	nd wastes	should be
		•	to ensur be given systems readings process	e that cor to proce based or s such as paramete	ntrol is mai ss design n micropro ultrasonic ers;	ntained in em alarms, trips a cessor contro gauges, high	nergency and other of and pas -level wa	situations, c r control asp ssing valve c irnings and p	onsiderat ects, e.g. control, tai process in	ion should automatic nk level Iterlocks and
		•	preventa from the	ative tech moveme	niques, su ent of vehic	ch as suitable les, should be	e barriers e include	to prevent o d as approp	lamage to riate;	equipment
		•	appropri containn	ate conta nent;	inment sh	ould be provid	ded, e.g.	bunds and c	atchpots,	building
		•	techniqu tanks (lic high-leve	les and pl quid or po el cut-off,	rocedures owder), e.g and batch	should be im level measu metering;	plemente irement,	ed to preven independent	t overfillin high-leve	ig of storage श alarms,
		•	installati appropri	on securi ate and s	ty systems	to prevent ui ude maintena	nauthoris nce arrai	ed access s ngements wh	hould be here nece	provided as ssary;
		•	there sh changes	ould be a to proce	in installati dures, abr	on log/diary to ormal events	o record and find	all incidents, ings of main	near-mis tenance i	ses, nspections;
		•	procedu incidents	res shoul s;	d be estab	lished to iden	itify, resp	ond to and l	earn from	such
		•	the roles be identi	and resp ified;	ponsibilitie	s of personne	l involve	d in accident	manager	ment should

Cont.

INTRODU	JCTION	TION TECHNIQUES EMISSIONS IMPACT										
Management	Materials inputs a	Activitie abateme	es/ Ground ent water	Waste	Energy Accidents	Noise	Monitoring Closure	Installation issues				
BAT for control of accidents (cont.)			 clear gui managed procedur commun attor or 	dance sho d, e.g. cor es should ication an	ould be available on how ntainment or dispersion, to be in place to avoid inci- nong operations staff duri	each ac o exting dents oc ing shift	ccident scenario sho uish fires or let them ccurring as a result o changes and mainte	uld be burn; f poor enance or				
			 safe shu 	tdown pro	work, ocedures should be in pla	ce;						
			commun emergen procedur redress t	ication ro cy service es should his;	utes should be establishe es both before and in the d include the assessment	ed with r event o of harm	elevant authorities a f an accident. Post- n caused and steps r	nd accident needed to				
			 appropria accident authoritie 	ate contro such as and eva	I techniques should be in oil spillage equipment, iso acuation procedures;	i place to olation c	o limit the conseque of drains, alerting of i	nces of an relevant				
			• personne	el training	requirements should be i	identifie	d and provided;					
			 the syster (Sections) 	ms for the 2.3.11 a	e prevention of fugitive er and 2.3.12) and in additior	missions n, for dra	s are generally releva ainage systems:	ant				
			- proce a bur treatr	edures sh nd sump, ment or di	ould be in place to ensure or sump connected to a c isposal;	e that th drainage	e composition of the system, are checke	contents of d before				
			- drain autor to en	age sump natic pum sure that	os should be equipped wi np to storage (not to disch sump levels are kept to a	th a higl narge); t n minimu	h-level alarm or sens here should be a sys um at all times;	or with stem in place				
			- high- level	level aları control;	ms etc. should not be rou	tinely us	sed as the primary m	nethod of				
		c2.	- the following risks as iden	plus any tified in 1	other specific techniques and 2 above	s identifi	ed as necessary to r	minimise the				
			 adequate testing to 	e redunda the same	ncy or standby plant sho e standards as the main p	uld be p plant;	provided with mainter	nance and				
			 process i container respect) contain s waters o achieved of accide prevent t also take Emergen reaching 	waters, si ated wate d (see Se and wher r sewer. S . There s ental emis heir entry account ncy storag controlled	te drainage waters, emer ers and spillages of chem ection 3.4.1 for information e necessary, routed to th d storm-water flows, and Sufficient storage should should also be spill contin sion of raw materials, pro r into water. Any emerger of the additional firewater le lagoons may be needed d waters (see Refs. 14 ar	gency fi inicals sh n regard e effluer treated be prov gency p oducts a ncy firev r flows c d to pre nd 15);	irewater, chemically would, where appropr ling legislative obligant system, with provi- before emission to co- rided to ensure that to procedures to minimi nd waste materials a water collection system or fire-fighting foams. went contaminated fi	iate, be itions in this sion to ontrolled his could be se the risk and to em should				
			 consider accidenta may be in probabilition 	ation shou al emissic nadvisabl ty of the e	uld be given to the possib ons from vents and safety e on safety grounds, atte emission;	vility of c relief vantion sh	containment or abate alves/bursting discs. lould be focused on t	ment for Where this reducing the				
	Secto	r spec	ific techni	iques								
		с3.	The following risk, and the comprehens guide to the above:	g techniqu breadth o ive, nor a outcome	ues are sector specific. C of waste types this guidar pplicable in all circumstar of the systematic assess	Dwing th nce may nces. T ment ou	ne site-specific nature v relate to, they are n hey are therefore pro itlined in the general	e of accident either ovided as a guidance				

Cont.

INTROD	UCTIO	N TEC	TECHNIQUES			EMISSIONS			IMPACT			
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		

Table 2-10 - Accident Risks

Aspect	Consequence of Release	Controls
Waste storage failure	• Litter,	Secure storage
Incoming waste or raw material handling / storage failure	 Contaminated land Spillage overfilling Putrefaction leading to odours / fire risk 	 Containment e.g. sealed floors Liquid reagents in bunds High level alarms Incoming waste mixing and rapid processing Fire detection and sprinkler operation
Waste charging failure	 Combustion conditions upset Releases to air (e.g. CO) 	 charging design / maintenance Waste inspection
Furnace control failure	Combustion conditions upsetRelease to air?Plant shut down?	 Waste feed quality control Maintenance of air systems Effective control parameter monitors
Residues handling / storage failure	 Contaminated land Damage to aquatic systems Potential releases to air 	Secure storageControlled or contained drainage
APC equipment failure e.g. power failure reagent shortage blockage damage to equipment	 Release of untreated combustion gases to air Plant shutdown? 	 Waste feed quality control Emergency power for fans / pumps Low level reagent alarms Pump maintenance Standby equipment provision (e.g. multiple smaller feed systems) Key parameter monitoring e.g. filter pressure drop

Abnormal Operating Conditions

BAT for control of accidents (cont.)

The incineration directives (HWID / WID) recognise that, at the emission limit values set, the nature of waste is such that its incineration may, in some circumstances lead to short term exceedences of the emission standards set. It therefore makes provision for member states to set out conditions in respect of periods of abnormal operation. Abnormal operation is any circumstance that leads to the exceedence of any emission limit value.

The Agency is currently developing standard permit conditions to ensure that these requirements are implemented and that they reflect BAT.

BAT for abnormal operation

The operator should implement a programme of preventative maintenance and control to ensure that abnormal operation is avoided.

In the event of abnormal operation the operator shall:

- 1 reduce or close down operations as soon as practicable until normal operations can be restored.
- 2. initiate the shutdown sequence on those <u>process lines</u> (i.e. emission averaging is not permitted for multiple line plant) where the continuous emission monitors serving them indicate that:
 - any individual continuously monitored emission limit value has been exceeded for 4 hours or more consecutively, or
 - a part of the flue gas abatement system has failed <u>and</u> any individual continuously monitored emission limit value has been exceeded for 4 hours or more consecutively, or
 - any emission limit has been exceeded for more than 60 hours cumulatively in any one year.
- 3. immediately initiate the shutdown sequence if any of the continuous emission monitors indicates that the half-hourly emission limit values for releases to air are exceeded by more than 100%.
- 4. initiate the shutdown sequence if any emissions measurement equipment is out of service for more than 24-hours or such shorter time if any other relevant monitored parameter (i.e. a parameter that might be considered to be a surrogate e.g. sulphur dioxide for HCI or CO for VOCs) is in breach of its specified emission limit.

INTRODUCTION TECHNIQUES					EMISSIONS			IMPACT			
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

2.9 Noise and Vibration

Within this section "*noise*" should be taken to refer to "*noise and/or vibration*" as appropriate, detectable beyond the site boundary.

The PPC Regulations require installations to be operated in such a way that "all the appropriate preventative measures are taken against pollution, in particular through the application of BAT". The definition of pollution includes "emissions which may be harmful to human health or the quality of the environment, cause offence to human senses or impair or interfere with amenities and other legitimate uses of the environment". BAT is therefore likely to be similar, in practice, to the requirements of the statutory nuisance legislation, which requires the use of "best practicable means", to prevent or minimise noise nuisance.

In the case of noise, "offence to any human senses" can normally be judged by the likelihood of complaints, but in some cases it may be possible to reduce noise emissions still further at reasonable costs, and this may exceptionally therefore be BAT for noise emissions.

For advice on how noise and/or vibration related limits and conditions will be determined see "IPPC Noise – Part 1 Regulation and Permitting", (see Ref. 19).

Principal sources of noise on incineration plant are:

- induced draft fans;
- harmonics between induced draft fans and the chimney;
- primary and secondary air fans;
- vehicle noise;
- WHB safety relief valves;
- transformers;
- cooling towers mainly noise from falling water but also fan noise;
- general mechanical handling such as dragging rather than lifting skips.



Describe the main sources of noise and vibration (including infrequent sources); the nearest noise-sensitive locations and relevant environmental surveys which have been undertaken; and the proposed techniques and measures for the control of noise.

With the Application the Operator should:

- 1. provide the following information for **each main source of noise and vibration** that fall within the IPPC installation:
 - the source and its location on a scaled plan of the site;
 - whether continuous/ intermittent, fixed or mobile;
 - the hours of operation;
 - its description, (e.g. clatter, whine, hiss, screech, hum, bangs, clicks, thumps or tonal elements);
 - its contribution to overall site noise emission (categorise each as high, medium or low unless supporting data is available).

A common sense approach needs to be adopted in determining which sources to include. The ones which need to be considered are those which may have environmental nuisance impact; e.g. a small unit could cause an occupational noise issue in an enclosed space but would be unlikely to cause an environmental issue. Conversely a large unit or a number of smaller units enclosed within a building could, for example, cause a nuisance if doors are left open. It must also be remembered that noise, which is not particularly noticeable during the day, may become more noticeable at night.

 provide the information required in (1) for each source plus its times of operation for *Infrequent* sources of noise and vibration, not listed above that fall within the IPPC installation: (such as infrequently operated/ seasonal operations, cleaning/maintenance activities, on-site deliveries/collections/transport or out-of-hours activities, emergency generators or pumps and alarm testing),

Cont.

Information needed to determine BAT for noise and vibration

INTRODUC	TIC	TION TECHNIQUES EMISSIONS IMPACT												
Management Mat	terials	Activities/	Ground	Waste	Enerav	Accidents	Noise	Monitoring	Closure	Installation				
in	puts	abatement	water					·······································	01000.0	issues				
Information needed to	3.	identify <i>the r</i> schools, hos there) and ar officers or as (a) the loca	pitals and pitals and ny other po part of a al environr	Dise-sens commerc oints/bour planning nent:	sitive sites sial premise ndary when consent, re	(typically dv es <u>may</u> be, d e conditions elating to:	vellings, lepending have be	parkland and g upon the ac en applied b	l open spa ctivities ur y Local Aı	aces – idertaken uthority				
determine BAT for noise and		• prov	/ide an ac	curate ma	ap or scale	d plan showi	ing grid r	eference, na	ture of the	ereceiving				
vibration (cont.)		site	, distance	and direc	tion from s	ite boundary	/;							
		(b) condition	sensitive	mposea v receptor)	vnich relate	e to other loc	ations (I.	e. boundary	tence or s	surrogate for				
		 any 	planning	condition	s imposed	by the Local	Authority	y;						
		 other 	er conditio	ns impos	ed by agre	ements, e.g.	limits on	operating tir	nes, tech	nologies etc;				
		• any	requireme	ents of an	y legal not	ices etc.								
		(c) the nois	se environ	ment:	: f 1	(- (
		 bac ano 	kground n	oise level	, if known	(day/night/e\ (pight) L	/ening) L : and/or	A,90,T;						
		 speriod amb 	vient noise	level (da level (da	v/evening/	night) LA eq,T	, anu/or as ann	ronriate [.]						
		 vibr 	ation data	which ma	ay be expr	essed in tern	ns of the	peak particle	velocity	(ppv) in mm				
		s⁻¹ (or the vibra	ation dose	e value (VI	0V) in m s ^{–1.}	.75		2	,				
		s ⁻¹ or the vibration dose value (VDV) in m s ^{-1.75} . For noise these are given the meaning as defined in BS4142:1997 "Method for rating industrial noise affecting mixed residential and industrial areas", and to which reference should be made for a full description. For vibration, the appropriate standard is BS6472:1992 "Evaluation of human exposure to vibration in buildings1 to 80 Hz". In very general terms "background" is taken to be the equivalent continuous A-weighted noise remaining when the source under investigation is not operational averaged over a representative time period, T. The "ambient" level is the equivalent continuous A-weighted combination of all noise sources far and distant, including the source under investigation and "specific noise" is the equivalent continuous A-weighted noise level produced by the source under investigation as measured at a selected assessment point. Both are averaged over a time period, T. BS4142 gives advice on the appropriate reference periods. "Worst case" situations and impulsive or tonal noise should be accounted for separately and not												
	4.	provide deta noise measu	ils of any rements u	environi Indertake	<i>mental no</i> n relevant	ise measure to the enviro	e ment su nmental	irveys, mode impact of the	elling or a site, ider	ny other ntifying:				
		• the purpo	se/contex	t of the s	urvey;									
		 the location 	ons where	measure	ements we	re taken;								
		 the source 	e(s) inves	tigated or	dentified;									
	_		ines.											
	5.	identify any s	specific loc	cal issues	and propo	sals for impi	rovemen	tS.						
	6.	describe the any others w	current or hich are p	propose ertinent t	d position v o the instal	with regard to lation	o the tech	niques belo	w, any in	Ref. 19 or				
	7.	demonstrate requirements Applicants) c	that the p s, by justify or alternati	roposals ying depa ve measu	are BAT, b rtures (as ires.	y confirming described in	complia Section	nce with the 1.2 and in the	indicative e Guide fo)r				
	In	dicative BA	AT Requ	iremen	ts									
	1.	The Operato adequate ma increases in specific nois	r should e aintenance noise (eg e attenuat	mploy ba e of any p maintena ion meas	sic good p arts of plar nce of bea ures assoc	ractice meas nt or equipme rings, air hau iated with pla	sures for ent whos ndling pla ant, equi	the control of e deterioratio ant, the build pment or ma	f noise, in on may giv ng fabric chinery).	cluding /e rise to as well as				
	2.	In addition th noise from th the Regulato installation e levels of 50d exceeded in	e Operato ne installat r and, in p xceed the B L _{Aeq} by certain cir	or should ion does particular numerica day or 45 cumstanc	employ sur not give ris should just al value of by night a ces are giv	ch other nois to reasona ify where eit the Backgrou re exceeded en in Ref. 19	e control able caus her Ratin und Sour . Reasor	l techniques se for annoya ig Levels (L _{Ag} id Level (L _{Ag} is why these	to ensure ince, in th _{eq,T}) from t _{b,T}), or the levels ma	that the e view of the absolute ay be				

INTRODUCTION TECHNIQUES				EMISSIONS			IMPACT			
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

- 3. In some circumstances "creeping background" (see Ref. 19) may be an issue. Where this has been identified in pre-application discussions or in previous discussions with the Local Authority, the Operator should employ such noise control techniques as are considered to be appropriate to minimise problems of to an acceptable level within the BAT criteria.
- 4. Noise surveys, measurement, investigation (which can involve detailed assessment of sound power levels for individual items of plant) or modelling may be necessary for either new or existing installations depending upon the potential for noise problems. Operators may have a noise management plan as part of their management system. More information on such techniques is given in Part 2 of Ref. 19.

INTROD	N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.10 Monitoring

Mandatory minimum monitoring requirements for waste incineration are outlined in European Legislation (Ref. 25). These requirements **must** be met wherever this legislation applies. Operators who burn wastes or waste derived fuels should consult the Regulator at an early stage in order to discuss whether the standards of these directives will apply to their situation.

In general monitoring standards in the following Directives will apply:

- WID standards apply to installations burning any waste or waste derived fuel (except those that are exempt from the Directive or are covered by HWID)
- HWID standards apply to the incineration and co-incineration of hazardous wastes (in non-HWID exempt installations) until 28 December 2005 when WID will apply.

The Regulator takes the view that installations not covered by WID must also meet the standards required by WID unless the Operator can clearly demonstrate that to do so would be excessively costly.

This section describes monitoring and reporting requirements for emissions to all environmental media. Guidance is provided for the selection of the appropriate monitoring methodologies, frequency of monitoring, compliance assessment criteria and environmental monitoring.

Application Form Question 2.10 Describe the proposed measures for monitoring emissions including any environmental monitoring, and the frequency, measurement methodology and evaluation procedure proposed.

With the Application the Operator should:

- describe the current or proposed position with regard to the monitoring requirements below or any others which are pertinent to the installation for "Emissions monitoring", "Environmental monitoring", "Process monitoring" (where environmentally relevant) and "Monitoring standards" employed;
- 2. provide, in particular, the information described in requirement 26 below;
- 3. provide justifications for not using any of the monitoring requirements described;
- 4. Identify shortfalls in the above information which the Operator believes require longer term studies to establish.
- 5. provide a direct comparison of their intended monitoring with the minimum requirements of the relevant European legislation this should aim to clearly demonstrate compliance with Directive requirements.

Emissions monitoring

The following monitoring parameters and frequency are normally appropriate in this sector. Generally, monitoring should be undertaken during commissioning, start-up, normal operation and shut-down unless the Regulator agrees that it would be inappropriate to do so.

Where effective surrogates are available they may be used to minimise monitoring costs unless specifically required by European legislation.

Where monitoring shows that substances are not emitted in significant quantities, consideration can be given to a reduced monitoring frequency unless specifically required by European legislation.

Monitoring and reporting of emissions to water and sewer

 European directives (WID & HWID) contain specific requirements in respect of the minimum standards for monitoring of releases to water. They also contain the criteria that will be used to demonstrate that the emission limit values so assessed are being complied with.

Table 2-11 below outlines the requirements of the WID in respect of the monitoring of releases from installations generally and those arising from air pollution control devices in particular. Operators of installations falling with in the scope of WID must comply with these minimum standards. Operators of non-WID installations must justify departures from these standards:

Cont.

water

Emissions

monitoring

INTRODUCTION TECHNIQUES				EMISSIONS			II	IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Table 2-11 Emissions monitoring to water

Parameter	Monitoring Type / Frequency	Method / Specification Ref(s)
рН	Continuous	CEN if available, ISO or national or international if not.
Temperature	Continuous	CEN if available, ISO or national or international if not.
Flow	Continuous	CEN if available, ISO or national or international if not.
Total suspended solids (as defined by 91/271/EEC)	Daily spot sample or Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
Hg and its compounds, expressed as Hg.	Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
Cd and its compounds, expressed as Cd.	Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
TI and its compounds, expressed as TI.	Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
As and its compounds, expressed as As.	Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
Pb and its compounds, expressed as Pb.	Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
Cr and its compounds, expressed as Cr.	Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
Cu and its compounds, expressed as Cu.	Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
Ni and its compounds, expressed as Ni.	Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
Zn and its compounds, expressed as Zn.	Monthly flow proportional sample over 24 hours	CEN if available, ISO or national or international if not.
Dioxins and Furans (TEQ)	Every 6 months, but every 3 months during the first year of operation.	CEN if available, ISO or national or international if not.

Notes:

- 1. Monitoring should be carried out at the point where waste water is discharged off site and may therefore be carried out down stream of an effluent treatment plant.
- 2. For discharges of metals, SS and dioxins, WID requirements should be adhered to in respect of using mass balance calculation to demonstrate that releases to water attributable to air pollution control devices comply with the ELVs in the Directive.
- 3. pH, temperature and flow monitoring requirements apply to all off site discharges.
- 4. WID also contains requirements in respect of combined discharges and off site treatment plant that must be adhered to.
- 5. Up to date guidance on the actual technique that should be used for monitoring should be obtained from the Regulator.

Cont.

INTRODU	JCTIC	ON TE	CHNIC	UES	E	MISSIO	NS	IMPACT					
Management I	Materials inputs	Activities/ abatemen	dia Ground t water	Waste	Energy	Accidents	Noise	Monitoring Closure Installation issues					
	2.	In addition to controlle	to consider d waters ar	ation of th	ne Directi [,] s should ii	ve requireme nclude at lea	ents, mor st:	nitoring of process effluents released					
Table 2-12		Paramet	er	Ν	Ionitorin	g frequency	1						
effluent		Flow rate		(Continuous and integrated daily flow rate								
monitoring to water		рН		C	Continuous								
		Tempera	ture	C	Continuous								
		COD/BOI	D	F r	low weigl eported a	nted sample s flow weight	or compo ted mont	osite samples, weekly analysis, hly averages					
		TOC		C	Continuou	s							
		Turbidity		C	Continuou	S							
		Dissolved	l oxygen	C	Continuou	S							
		NB - other appropriate water and	NB - other parameters specifically limited in the Permit should be monitored. The appropriateness of the above frequencies will vary depending upon the sensitivity of the recei water and should be proportionate to the scale of the operations.										
		BOD/ADt a	and COD/Al	Ot should	be estab	lished annua	lly as an	annual average.					
Emissions monitoring to water (cont.)	3.	In addition substances the release unless it is at least an	In addition, the Operator should have a fuller analysis carried out covering a broad spectrum of substances to establish that all relevant substances have been taken into account when setting the release limits. This should cover the substances listed in Schedule 5 of the Regulations unless it is agreed with the Regulator that they are not applicable. This should normally be done at least annually.										
	4.	Any substa environme monitored metals. Us concentrat	ances found nt may be s more regula sing composion does no	to be of usceptibl arly. This site samp t vary exc	concern, o e and upo would pa les is the cessively.	or any other on which the irticularly app technique m	individua operation oly to the lost likely	Il substances to which the local ns may impact, should also be common pesticides and heavy to be appropriate where the					
Emissions	Мог	nitoring ar	nd reportii	ng of en	nissions	to air							
monitoring to a	ir _{5.}	Continuou: significant	s monitoring or where it	y shall be is needeo	used whe	ere it is requi ain process o	red by leg control;	gislation, where the releases are					
	6.	Gas flow s	hould be me	easured t	o relate c	oncentration	s to mas	s releases;					
	7.	To relate n recorded:	neasuremer	nts to refe	erence coi	nditions, the	following	will need to be determined and					
		 temper 	ature and p	ressure;									
		• oxygen	, where the	emissior	is are the	result of a co	ombustio	n process;					
		 water v other w exceed 	apour conte et gas strea 3% v/v or v	ent, where am. It wo where dry	e the emis uld not be gas mea	e required wh suring techn	e result onere the virtual of the second sec	of a combustion process or any water vapour content is unable to e used for all pollutants.					
	8.	Where app verify that mist, fume complaints	oropriate, pe all final rele or droplets and other i	eriodic vis ases to a Video re nvestigat	dic visual and olfactory assessment of releases should be undertaken s to air should be essentially colourless, free from persistent trailing ideo records of stack emissions may help with the resolution of estigations and should be adopted where concerns arise.								
	9.	The monito and should within the s Operator c	oring require l be referred scope of WI an clearly d	ements ou d to in res D will be emonstra	utlined in pect of th expected ate that to	the WID are eir detailed r to meet the do so would	consider equireme standard not repre	ed to represent BAT for this sector ents. Installations that do not fall Is outlined in WID except where the esent BAT.					
								Cont.					

INTRODUCTION			CHNIQ	UES	EMISSIONS			II	IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

	Table 2-13 below provides a	summary of the monitoring requiremen	ts for releases to air:
Table 2-13 Emissions	Parameter	Frequency	Method / Specification Ref(s)
monitoring to air	NOx (NO and NO ₂ expressed as NO ₂)	Continuous	MCERTS Performance Standards for CEMs
	CO	Continuous	MCERTS Performance
	Total dust	Continuous	MCERTS Performance
	VOC (expressed as	Continuous	MCERTS Performance
	HCI	Continuous or	MCERTS Performance
		Periodic (only where raw flue gas cannot exceed ELV) at same	Standards for CEMs and for periodic HCI see NCAS
		frequency as metals	Technical Guidance (on BS EN1911)
	HF	Continuous or Periodic (only where raw flue gas	CERAM special publication 35, Recommendations of
		cannot exceed ELV, or where HCI ELVs are complied with) at same frequency as metals	ISO/TC/146/SC1/WG14
	SO ₂	Continuous or Periodic (only where raw flue gas cannot exceed ELV) at same frequency as metals	MCERTS Performance Standards for CEMs
	Cd + Tl	Periodic – 2 per year but every 3 months in first year of operation. Average value over sample period of between 30 minutes and 8 hours.	USEPA method 29 (CEN standard under development)
	Hg	Periodic – 2 per year but every 3 months in first year of operation. Average value over sample period of between 30 minutes and 8 hours.	USEPA method 29 (CEN standard under development)
	Sb + As + Pb + Cr + Co + Cu + Mn + Ni + V	Periodic – 2 per year but every 3 months in first year of operation. Average value over sample period of between 30 minutes and 8 hours.	USEPA method 29 (CEN standard under development)
	Dioxins and Furans (TEQ as per annexe II of the directive)	Periodic – 2 per year but one every 3 months in first 12 months of operation. Average value over sample period of between 6 and 8 hours. Determination in accordance with CEN standard.	CEN standard equates to BS EN1948 (parts 1-3)
	Exhaust gas Oxygen concentration	Continuous	MCERTS Performance Standards for CEMs
	Exhaust gas Pressure	Continuous	MCERTS Performance Standards for CEMs
	Exhaust gas Temperature	Continuous	MCERTS Performance Standards for CEMs
	Exhaust gas water content.	Continuous (not required if sampled exhaust gas is dried prior to analysis)	MCERTS Performance Standards for CEMs

Emissions monitoring to air (cont.)

- 10. Where not indicated in the table above, averaging periods and compliance criteria are those stated in the directive. Averaging periods are also included in Section 3.3 Releases to air.
- 11. Although WID will eventually apply to installations burning hazardous waste, until this is the case, HWID applies. Guidance on the monitoring requirements of HWID is contained in Agency regulatory guidance of 14 July 1999.
- 12. Specific additional determinants may be required in relation to certain waste types as detailed below. In general this is because of the possible presence of particular substances in such waste derived fuels, and the need to demonstrate their fate in the process. The need for, scope and frequency of such additional monitoring should be reviewed in the light of the data obtained.

Cont.

INTROD	N TEO	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Table 2-14 -Additional monitoring requirements for selected wastes

Waste	Additional extractive monitoring
Substitute liquid fuel	Pentachlorophenol
Tyres (whole or shredded)	Zinc PAHs, benzene, butadiene, styrene, HBr and chloromethane. These determinands are currently under review.
Refuse derived fuel	Zinc
Waste wood chips and sawdust	Pentachlorophenol Hexchlorocyclohexane (all isomers) Tributyl tin compounds
Sewage sludge	Zinc
Waste photographic emulsions e.g. X-ray film	Silver
Commercial waste (paper, cardboard, rags etc.)	Zinc
Any material containing organophosphates	Total phosphorous compounds

- 13. The readouts from continuous emission monitors should be processed using software that reports monitoring compliance information to enable direct comparison with the emission limit values specified relevant European legislation and in this guidance (see Section 3 Emissions) i.e. the software should be capable of calculating half hourly, daily or other averages as appropriate.
- 14. At sites where there is significant public concern, the Regulator should be provided with the means for 24 hour remote access to real time and historical emissions readouts and reports. The information viewable by the regulator should include:
 - Real time CEM readout graphs
 - The most recently calculated average concentrations
 - Access to historical emissions and operational records
 - The cctv image of the stack emissions (where available)
 - Other key operational parameters e.g. furnace temperatures

It should be noted that providing this facility will not in any way absolve the responsibility of the Operator to undertake their own monitoring and reporting of emissions.

Monitoring and reporting of waste emissions

- 15. Gate and fly ash are the main wastes produced at the installation. Separate guidance on the methods for securing representative ash samples is being produced by the Agency (Ref. 30). Samples should be taken in accordance with this protocol.
- 16. The scope of the analyses should:
 - Follow the specification in the protocol where indicated for particular sectors (e.g. animal remains incineration)
 - Provide sufficient information to fulfil the waste management duty of care (this should be agreed according to local circumstances but must address the legal principles involved)
 - Be selected according to the risks associated with the fate of that ash i.e. disposal of a
 consistent low leaching, stable ash at a highly engineered landfill is likely to require less
 rigorous testing than where it is recycled as an aggregate near a sensitive water course.
 - Take account of changes in waste input or operating regime that may influence ash composition.
 - Take account of any subsequent ash treatment / stabilisation stages i.e. the tests carried out
 in such circumstances may be less rigorous owing to the controls in place at the ash treatment
 site where a testing regime will itself fulfil the need to address subsequent disposal / recycling
 risks.

Cont.

Emissions

INTRODUC	CTIC	ON TEC	HNIQ	UES	E	MISSIC	ONS	IMPAC ⁻	Г
Management Ma	terials	Activities/	Ground	Waste	Energy	Accidents	Noise	Monitoring Closure	Installation
inanagement	nputs	abatement	water	Waste	Lifergy	Accidents	NOISE	Monitoring	issues
		Sampling free	uencies f	or ash sh	nould be	selected as	follows:		
		 Bottom as as selected 	sh and fly	ash - at l	east one n 14 abo	sample tes	ted every	three months for the wide	er analysis
		 Bottom as 	sh only:	ing to iter	11 14 000	vc.			
		- one sa	ample test	ted for To	otal Organ	nic Carbon	(TOC) at	least every week during t	he initial
		period	of testing	(i.e. fror	n day one	e of operation	on of new	installations or from the	date of
		implen	nentation	of this re	gime at e	existing insta	allations	amont with the Degulator	if there is
		- the he	tent com	bliance w	ith the re	quired stan	dard (WII	D standard is maximum 3 ^o	% TOC w/w
		dry sa	mple). Tł	ne minim	um frequ	ency will be	one eve	ry three months.	
	17.	Operators sho	ould repo	rt the res	ults of the	e tests carri	ed out to	the Regulator at least eve	erv 6
		months. Indiv permit require	vidual nor ements.	n-complia	ince tests	should be	reported	without delay and in acco	rdance with
	18.	Operators mu	ist state ir	n their ap	plication	the propose	ed treatm	ent / recycling / disposal f	ate of the
		fly ash and gr	ate ash p	roduced.	Operato	ors should a	udit the o	disposal (or recycling) cha	in in order
		information th	at they ha	ave provi	ded in the	eir application	on remail	ns current.	ne
	10	Operators ab		et to tho E	Dogulator	the questit	ioo ond n	roportions of fly ash and k	nottom och
	19.	that have bee	in sent for	disposa	l or recyc	ling at least	t every si	x months.	000000000000000000000000000000000000000
	20.	At least every	year Ope	erators sl	hould inve	estigate and	d report to	the Regulator the feasib	ility of
		the economic ash.	s of deve	loping (o	n or off si	t is carried te) ash trea	out. This itment fac	cilities to promote recycling	g of the
	21.	For all waste	emissions	s the follo	wing sho	ould be mon	itored an	d recorded:	
		 the physic 	al and ch	emical co	ompositio	on of the wa	ste;		
		 its hazard 	characte	ristics;	-				
		 handling p 	precaution	ns and su	Ibstances	with which	it canno	t be mixed;	
		 where was a program potential o the food c 	ste is disp ime of mo contamina hain.	oosed of onitoring onts and p	directly to should be potential) land, for e e establishe pathways fr	xample s d that tak om the la	ludge spreading or an on- kes into account the mater and to groundwater surfac	∙site landfill, rials, e water or
	En	vironmenta	l monite	oring (l	beyond	the insta	allation)	
Environmental monitoring	22.	The Operator emissions to	should controlled	onsider tl water, g	he need f roundwat	or environn ter, air or la	nental mo nd or em	nitoring to assess the effe issions of noise or odour.	ects of
		Environmenta	al monitor	ing may l	be require	ed, e.g. whe	en:		
		• there are	vulnerable	e recepto	ors;				
		 the emissimation may be at 	ions are a risk;	ı significa	int contrib	outor to an I	Environm	ental Quality Standard (E	QS) which
		the Opera environme	tor is looł ent;	king for d	epartures	from stand	lards bas	ed on lack of effect on the	÷
		• there is a	need to v	alidate m	odelling	work.			
		The need sh	ould be d	onsider	ed for:				
		 groundwa and take i place both 	ter, where nto accou n up-gradi	e the mor int short a ent and o	nitoring sl and long- down-gra	hould be de term variati dient of the	signed to ons in bo site:	o characterise both quality oth. Monitoring will need to	and flow o take
		 surface was upstream 	ater, when	re consid	eration w	ill be neede	ed for sar	npling, analysis and repor	ting for
		• air, includi	ing odour	,	-				
		land conta	amination	, includin	g vegetat	ion, and ag	ricultural	products;	
		 assessme 	nt of heal	Ith impac	ts;				
		 noise. 							

Incineration

INTRODUC	TION TECHNIQUES EMISSIONS IMPACT										
Management Ma	erials Activities/ Ground Waste Energy Accidents Noise Monitoring Closure Install	lation									
in	buts abatement water water house include the house monitoring closure issu	Jes									
	Where environmental monitoring is needed the following should be considered in dra	wina									
	up proposals:	wing									
	 determinands to be monitored, standard reference methods, sampling protocols; 										
	• monitoring strategy, selection of monitoring points, optimisation of monitoring approach;										
	 determination of background levels contributed by other sources; upportainty for the amplexed methodologies and the resultant everall upportainty of 										
	 uncertainty for the employed methodologies and the resultant overall uncertainty of measurement; 										
	 quality assurance (QA) and quality control (QC) protocols, equipment calibration and maintenance, sample storage and chain of custody/audit trail; 										
	 reporting procedures, data storage, interpretation and review of results, reporting format the provision of information for the Regulator. 	for									
	Guidance on air quality monitoring strategies and methodologies can be found in Technical Guidance Notes M8 and M9 (see Ref. 20), for noise (see Ref. 19) and for odour (see Ref. 22)	2).									
	Specific environmental monitoring requirements which may be appropriate for this sector:										
	To water:										
	Effluent treatment plant discharges to controlled waters;										
	Cooling water discharges										
	To air:										
	 daily visual monitoring to air for smoke, dust, litter, vermin, plumes and daily olfactory odd monitoring, with more extensive monitoring if nuisance is occurring or appears likely (see 22). 	our e Ref.									
	 24 hour time / date coded CCTV / video recordings of the chimney stack emissions shou kept for those sites where there is significant public concern / record of complaint. The camera should be positioned such that the recording provides a good representation of the view likely to be observed by complainants. 	ld be he									
	All plants should record wind speed and direction data to assist with complaint investigat	ion									
	To land:										
	Monitoring surveys will need to be established where sensitive soil systems or terrestrial ecosystems are at risk from indirect emission via the air (i.e. deposition related) or direct imit	nacts									
	of any on site waste storage and treatment operations.	puoto									
	To groundwater:										
	Groundwater sampling may be needed where:										
	 there is uncertainty about drainage systems, especially on older sites; there are deliberate discharges to groundwater; 										
	 there are any other deposits to land. 										
	Noise:										
	See Section 2.9, and Ref. 19.										
	Monitoring of process variables										
Monitoring process	23. The following process variables have particular potential to influence emissions. Their monit and control is necessary to demonstrate BAT:	toring									
variables	 Waste feedstock composition should be analysed at a frequency and in a manner appropriate to the plant type concerned. The type and frequency of the analyses carried out should be selected with reference to the plant performance and the value that may be accrued from additional knowledge of the feedstock stream. Plants with a demonstrably wide operation envelope and good emission compliance record are less likely to require stringent measure adopted in this respect. 	oriate be 1 nal ures									
	 Waste throughput should be recorded in such a way that it enable comparison with the d throughput. In addition, an hourly and annual throughput shall be recorded as a minimur 	esign m.									

Combustion temperature at the agreed 2 second residence time location;

Cont.

		FINIQUE		5510145		IIVI	PACI					
Management in	outs abatement	water Was	ste Energy Acc	idents Noise	Monito	o <mark>ring</mark> C	losure issues					
	 differential pressure across bag filters can indicate filtration efficiency and bag blowouts. potential difference across the EP plates; reagent feed rates; upstream HCl concentration (to enable linkage to and automatic control of scrubbing medium dose rate); 24. Temperature monitoring is required for process monitoring and control and in order to demonstrate compliance with the required combustion temperatures (see Section 2.3.4 - Furnace requirements). The performance of temperature monitoring equipment varies depending upon its location in the process and the environment to which it is subjected. Table 2-15 below outlines the main factors relevant to the selection of temperature monitoring equipment.											
Table 2-15 - Gas temperature	Mathada	Magging	Response	Equipment	Suita Meas	ble for uring:	Commente					
measurement systems	weinoas	weasures	Time	⊂quipment	Point	Bulk Mean	Comments					
	Acoustic Pyrometry	Average over path	Instantaneous, continuous	Non-intrusive, permanent	Yes ¹	Yes	Good accuracy					
	Optical and Radiation Pyrometry	Average over path	Instantaneous, continuous	Viewing point	Yes ²	Yes	"cold wall" and particle effects					
	Shielded Thermocouples	Gas temperature at point	Delayed response but can operate continuously	Permanent	Yes	Yes ³	Negative bias. Can droop if orientated horizontally					
	Suction Pyrometers	Gas temperature at point	Instantaneous, periodically	Special probe for periodic working only	Yes	Yes ³	Very accurate but needs manual attendance. Can be cumbersome					
	Heat Balance	Estimate of bulk mean gas temperature	Requires computations	Off-line, based on knowledge of operating parameters	No	Yes	Estimate only, but acceptable for bulk mean temperature estimation.					
	Notes:											
	 Can only deliver point data if several arrays are installed. Only some systems can be focused. Can deliver profile information only if several probes are used or traversing is undertaken. It can be seen that suction pyrometers offer the best performance and may be BAT in many situations. This will be particularly the case in respect of temperature validation work (see Section 2.3.4.4) where they should form at least the basis for the calibration and correction of other monitors. Monitoring standards (standard reference methods) 											
Equipment standards MCERTS	quality of monitorin monitoring are fit f monitoring system	Agency has intring data and to e or purpose. Pe s (CEMs) and c	ensure that the in rformance standa continuous ambie	strumentation and ards have been ent air quality mo	nd metho published pnitoring s	dologies dologies for cont systems (employed for inuous emissions (CAMs).					

Other MCERTS standards are under development to cover:

manual stack emissions monitoring;

Incineration

INTROD	UCTION	TECHN	IIQUES	E	MISSIO	NS	I	MPAC	Т			
Management	Materials Ac	tivities/ Gro	und Wast	e Energy	Accidents	Noise	Monitoring	Closure	Installation			
		and the second s							100000			
	 portat ambie water data a Opera equip 25. As f requusin acco the MCI 26. The com i 	 ambient air quality monitors; water monitoring instrumentation; data acquisition; Operators' own arrangements, such as for installation, calibration and maintenance of monitoring equipment, position of sampling ports and provision of safe access for manual stack monitoring. 25. As far as possible, Operators should ensure their monitoring arrangements comply with the requirements of MCERTS where available, e.g. using certified instruments and equipment, and using a registered stack testing organisation etc. Where the monitoring arrangements are not in accordance with MCERTS requirements, the Operator should provide justification and describe the monitoring provisions in detail. See Environment Agency Website (Ref. 20) for listing of MCERTS equipment 26. The following should be described in the application indicating which monitoring provisions comply with MCERTS requirements or for which other arrangements have been made: monitoring methods and procedures (selection of Standard Reference Methods); justification for continuous monitoring or spot sampling; reference conditions and averaging periods; measurement uncertainty of the proposed methods and the resultant overall uncertainty; the proposed criteria for the assessment of non-compliance with Permit limits and details of monitoring strategy aimed at demonstration of compliance (this should comply with any relevant legislative requirements); reporting procedures and data storage of monitoring results, record keeping and reporting intervals for the provision of information to the Regulator; procedures for monitoring during start-up and shut-down and abnormal process conditions; 										
	• (drift correction	calibration	intervais ar amplers ar	nd methods (nd laboratorie	Note: sy	stems without ails of the peo	ple used a	vallable); nd the			
		training/compe	etencies.									
Standards for sampling and	Sampling	g and analysi	s standard	5								
analysis	27. The nee	analytical me ding to be mo	thods given hitored, stan	in Append dards show	x 1 should b uld be used in	e used. h the foll	In the event o owing order of	f other sub f priority:	ostances			
	•	Comité Europe British Standa	en de Norn ds Institutio	nalisation (n (BSI):	CEN);							
	•	International S	tandardisati	on Organis	ation (ISO);							
	•	United States	Environmen	tal Protecti	on Agency (I	JS EPA);					
	• /	American Soc	ety for Test	ng and Ma	terials (ASTI	M);						
	•	Deutches Insti	tute für Nori	mung (DIN);							
	• `	Verein Deutch	er Ingenieu	e (VDI);								
	 Association Française de Normalisation (AFNOR). Further guidance on standards for monitoring gaseous releases relevant to IPC/IPPC is given in the Technical Guidance Note 4 (Monitoring) (see Ref. 20). A series of updated Guidance Notes covering this subject is currently in preparation. This guidance specifies manual methods of sampling and analysis which will also be suitable for calibration of continuous emission monitoring instruments. Further guidance relevant to water and waste is available from the publications of the Standing Committee of Analysts. 											
	If in doub	t the Operator	should con	sult the Re	gulator.							

INTROD	N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.11 De-commissioning

The IPPC application requires the preparation of a site report whose purpose, as described in more detail in Refs. 3 and 4 is to provide a point of reference against which later determinations can be made of whether there has been any deterioration of the site and information on the vulnerability of the site.



With the Application the Operator should:

- 1. supply the site report;
- describe the current or proposed position with regard to the techniques below or any others which are pertinent to the installation;
- 3. for existing activities, identify shortfalls in the above information which the Operator believes require longer term studies to establish.

Indicative BAT Requirements

Operations during the IPPC Permit

BAT for decommissioning

1.

Operations during the life of the IPPC Permit should not lead to any deterioration of the site if the requirements of the other sections of this and the specific sector notes are adhered to. Should any instances arise which have, or might have, impacted on the state of the site the Operator should record them along with any further investigation or ameliorating work carried out. This will ensure that there is a coherent record of the state of the site throughout the period of the IPPC Permit. This is as important for the protection of the Operator as it is for the protection of the environment. Any changes to this record should be submitted to the Regulator.

2. Steps to be taken at the design and build stage of the activities

Care should be taken at the design stage to minimise risks during decommissioning. For existing installations, where potential problems are identified, a programme of improvements should be put in place to a timescale agreed with the Regulator. Designs should ensure that:

- underground tanks and pipework are avoided where possible (unless protected by secondary containment or a suitable monitoring programme);
- there is provision for the draining and clean-out of vessels and pipework prior to dismantling;
- · lagoons and landfills are designed with a view to their eventual clean-up or surrender;
- · insulation is provided which is readily dismantled without dust or hazard;
- materials used are recyclable (having regard for operational or other environmental objectives).

3. The site closure plan

A site closure plan should be maintained to demonstrate that, in its current state, the installation can be decommissioned to avoid any pollution risk and return the site of operation to a satisfactory state. The plan should be kept updated as material changes occur. Common sense should be used in the level of detail, since the circumstances at closure will affect the final plans. However, even at an early stage, the closure plan should include:

- either the removal or the flushing out of pipelines and vessels where appropriate and their complete emptying of any potentially harmful contents;
- plans of all underground pipes and vessels;
- the method and resource necessary for the clearing of lagoons;
- the method of ensuring that any on-site landfills can meet the equivalent of surrender conditions;
- the removal of asbestos or other potentially harmful materials unless agreed that it is reasonable to leave such liabilities to future owners;

INTRODUC	TION TE		UES	E	MISSIO	NS		IMPAC	Т
Management In	terials Main puts Activities	Abatement & control	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation Issues
 methods of dismantling buildings and other structures, see Ref. 24 which gives guidance on the protection of surface and groundwater at construction and demolition-sites; testing of the soil to ascertain the degree of any pollution caused by the activities and the need for any remediation to return the site to a satisfactory state as defined by the initial site report. 									
	co-ordinated with responsibilities under the Radioactive Substances Act 1993.)								
	For existing acti improvement co	vities, the sit Indition.	e closure	e plan ma	y, if agreed v	with the F	Regulator, be	submitted	as an

INTROD	UCTIO	N TEC	CHNIQ	UES	E	MISSIC	ONS		IMP	\CT
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.12 Installation Wide Issues

In some cases it is possible that actions which benefit the environmental performance of the overall installation will increase the emissions from one Permit holder's activities. For example, taking treated effluent as a raw water supply will probably slightly increase emissions from that activity but could dramatically cut the total emissions from the whole installation.



Where you are not the only Operator of the installation, describe the proposed techniques and measures (including those to be taken jointly by yourself and other Operators) for ensuring the satisfactory operation of the whole installation.

With the Application the Operator should:

- 1. where there are a number of separate Permits for the installation (particularly where there are different Operators), **identify** any installation wide issues and opportunities for further interactions between the Permit holders whereby the performance of the overall installation may be improved; and in particular
- 2. describe the current or proposed position with regard to the techniques below, or any others which are pertinent to the installation;

Indicative BAT Requirements

The possibilities will be both sector and site-specific, and include:

- 1. communication procedures between the various Permit holders; in particular those needed to ensure that the risk of environmental incidents is minimised;
- 2. benefiting from the economies of scale to justify the installation of a CHP plant;
- 3. the combining of combustible wastes to justify a combined waste-to-energy/CHP plant;
- 4. the waste from one activity being a possible feedstock for another;
- 5. the treated effluent from one activity being of adequate quality to be the raw water feed for another activity;
- 6. the combining of effluent to justify a combined or upgraded effluent treatment plant;
- 7. the avoidance of accidents from one activity which may have a detrimental knock-on effect on the neighbouring activity;
- 8. land contamination from one activity affecting another or the possibility that one Operator owns the land on which the other is situated.

BAT across the whole installation

3 EMISSION BENCHMARKS

3.1 Emissions Inventory and Benchmark Comparison

Application Form Question 3.1 Describe the nature, quantities and sources of foreseeable emissions into each medium (which will result from the techniques proposed in Section 2).

With the Application the Operator should:

- provide a table of significant emissions of substances (except noise, vibration, odour or heat which are covered in their respective sections) that will result from the proposals in Section 2 and should include, preferably in order of significance:
 - substance (where the substance is a mixture, e.g. VOCs or COD, separate identification of the main constituents or inclusion of an improvement proposal to identify them may be required);
 - source, including height, location and efflux velocity;
 - media to which it is released;
 - any relevant EQS or other obligations;
 - benchmark;
 - proposed emissions normal/max expressed, as appropriate (see Section 3.2), for:
 - mass/unit time;
 - concentration;
 - annual mass emissions.
 - statistical basis (average, percentile etc.);
 - notes covering the confidence in the ability to meet the benchmark values;
 - if intermittent, the appropriate frequencies;
 - plant loads at which the data is applicable;
 - whether measured or calculated (the method of calculation should be provided).

The response should clearly state whether the emissions are current emission rates or those planned following improvements, and should cover emissions under both normal and abnormal conditions for:

- · point source emissions to surface water, groundwater and sewer;
- waste emissions (refer to Sections 2.5 and 2.6 Waste Management);
- point source emissions to air;
- significant fugitive emissions to all media, identifying the proportion of each substance released which is due to fugitives rather than point source releases;
- abnormal emissions from emergency relief vents, flares etc.;
- indirect and direct emission of carbon dioxide associated with energy consumed or generated.

Emissions of carbon dioxide associated with energy use should be broken down by energy type and, in the case of electricity, by source e.g. public supply, direct supply or on site generation. Where energy is generated on site, or from a direct (non-public) supplier, the Operator should specify and use the appropriate factor. Standard factors for carbon dioxide emissions are provided in the Energy Efficiency Guidance Note.

Where VOCs are released, the main chemical constituents of the emissions should be identified. The assessment of the impact of these chemicals in the environment will be carried out as in response to Section 4.1.

For waste, emissions relate to any wastes removed from the installation, or disposed of at the installation under the conditions of the Permit, e.g. landfill. Each waste should have its composition determined and the amounts expressed in terms of cubic metres or tonnes per month.

A suitable table on which to record this information is provided in the electronic version of this Guidance Note.

- 2. compare the emissions with the benchmark values given in the remainder of this Section;
- 3. where the benchmarks are not met, revisit the responses made in Section 2 as appropriate (see Section 1.2) and make proposals for improvements or justify not doing so.

3.2 The Emission Benchmarks

Introduction to emission benchmarks Guidance is given in section 3.3 on release concentrations achieved for key substances for each subsector and a benchmark emission limit value. The mandatory release limits specified in European legislation for various incineration plants are included and must be met as a minimum where they apply.

The "reported ranges" provide a measure of current performance against the required standards. At new installations it may be possible for releases at the lower end of the reported range to be achieved. Existing installations should operate to the lowest practicable figure taking into account the BAT criteria. Where they apply, installations that cannot meet the minimum standards met by European legislation will not be permitted to operate.

3.2.1 Emissions to air associated with the use of BAT

Emissions quoted as averages are based upon continuous monitoring or are averaged over the period of the sampling as specified, during the period of operation. Care should always be taken to convert benchmark and proposed releases to the same reference conditions for comparison. To convert measured values to reference conditions, see Technical Guidance Note M2 (Ref. 20) for more information

Limits in Permits should always reflect the averaging periods stated. Additional limits may be set for mean or median values over long or short periods. The periods and limits selected should reflect:

- · the manner in which the emission may impact upon the environment;
- likely variations which will arise during operation within BAT;
- possible failure modes and their consequences;
- the capabilities of the monitoring and testing system employed.

For any additional, non-legislative limits that are not specified in this guidance, where emissions are expressed in terms of concentrations and where continuous monitors are employed, it is recommended that limits are defined such that:

- not more than one calendar monthly average during any rolling twelvemonth period shall exceed the benchmark value by more than 10%;
- not more than one half hour period* during any rolling 24 hour period shall exceed the benchmark value by more than 50%.
- * for the purpose of this limit half hourly periods commence on the hour and the half hour.

For any additional non-legislative limits that are not specified in this guidance, where spot tests are employed:

- the half hour limit above shall be applied over the period of the test;
- the mean of three consecutive tests taken during a calendar year shall not exceed the benchmark value by more than 10%.

3.2.2 Emissions to water associated with the use of BAT

Wastewater treatment systems can maximise the removal of metals using precipitation, sedimentation and filtration. The reagents used for precipitation will be defined by the mix of metals present, and may include hydroxide, sulphide or a combination of both. Concentrated effluents should be pre treated before discharge into the final effluent treatment system, and techniques such as electrolysis, reverse osmosis and metal removal using ion exchange systems may need to be employed. Water discharges should be kept to a minimum by using closed cycle cooling systems and by maximising the reuse of treated process water.

Where automatic sampling systems are employed, limits must comply with European legislation and if this is not jeopardised, may be defined such that:

• not more than 5% of samples shall exceed the benchmark value.

Where spot samples are taken:

• no spot sample shall exceed the benchmark value by more than 50%.

3.2.3 Introduction to Standards and obligations

In addition to meeting the requirements of BAT, and any mandatory release concentration (and process operation) specifications in European legislation, there are other national and international standards and obligations which must either be safeguarded through the IPPC Permit or, at least, taken into account in setting Permit conditions. This is particularly the case for any EC based EQSs. The most likely of these to be relevant in this sector are referred to below. The extracts from standards are, however, quoted for ease of reference; the relevant and most up to date standards should be consulted for the definitive requirements.

EC based EQ Standards

IPPC: A Practical Guide (see Ref. 3) explains how these should be taken into account and contains an annex listing the relevant standards. (See Appendix 2 for equivalent legislation in Scotland and Northern Ireland). They can be summarised as:

Mandatory incineration and co-incineration Standards

The standards are currently in a transitional stage. Directive 2000/76/EC is likely to be the most relevant the installations covered by this note. Some of the older Directives contain standards that are likely to be improved upon when using BAT or BATNEEC.

• Directive 2000/76/EC on the incineration of waste.

This Directive will replace the other incineration directives and set requirements for the majority of incineration plant burning wastes. Reference should be made to the Directive regarding the scope to various plants and the dates when the obligations it contains must be complied with. The standards in this guidance incorporate the requirements of this Directive where applicable.

Directive 94/67/EC on the incineration of hazardous waste

This Directive will apply to the incineration and co-incineration of specified hazardous wastes until replaced by 2000/76/EC. The standards in this guidance incorporate the requirements of this Directive where applicable.

Directive 89/429/EEC on the prevention of air pollution from existing municipal waste incineration plants

This Directive sets minimum mandatory standards for "existing" (see directive for definition) municipal waste incinerators and will be replaced by 2000/76/EC. The Agency is of the opinion that BAT / BATNEEC requirements now go beyond the requirements of this Directive.

 Directive 89/369/EEC on the prevention of air pollution from new municipal waste incineration plants

This Directive sets minimum standards for "new" (see directive for definition) municipal waste incinerators and will be replaced by 2000/76/EC. The Agency is of the opinion that BAT / BATNEEC requirements now go beyond the requirements of this Directive.

Air Quality

- Statutory Instrument 1989 No 317, Clean Air, The Air Quality Standards Regulations 1989.
- Statutory Instrument 1997 No 3043, Environmental Protection, The Air Quality Regulations 1997.

Water Quality

- Directive 76/464/EEC on Pollution Caused by Dangerous Substances Discharged to Water contains two lists of substances. List I relates to the most dangerous, and standards are set out in various daughter Directives. List II substances must also be controlled. Annual mean concentration limits for receiving waters for List I substances can be found in SI 1989/2286 and SI 1992/337 the Surface Water (Dangerous Substances Classification) Regulations. Values for List II substances are contained in SI 1997/2560 and SI 1998/389. Daughter Directives cover EQS values for mercury, cadmium, hexachlorocyclohexane, DDT, carbon tetrachloride, pentachlorophenol, aldrin, dieldrin, endrin, isodrin, hexachlorobenzene, hexachlorobutadiene, chloroform, 1,2-dichloroethane, trichloroethane, perchloroethane and trichlorobenzene.
- Other waters with specific uses have water quality concentration limits for certain substances. These are covered by the following Regulations:
 - SI 1991/1597 Bathing Waters (Classification) Regulations;
 - SI 1992/1331 and Direction 1997 Surface Waters (Fishlife) (Classification) Regulations;
 - SI 1997/1332 Surface Waters (Shellfish) (Classification) Regulations;
 - SI 1996/3001 The Surface Waters (Abstraction and Drinking Water) (Classification) Regulations.

Future likely changes include:

Some air and water quality standards may be replaced by new standards in the near future.

Other standards and obligations

Those most applicable to this sector are:

- Large Combustion Plant Directive;
- Reducing Emissions of VOCs and Levels of Ground Level Ozone: a UK Strategy;
- Water Quality Objectives assigned water quality objectives to inland rivers and water courses (ref. Surface (Rivers Ecosystem) Classification);
- The UNECE convention on long-range transboundary air pollution;

- The Montreal Protocol;
- The Habitats Directive (see Section 4.3).

3.2.4 Units for benchmarks and setting limits in Permits

Releases can be expressed in terms of:

- "concentration" (e.g. mg/l or mg/m³) which is a useful day-to-day measure of the effectiveness of any abatement plant and is usually measurable and enforceable The total flow must be measured/controlled as well;
- "specific mass release" (e.g. kg/ tproduct or input or other appropriate parameter) which is a
 measure of the overall environmental performance of the plant (including the abatement plant)
 compared with similar plants elsewhere;
- "absolute mass release" (e.g. kg/hr, t/yr) which relates directly to environmental impact.

When endeavouring to reduce the environmental impact of an installation, its performance against each of these levels should be considered, as appropriate to the circumstances, in assessing where improvements can best be made.

When setting limits in Permits the most appropriate measure will depend on the purpose of the limit. It may also be appropriate to use surrogate parameters which reflect optimum environmental performance of plant as the routine measurement, supported by less frequent check-analyses on the final concentration. Examples of surrogate measures would be the continuous measurement of conductivity (after ion-exchange treatment) or total carbon (before a guard-column in activated carbon treatment) to indicate when regeneration or replacement is required.

In all cases, where European legislation applies, suitable limits should be set that enable direct comparison of installation performance against the relevant legislative standard.

3.2.5 Statistical basis for benchmarks and limits in Permits

Conditions in Permits can be set with percentile, mean or median values over yearly, monthly or daily periods, which reflect probable variation in performance. In addition absolute maxima can be set.

Where there are known failure modes, which will occur even when applying BAT, limits in Permits may be specifically disapplied but with commensurate requirements to notify the Regulator and to take specific remedial action.

For Water: UK benchmarks or limits are most frequently 95 percentile concentrations or absolute concentrations, (with flow limited on a daily average or maximum basis).

For Air: benchmarks or limits are most frequently expressed as daily averages or, typically 95% of hourly averages.

In all cases, where European legislation applies, suitable standards should be set that enable direct comparison of installation performance against the relevant legislative standard.

3.2.6 Reference conditions for releases to air

The reference conditions of substances in releases to air from point sources are stated in European legislation. They are:

Incineration plants

- temperature 273 K (0°C), pressure 101.3 kPa (1 atmosphere), 11% oxygen, dry gas.

Waste Oil Incinerators

- temperature 273 K (0°C), pressure 101.3 kPa (1 atmosphere), 3% oxygen, dry gas.

Co-incinerators

- refer to HWID or WID as appropriate.

These reference conditions relate to the benchmark release levels given in this Note and care should always be taken to convert benchmark and proposed releases to the same reference conditions for comparison. The Permit may employ different reference conditions if they are more suitable for the process in question, but the standards of European legislation must be included to enable direct comparison.

To convert measured values to reference conditions see Technical Guidance Note M2 (Ref. 20) for more information.

3.3 Releases to Air

The incineration of waste is subject to a number of European Directives. This section outlines the standards required by those directives in respect of releases to air. In addition to the legislative standards, the use of BAT may result in additional emission reductions.

3.3.1 Introduction to the European Directive Emission Limits

Where an installation falls within the scope of a particular Directive, the standards set by that Directive apply **as a minimum**. In some circumstances the use of BAT may result in tighter controls.

Each of the Directives contains sections which define the scope of the directive and its provisions. The interpretation of the scope of a particular directive will depend upon legal interpretation of the Directive.

National governments have the responsibility for implementing the enabling legislation that brings European Directives into force, and for appointing an appropriate Regulator. The enabling legislation generally takes the form of Regulations or Direction by the Secretary of State. Once the enabling legislation has been passed the appointed Regulator is required to enforce the requirements of the Directive as advised by Government.

At the time of writing this guidance, the "new" waste incineration directive or WID (2000/76/EC), whilst in force at a European level, remains to be implemented by means of national legislation. For the purposes of this guidance the requirements of the Directive itself have therefore been considered. Any subsequent implementing legislation will need to be taken into account in due course.

The WID is of particular relevance because it will repeal earlier directives on municipal and hazardous waste incineration, and in replacing them, substantially broaden the scope of installations that will be effected by European Legislation. There will be very few installations that burn waste that do not fall within the scope of WID. The Agency intends to produce separate "Regulatory Guidance" which will outline the scope and application of WID in more detail.

It is **important** that Operators of existing installations and those who propose any new installation **contact the Agency at an early stage** to discuss this matter if:

- They intend to burn any waste
- They intend to burn any fuel derived from waste

3.3.2 Waste Incineration Directive 2000/76/EC

The Waste Incineration Directive has been developed to fill the gaps between existing Directives on municipal waste incineration and hazardous waste incineration. The existing Directives dealing with municipal and hazardous waste incineration will be repealed on 28 December 2005, five years after the Waste Incineration Directive comes into force. The Agency intends to issue detailed regulatory guidance on the Directive to assist in interpretation, and provide links as appropriate to this guidance.

Implementation provisions are:

- for new plant: shall comply with this Directive from 28 December 2002;
- for existing plant: shall comply with this Directive no later than 28 December 2005.

Co-incineration plants which start co-incinerating waste not later than 28 December 2004, are to be regarded as existing co-incineration plants. The emission limit values for co-incineration plants are not included here – readers should refer to the Directive and to the sector specific guidance.

The emission limit values for incineration plants falling within WID are summarised in Table 3-1. Readers should consult the Directive for full details.

INTRODUCTION	TECHNIQUES	EMISSIONS	IMPACT
Benchmark Comparison	Benchmark Status	Releases to Air	Releases to Water

Table 3-1 - Waste Incineration D	Directive Annex V: air	emission limit values (ELV)
----------------------------------	------------------------	-----------------------------

		Directive Requirement				
Pollutant	ELV (mg/m ³ unless stated)*	Averaging / Monitoring Period	Monitoring Frequency			
Total Dust	10	Daily average	Continuous			
Total Dust	30	100% 1/2 hourly averages				
Total Dust	10	97% 1/2 hourly averages				
VOCs (as TOC)	10	Daily average	Continuous			
VOCs (as TOC)	20	100% 1/2 hourly averages				
VOCs (as TOC)	10	97% ½ hourly averages				
HCI	10	Daily average	Continuous			
HCI	60	100% 1/2 hourly averages	(periodic may be used where			
HCI	10	97% ½ hourly averages	emission <i>cannot</i> exceed ELV)			
HF	1	Daily average	Continuous			
HF	4	100% 1/2 hourly averages	(periodic may be used where			
HF	2	97% ½ hourly averages	emission cannot exceed ELV or where HCI ELVs complied with)			
SO ₂	50	Daily average	Continuous			
SO ₂	200	100% 1/2 hourly averages	(periodic may be used where			
SO ₂	50	97% 1/2 hourly averages	emission cannot exceed ELV)			
NOx (nitrogen monoxide and nitrogen dioxide expressed as nitrogen dioxide)	200	Daily average	Continuous			
NOx (as above)	400	100% 1/2 hourly averages				
NOx (as above)	200	97% ½ hourly averages				
СО	50	Daily average	Continuous			
СО	150	95% of 10min averages				
СО	100	100% of ½ hourly averages				
Cd and TI	total 0.05	All average values over the	Periodic – 2 per year but one every			
Hg	0.05	sample period (30 minutes	3 months during the first year of			
Sb, As, Pb, Cr, Co, Cu, Mn, Ni and V	total 0.5	these limits				
Dioxins and furans	0.1 ng/m ³ TEQ	CEN method (EN 1948, parts 1, 2 and 3) sample period 6 to 8 hours	Periodic – 2 per year but one every 3 months during the first year of operation.			

Notes:

1. * reference conditions: 273 K, 101.3 kPa, 11% O₂, dry gas.

2. ELVs apply at all times when waste is being burned (except for CO during start-up and shut down)

3. The ELVs for metals include solid, gaseous and vapour forms as well as their compounds.

4. TEQ should be calculated as described in annexe I of the Directive

5. The Directive provides for certain derogations in respect of NOx and particulate emissions from existing plants. These are not generally expected to be applicable in the UK as BAT will achieve the required ELV.

6. The Agency will generally apply the 100% ile limits for both daily and ½ hourly ELVs.

7. Derogation from the above CO ELVs is available for fluidised beds up to 100mg/m³ as an hourly average.

8. The Waste Incineration Directive goes beyond emission limit values. Permits will also be required to include an extensive range of conditions to ensure high operational standards.

9. Monitoring techniques should be CEN or where not available, national or international standards. Further guidance is given in Section 2.10.

3.3.3 Hazardous Waste Incineration Directive 94/67/EC

The Hazardous Waste Incineration Directive (HWID) was implemented in the UK early 1998 by a combination of Secretary of State Direction and Regulations (SI1998 No.767). A Regulatory Guidance Note issued 14 July 1999, and other supplementary guidance on clinical waste incineration, provides detailed guidance and is available from the Agency on request.

When hazardous waste is burned, this directive must be complied with, until superseded by WID (see Section 3.3.2 above).

The Directive recognises the burning of hazardous waste in plants not intended primarily for that purpose, such as cement and lime kilns. The Directive requires that if more than 40% of the heat input to a kiln is supplied by fuel classified as hazardous waste under Article 2(1), then the requirements in table 3-2 must be applied. Operators should note that the hazardous waste list has been revised (Ref. 2000/532/EC), which may have the effect of extending the scope of the HWID. Further amendments to the list are planned to take effect by 1 January 2002.

If the heat obtained from burning hazardous waste is 40% or less of the total kiln heat input at any time then, in accordance with Annex II of the Directive, the emission limit values in Table 3-2 only apply to that portion of the exhaust gases generated by combustion of the waste. Overall emission limit values are based on pro-rating between the limits in Table 3-2, any existing authorised kiln limits based on the exhaust gas flows resulting from incinerating the waste and the flows from the kiln process when fired on non-hazardous waste fuels.

	Emission li	imits (mg/m ³)*	Monitoring	Compliance conditions**	
Substance	Daily average	Half hourly average	requirements		
Particulates***	10	30 (10)	continuous	Daily averages to be	
VOCs as carbon***	10	20 (10)	continuous	less than these limits;	
HCI***	10	60 (10)	continuous	averages over a year	
HF***	1	4 (2)	spot where HCI can be used as a surrogate	to be less than limits; or 97% of 30 min averages over a year	
SO ₂ ***	50	200 (50)	continuous	bracketed limits	
со	50	100 (150)	continuous	Daily averages to be less than these limits; and either all 30 min averages in 24 hours to be less than limits; or 95% of 10 min averages over a year to be less than bracketed limits	
	New plants	Existing plants			
Cd and TI in total***	0.05	0.1	Every 2 months	All average values	
Hg***	0.05	0.1	of operation.	over the sample period (30 minutes to	
Pb, Cr, Cu, Mn, Ni, As, Sb, Co, V and Sn in total***	0.5	1	then twice per year	8 hours) to be less than these limits	
Dioxins TEQ (ng/m ³)	0.1	0.1		CEN method (EN 1948, parts 1,2 and 3) sample period 6 to 8 hours	

Table 3-2 - Hazardous Waste Incineration Directive: Emission limit values

* Reference conditions are dry, temperature 273 K, pressure 101.3 kPa, 11% oxygen

** Permits usually specify half-hourly limits as all 30-minute averages over a year to be less than limits.

^{***} Measurements of these pollutants shall not be necessary, provided that the Permit allows the incineration of only those hazardous wastes which cannot cause average values of those pollutants higher than 10% of the emission limit values. This is important for plants burning low chlorine or low sulphur wastes.

INTRODUCTION	TECHNIQUES	EMISSIONS	IMPACT
Benchmark Comparison	Benchmark Status	Releases to Air	Releases to Water

The Directive requirements go beyond air emission limit values. Regulators are required to:

- specify types/ quantities of hazardous waste which can be used in the process;
- set conditions for delivery/ receipt of wastes;
- set operating conditions such as minimum temperature, residence time, % oxygen;
- · monitor releases as specified in the Directive.

In addition, where the hazardous waste provides 40% or less of the thermal input to the process, the Permit shall specify the minimum and maximum flow rates of hazardous wastes, minimum and maximum calorific values and the maximum concentration of pollutants, for example PCB, PCP, chlorine, fluorine, sulphur and heavy metals.

Existing Municipal Waste Incinerators Directive 89/429/EEC

Whilst setting the current European legislative baseline standards for all existing MWIs, the emission limits and other requirements specified in this directive are no longer considered to reflect the application of BAT (or BATNEEC). All installations are required to meet the standards of the "new" WID from 28 December 2005 at the latest.

New Municipal Waste Incinerators Directive 89/369/EEC

Whilst setting the current European legislative baseline standards for all existing MWIs, the emission limits and other requirements specified in this directive are no longer considered to reflect the application of BAT (or BATNEEC). All installations are required to meet the standards of the "new" WID from 28 December 2005 at the latest.

Other applicable Standards and obligations

(Extracts from standards are quoted for ease of reference. The relevant standards should be consulted for the definitive requirements)

Statutory Instrument 1989 No 317, *Clean Air, The Air Quality Standards Regulations 1989* gives limit values in air for nitrogen dioxide.

Statutory Instrument 1997 No 3043, *Environmental Protection, The Air Quality Regulations 1997* gives air quality objectives to be achieved by 2005 for nitrogen dioxide

The UNECE convention on long-range transboundary air pollution - Negotiations are now underway which could lead to a requirement further to reduce emissions of NO_x.

Statutory Instrument 1989 No 317, Clean Air, The Air Quality Standards Regulations 1989 gives limit values in air for suspended particulates.

Statutory Instrument 1997 No 3043, *Environmental Protection, The Air Quality Regulations 1997* gives air quality objectives to be achieved by 2005 for PM₁₀

Statutory Instrument 1989 No 317, Clean Air, The Air Quality Standards Regulations 1989 gives limit values in air for sulphur dioxide.

Statutory Instrument 1997 No 3043, *Environmental Protection, The Air Quality Regulations 1997* gives air quality objectives to be achieved by 2005 for sulphur dioxide

The UNECE convention on long-range transboundary air pollution. Under this Convention, a requirement further to reduce SO₂ emissions *from all sources* has been agreed. The second Sulphur Protocol (Oslo, 1994) obliges the UK to reduce SO₂ emissions by 80% (based on 1980 levels) by 2010.

"Reducing Emissions of VOCs and Levels of Ground Level Ozone: A UK Strategy" was published by the Department of the Environment in October 1993. It sets out how the Government expects to meet its obligations under the UNECE VOCs Protocol to reduce its emissions by 30% (based on 1988 levels) by 1999, including the reductions projected for the major industrial sectors. Although Pulp and Paper is included in the "other miscellaneous industries" sector, no specific reduction targets are stated.

The UNECE convention on long-range transboundary air pollution - Negotiations are now under way which could lead to a requirement further to reduce emissions of VOCs.

3.3.4 Municipal waste incinerators

Standards and obligations for MWIs

- Existing installations must meet the requirements of the *current* municipal waste incineration directives (the 1989 directives) as an absolute minimum. It is unlikely that operation to this standard will be acceptable, as the use of BAT will result in improved environmental performance.
- All existing installations must comply with WID standards as a minimum from 28 December 2005 at the latest.
- · All new plants must comply with WID standards as a minimum with immediate effect.
- Although compliance with WID standards will be taken to indicate that BAT is in use, Operators will be required to demonstrate why further emission reductions cannot be achieved through the use of BAT. This will be particularly the case at larger throughput plant, or where local environmental conditions require installation pollutant contributions to background levels are further reduced.

Substance(s)	Currently reported (mg/m ³ unless stated)	Benchmark ELV	Comments
Total particulate	1 - 28	WID	Bag filters are capable of meeting WID standard.
			Wet scrubbing or EP are not adequate on their own
VOCs (as TOC)	<1 - <8	WID	Combustor design and combustion control important.
HCI	1 - 98	WID	Optimisation of alkaline dosing by upstream HCI monitoring and feed forward dosing control or extensive waste selection/ pre-treatment required to control short term peaks
			avoid excessive alkaline reagent dosing
			wet systems may meet standard but effluent treatment required.
			Waste pre-treatment may be required to prevent peaks
HF	0.1 – 2.8	WID	See measures under HCI
SO ₂	<3 - 479	WID	See measures under HCI
NOx (NO and	276 - 479	WID	Some existing plant will require upgrade.
NO ₂ as NO ₂)			Combustor and grate design and combustion control important.
			FGR likely to be BAT for all new plant and existing plant except where high retrofit costs.
			Fluid bed may offer advantages with treated waste feeds
			SNCR (urea or ammonia) required to guarantee ELVs.
			All (particularly new) plant must justify why SCR and ELV of 100mg/m3 is not BAT.
NH ₃		10 mg/m ³ daily avg.	Not covered by European legislation.
		20 mg/m ³ 1/2hr avg.	Limits to be applied to control slip from NOx control systems
N ₂ 0		30 mg/m ³ avg. over	Not covered by European legislation.
		8 hr sample period	Limits to be applied to control slip from NOx control systems
СО	3 - 198	WID	Combustor design and combustion control important.
			Charging systems need control
Cd and TI	0.02 - 0.12	WID	Avoid excessive furnace temperatures
			Barrier filtration required to control particulate bound metals.
Hg	0.005 - 0.08	WID	Carbon injection required.
Sb, As, Pb, Cr,	0.04 - 0.53	WID	Avoid excessive furnace temperatures
Co, Cu, Mn, Ni and V			Barrier filtration required to control particulate bound metals
Dioxins and	0.01 – 0.14 ng/m3	WID	Combustor design and combustion control important.
furans			FGR likely to be BAT for all new plant and existing plant except where high retrofit costs.
			Carbon injection required.
			Use of SCR to be considered in BAT assessment.
			Use of catalytic filter bags to be considered in BAT assessment

Benchmark emission values

3.3.5 **Clinical waste incinerators**

Standards and Obligations for CWIs

- Where hazardous wastes are burned existing installations must meet the requirements HWID as an • absolute minimum.
- All existing installations must comply with WID standards as a minimum from 28 December 2005 at the latest.
- All new plants must comply with WID standards as a minimum with immediate effect as BAT
- Although compliance with WID standards will be taken to indicate that BAT is in use, Operators will be required to demonstrate why further emission reductions cannot be achieved through the use of BAT. This will be particularly the case at larger throughput plant, or where local environmental conditions require installation pollutant contributions to background levels are further reduced.

Benchmark emission values

Substance(s)	Currently reported (mg/m ³ unless stated)	Benchmark ELV	Comments
Total particulate	5 (daily) - 20 (half	WID / HWID	See comments for MWIs
	hour)		Ceramic filters may provide an alternative to fabric for smaller plant where higher temperate filtration required
VOCs (as TOC)	5 (daily) – 10(half	WID / HWID	See comments for MWIs
	nour)		Good secondary combustion required if semi-pyrolytic primary stages used
HCI	5 (daily) – 30 (half	WID / HWID	See comments for MWIs
	hour)		HCI peak loading and variation generally less than MWIs
			Avoid use of PVC waste packaging
			Dry scrubbing with reagent recycle can also be BAT
HF	0.5 (daily) – 2 (half hour)	WID /HWID	See comments for HCI
SO ₂	20 (daily) – 50 (half hour)	WID /HWID	See comments for HCI
NOx (NO and NO ₂ as NO ₂)	100 (daily average)	WID	Relatively smaller plant size and lower NOx emission concentrations than MWIs
			Improvements achieved by abatement beyond WID unlikely to be BAT
			Staged combustion may assist to reduce NOx formation
NH ₃		10 mg/m ³ daily avg.	Not covered by European legislation.
		20 mg/m ³ 1/2hr avg.	Limits to be applied to control slip from NOx control systems
N ₂ 0		30 mg/m ³ avg. over 8 hr	Not covered by European legislation.
		sample period	Limits to be applied to control slip from NOx control systems
СО	20 (daily) – 100 (10	WID / HWID	See comments for MWIs
	min)		Good secondary combustion required if semi pyrolytic primary stages used
Cd and TI	0.02	WID / HWID	See comments for MWIs
Hg	0.02	WID / HWID	See comments for MWIs
			May be opportunities to reduce Hg loading by up stream waste control
Sb, As, Pb, Cr, Co, Cu, Mn, Ni and V	0.2	WID / HWID	See comments for MWIs
Dioxins and furans	0.05 ng/m3	WID / HWID	See comments for MWIs
Note: 1 Quoted per	formance figures may not	be directly comparable owin	a to different averaging periods

1. Quoted performance figures may not be directly comparable owing to different averaging periods.

Performance figures derived form plant upgrade guarantees 3.

3.3.6 Chemical waste incinerators

This section relates to dedicated chemical, hazardous waste incinerators. The standards applicable to other installations that co-incinerate hazardous wastes are covered in Section 3.3.12.

Standards and Obligations for ChWls

- Existing installations must meet the requirements HWID as an absolute minimum.
- Existing installations must comply with WID standards from 28 December 2005 at the latest.
- New plants must comply with HWID and WID standards as a minimum with immediate effect.
- Although compliance with HWID (or WID) standards will be taken to indicate that BAT is in use, Operators will be required to demonstrate why further emission reductions cannot be achieved through the use of BAT. This will be particularly the case at larger throughput plant, or where local environmental conditions require installation pollutant contributions to background levels are further reduced.

Benchmark emission values

Substance(s)	Currently reported (mg/m ³ unless stated)	Benchmark ELV	Comments
Total particulate	1 – 4	WID / HWID	Multi stage scrubbers may achieve ELVs
			Bag filtration may require gas reheat
VOCs (as TOC)	< 2	WID / HWID	High temperatures ensure good VOC destruction
HCI	0.2 – 2	WID / HWID	Multi stage scrubbers achieve low emissions to air
HF	0.02 – 2	WID /HWID	See HCI
SO ₂	0.1 – 1	WID /HWID	See HCI
NOx (NO and NO ₂ as NO ₂)	125 – 240	WID	Low throughputs mean NOx unlikely to represent significant pollution problem
			Achieving WID emission ELV may require additional controls (WID exemption only applies to plants burning ONLY hazardous waste)
NH ₃		10 mg/m ³ daily avg.	Not covered by European legislation.
		20 mg/m ³ 1/2hr avg.	Limits to be applied to control slip from NOx control systems
N ₂ 0		30 mg/m ³ avg. over 8 hr	Not covered by European legislation.
		sample period	Limits to be applied to control slip from NOx control systems
CO	1 – 71	WID / HWID	Low ELVs achieved
			Short term concentrations in WID may require attention
Cd and TI	0.001 - 0.024	WID / HWID	
Hg	0.001 – 1	WID / HWID	Care required where high input concentrations may arise
Sb, As, Pb, Cr, Co, Cu, Mn, Ni and V	0.02 - 0.5	WID / HWID	
Dioxins and furans	0.0007 – 0.03 ng/m3	WID / HWID	exceptionally low emission levels now achieved
			SCR, C injection and catalytic filter bags may offer further improvement where not employed.

3.3.7 Sewage sludge incinerators

Standards and Obligations for SSIs

- All existing installations must comply with WID standards as a minimum from 28 December 2005 at the latest.
- All new plants must comply with WID ELVs as a minimum with immediate effect. With the
 exception of NOx where this is derived from high fuel nitrogen and the relevant WID ELV cannot be
 achieved by means of primary measures on their own. Where this is the case exemption may be
 authorised until 28 December 2002.
- Although compliance with WID standards will be taken to indicate that BAT is in use, Operators will be required to demonstrate why further emission reductions cannot be achieved through the use of BAT. This will be particularly the case at larger throughput plant, or where local environmental conditions require installation pollutant contributions to background levels are further reduced.

Benchmark emission values

Substance(s)	Currently reported (mg/m ³ unless stated)	Benchmark ELV	Comments
Total particulate	1.5 – 25	WID	EPs or bag filters appear capable of meeting ELVs
			Wet scrubbers on their own not sufficient
			Fluidised beds require grit arrestment (cyclones?)
VOCs (as TOC)	1.4 – 28.6	WID	Temperatures and residence time needs to be sufficient to burn off VOCs generated from this high organic waste
HCI	0.7 – 17	WID	Not generally problematic
HF	<0.1 – 2.5	WID	Not generally problematic
SO ₂	12.3 – 250	WID	May be some difficulty where high sulphur feedstock
			Elimination of sulphur from feedstock where possible may help e.g. Sulphates used for precipitation of sludges
			Wet caustic scrubbing used in some cases
NOx (NO and NO ₂	119 – 709	WID	High fuel nitrogen may lead to difficulties in reaching ELVs
as NO ₂)			Unlikely to be BAT for existing plant to upgrade to WID standards before WID deadline.
			Use of fluidised beds may help reduce thermal NOx addition
NH₃		10 mg/m ³ daily avg.	Not covered by European legislation.
		20 mg/m ³ 1/2hr avg.	Limits to be applied to control slip from NOx control system
N ₂ 0		30 mg/m ³ avg. over 8 hr	Not covered by European legislation.
		sample period	Limits to be applied to control slip from NOx control system
СО	5 – 100	WID	Combustor design and combustion control important.
			Charging systems need control to ensure consistent feed
			Fluidised beds must achieve at least 100mg/m3 as an hourly average.
Cd and TI	0.001 – 0.05	WID	Metals loading in sludge may be elevated in industrial catchments – prevention of large point discharges to upstream sewer may be cost effective in some cases
			Particulate reduction techniques will reduce metals emissions e.g. filtration techniques
Hg	<0.01 – 0.05	WID	Metals loading in sludge may be elevated in industrial catchments – prevention of large point discharges to upstream sewer may be cost effective in some cases
			Carbon injection may be required
Sb, As, Pb, Cr, Co, Cu, Mn, Ni and V	0.08 – 0.5	WID	Metals loading in sludge may be elevated in industrial catchments – prevention of large point discharges to upstream sewer may be cost effective in some cases
Dioxins and furans	0.015 – 0.1ng/m3	WID	Emissions of dioxins and furans generally very low
			Carbon injection and other techniques may be applicable

3.3.8 Animal carcass and animal remains incinerators

Standards and Obligations for ACIs

Although exempted from WID, BAT for all new animal carcass incinerators will be to meet emission limit values detail in WID from the first day of operation. However, it is possible that NOx. emissions higher than WID will be BAT owing to relatively high fuel nitrogen levels and the costs of NOx abatement. This should not prevent the use or primary NOx measures at the combustion stage.

All animal remains incinerators (e.g. MBM, SRM) and any animal carcass incinerators that burn anything other than animal carcasses will be covered by WID:

- All existing installations must comply with WID standards as a minimum from 28 December 2005 at the latest.
- All new animal remains incinerators must comply with WID standards as a minimum with immediate effect.

Substance(s)	Currently reported (mg/m ³ unless stated)	Benchmark ELV	Comments
Total particulate	<25	WID (see comments column)	Non WID plants are expected to achieve at least 25mg/m³ as a daily average and must justify why WID standard is not achieved
VOCs (as TOC)		WID (see comments column)	Non WID plants are expected to achieve at least 20mg/m ³ as a daily average and must justify why WID standard is not achieved
HCI		WID (see comments column)	Non WID plants are expected to achieve at least 30mg/m ³ as a daily average and must justify why WID standard is not achieved
HF		WID	Non WID plants are also expected to achieve WID standards
SO ₂	Some short term peaks of 300 to 1000	WID (see comments column)	Sulphur loading may be significant and require higher alkaline reagent dose rates.
			Non WID plants are expected to achieve WID daily average and at least 300mg/m ³ half-hour average.
NOx (NO and NO ₂ as NO ₂)	200 - 300	WID (see comments column)	Non WID plants are expected to achieve at least 300mg/m ³ as a daily average and must justify why WID standard is not achieved
NH ₃		10 mg/m ³ daily avg.	Not covered by European legislation.
		20 mg/m ³ 1/2hr avg.	Limits to be applied to control slip from NOx control system
N ₂ 0		30 mg/m ³ avg. over 8 hr sample period	Not covered by European legislation.
со		WID (see comments column)	Non WID plants are expected to achieve at least 150mg/m ³ as a half-hour average and 50 mg /m ³ as a daily average, and must justify why WID standards cannot be achieved.
Cd and Tl		WID	Non WID plants are also expected to achieve WID standards
			Metals levels in feed stock very low.
Hg		WID	Non WID plants are also expected to achieve WID standards
			Metals levels in feed stock very low.
Sb, As, Pb, Cr, Co, Cu, Mn, Ni and V		WID	Non WID plants are also expected to achieve WID standards
			Metals levels in feed stock very low.
Dioxins and furans	0.1 – 1.0 ng/m3	WID (see comments column)	Non WID plants are also expected to achieve WID standards
			Carbon injection and other techniques may be applicable to meet 0.1ng/m3.

Benchmark emission values

3.3.9 Drum incinerators

Standards and Obligations for DIs

These installations burn hazardous wastes and therefore fall within the control of HWID. The quantities of the hazardous materials actually combusted is likely to be far lower than those in dedicated HWIs and the process more likely to be of a batch nature.

- Existing installations must meet the requirements HWID as an absolute minimum.
- Existing installations must comply with WID standards from 28 December 2005 at the latest.
- New plants must comply with HWID and WID standards as a minimum with immediate effect.
- Although compliance with HWID (or WID) standards will be taken to indicate that BAT is in use, Operators will be required to demonstrate why further emission reductions cannot be achieved through the use of BAT. This will be particularly the case at larger throughput plant, or where local environmental conditions require installation pollutant contributions to background levels are further reduced.

Benchmark emission values

Substance(s)	Currently reported (mg/m ³ unless stated)	Benchmark ELV	Comments
Total particulate	20 – 26	WID / HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
VOCs (as TOC)	1 – 9	WID / HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
HCI	21 – 29	WID / HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
HF	1 – 2	WID /HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
SO ₂	3 – 10	WID /HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
NOx (NO and NO ₂ as NO ₂)		WID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
NH ₃		10 mg/m ³ daily avg.	Not covered by European legislation.
		20 mg/m ³ 1/2hr avg.	Limits to be applied to control slip from NOx control system
N ₂ 0		30 mg/m ³ avg. over 8 hr sample period	Not covered by European legislation.
			Limits to be applied to control slip from NOx control system
СО	205	WID / HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
Cd and TI		WID / HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
Hg		WID / HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
Sb, As, Pb, Cr, Co, Cu, Mn, Ni and V	0.6 – 0.9	WID / HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
			Paint on drums may lead to higher loading
Dioxins and furans	0.13 ng/m3	WID / HWID	Compliance with HWID / WID must be achieved
			Similar issues arising with DIs as HWIs (see above)
3.3.10 Pyrolysis and gasification plants

Standards and Obligations for pyrolysis and gasification

Pyrolysis and gasification processes only avoid obligations arising under the WID if they do not combust the products created i.e. solid char, liquids or the gas evolved. In the majority of cases, in order to make use of the energy and raw material potential of the products created, and to minimise waste production, BAT is likely to determine that the products will be combusted and hence WID apply. In some specific cases all of the products may go on to some subsequent non-fuel (non-combustion) use as a raw material – in this case WID would not apply.

For installations falling within the scope of WID:

- All existing installations must comply with WID standards as a minimum from 28 December 2005 at the latest.
- All new plants must comply with WID standards as a minimum with immediate effect.
- Although compliance with WID standards will be taken to indicate that BAT is in use, Operators will be required to demonstrate why further emission reductions cannot be achieved through the use of BAT. The generally smaller size (and hence mass emission rates) of these installations is likely to mean that, provided directive emission limits can be met, process contributions to background pollution levels will be low.

For installations not falling within WID, their purpose will be to capture the products for subsequent (non-combustion) use as a raw material in another process. Significant releases to air are not therefore anticipated. BAT techniques should be selected for process containment. Appropriate guidance may be found within the chemicals sector guidance on similar issues. **Note:** *It is not known whether any installations of this type exist. It would appear likely that they would be required to include some means of gas flaring as a safety measure, in which case WID would apply.*

Benchmark emission values

These emission values relate to the subsequent combustion stage of a pyrolysis or gasification installation.

The range of reported emissions is generally wide for this sector. This reflects

- the range of waste types burned
- the rage of combustion technologies used (i.e. boilers, engines, gas turbines)
- the relatively early stage of development of some applications (the figures are often derived from pilot trials and research rather than commercially operating plant)
- · the consequently limited and non-standardised published emissions data.

Further details of the available data may be found in the BAT report on waste pyrolysis and gasification.

INTRODUCTION Benchmark Comparison TECHNIQUES Benchmark Status EMISSIONS Releases to Air IMPACT Releases to Water

Substance(s)	Currently reported (mg/m ³ unless stated)	Benchmark ELV	Comments
Total particulate	0.01 – 106	WID	In general compliance with WID can be achieved
VOCs (as TOC)	5 – 10000	WID	Similar comments to those for CO below apply
			Engines appear to have the most difficulty achieving low emission levels.
			Emissions using waste derived products generally higher than primary fuels
			Effective residence time for engines and turbines may need increasing to allow burnout
HCI	0.2 – 49	WID	WID emission levels likely to be achievable using similar gas cleaning equipment to incinerators
			Peak smoothing may be achieved through the combustion of the effectively pre-treated fuels produced
			Higher halogen (x) content fuels will evolve more HX
HF	0.1 – 5	WID	See comments for HCI
SO ₂	1 – 200	WID	See comments for HCI
NOx (NO and NO ₂	5 – 470	WID	Similar emission levels to incineration plants achieved
as NO ₂)			Similar NOx abatement to incinerators required to meet WID (SNCR, SCR)
			Pyrolysis oil may give NOx at upper end of range shown
NH ₃		10 mg/m ³ daily avg.	Not covered by European legislation.
		20 mg/m ³ 1/2hr avg.	Limits to be applied to control slip from NOx control system
			Fuel gas cleaning may be required
N ₂ 0		30 mg/m ³ avg. over 8 hr	Not covered by European legislation.
		sample period	Limits to be applied to control slip from NOx control system
СО	2.5 – 100	WID	Boilers and gas turbines may achieve levels below 10mg./m ³ and 50 mg/m ³ respectively
			Higher emissions (1000mg/m ³ +) found with unabated engines where catalytic converters may achieve <100 mg/m3
			Turbine load and fuel air ration will effect CO production
Cd and TI	0.0007 - 0.09	WID	Performance similar to incineration plant achievable
Hg	0.002 - 0.09	WID	Performance similar to incineration plant achievable
Sb, As, Pb, Cr, Co, Cu, Mn, Ni and V	0.005 - 0.07	WID	Performance similar to incineration plant achievable
Dioxins and furans	0.001 – 1.2	WID	Performance similar to incineration plant
			Little data for engines and gas turbines
			Combustion control, residence time, rapid cooling from 450 – 200 °C and carbon injection may assist.

Note: 1. Quoted performance figures may not be directly comparable owing to different averaging periods.

INTRODUCTION	TECHNIQUES	EMISSIONS	IMPACT
Benchmark Comparison	Benchmark Status	Releases to Air	Releases to Water

3.3.11 Refuse derived fuel

This section deals with the burning of refuse derived fuels in dedicated plant. The co-incineration of refuse derived fuels is dealt with in Section 3.3.12 - Co-incineration.

Standards and Obligations for RDF

Legal interpretation suggests that all refuse derived fuels are likely to be considered wastes and therefore WID will apply. The similarity of techniques and economics of the waste derived fuel sector is likely to give rise to conclusions regarding BAT emission levels (and techniques) that are similar to those of the "waste" sector.

Substance(s)	Currently reported (mg/m ³ unless stated)	Benchmark ELV	Comments
Total particulate	15 – 100	WID	Use of filtration techniques should enable compliance with WID across a range of RDF types
VOCs (as TOC)	<11	WID	Close combustion control required to avoid excursions where waste feed characteristics are variable (e.g. improved oxygen monitoring, waste feed and air trimming)
			WID achievable through the use of BAT
HCI	5 – 167	WID	WID ELVs should be achievable using alkaline reagent injection.
			Short term fluctuations should be less of a problem owing to greater homogeneity of treated fuel.
			Higher acid content (or forming) RDFs will need greater absorbent dosing or wet systems.
HF	<1.3	WID	See HCI
SO ₂	149 – 400	WID	See HCI
NOx (NO and NO ₂ as NO ₂)	100 – 250	WID	Close combustion control required to avoid excursions where waste feed characteristics are variable (e.g. improved oxygen monitoring, waste feed and air trimming)
			WID achievable through the use of BAT
			Fluidised beds achieve low NOx levels.
NH ₃		10 mg/m ³ daily avg.	Not covered by European legislation.
		20 mg/m ³ 1/2hr avg.	Limits to be applied to control slip from NOx control system
N ₂ 0		30 mg/m ³ avg. over 8 hr	Not covered by European legislation.
		sample period	Limits to be applied to control slip from NOx control system
СО	80 – 141	WID	Close combustion control required to avoid excursions where waste feed characteristics are variable (e.g. improved oxygen monitoring, waste feed and air trimming)
			WID achievable through the use of BAT
Cd and TI	0.06 - 0.1	WID	Techniques to reduce particulate will also reduce metals
Hg	0.03 - 0.1	WID	Carbon injection required
Sb, As, Pb, Cr, Co, Cu, Mn, Ni and V	0.27	WID	Techniques to reduce particulate will also reduce metals
Dioxins and furans	0.01 – 1.0	WID	Combustion control, residence time, rapid cooling from 450 – 200 °C and carbon injection may assist.

Benchmark emission values

Note: 1. Quoted performance figures may not be directly comparable owing to different averaging periods.

IPPC

3.3.12 Co-incinerators

Different definitions of co-incineration plants are given in HWID and WID – these are outlined below.

Operators must demonstrate in their application that they will comply with the relevant provisions of the directives. Applications should refer to the directives (and any guidance issued by the Regulator) and should include calculations to show the emission limit required by the relevant directive along with a justification for its selection at BAT.

HWID Co-incinerators:

The HWID definition will only be used for those plants falling under HWID until such a time as they fall within WID.

HWID includes co-incinerators within the definition of incinerators by including

"plants burning [hazardous] wastes as a or additional fuel in any industrial process":

This definition therefore includes incinerators that burn some hazardous waste with non- hazardous waste as well as other industrial plants that burn some hazardous wastes with non waste fuels.

HWID includes emission limits that must be met. It also describes how emission limits should be calculated in situations where less than 40% of the heat raised in the process arises from the combustion of hazardous waste.

The Agency has issued guidance on how these ELVs should be calculated 9an other aspects of HWID applied) in a Regulatory Guidance Note (on the Incineration of Hazardous Waste dated14 July 1999).

In general compliance with the standards indicated in the directive will indicate that the BAT is being used. However a justification should be provided that details why further emission limit reductions cannot be achieved through the use of BAT.

WID Co-incinerators:

The WID definition of "co-incineration plant" is:

"any stationary or mobile plant whose main purpose is the generation of energy or the production of material products and;

- which uses waste as a regular or additional fuel; or
- in which the waste is thermally treated for the purpose of disposal"

This therefore includes installations that burn any amount of hazardous or non-hazardous waste along with non-waste derived fuels. Because the same emission limits apply to the burning of hazardous and non-hazardous wastes, the situation where some hazardous waste is burned with non-hazardous wastes only has relevance in respect of the temperature at which it must be burned and other aspects such as documentation and control of incoming waste (see Section 2.3.1 - Incoming waste handling).

The Agency intends to issue separate "regulatory guidance" on the application of the WID to coincineration plant.

The key aspects are:

- where the purpose of the co-incineration has become the thermal treatment of the waste (rather than the production of energy or products) the plant is regarded as an incineration plant and the emission limits for incineration plant apply in full
- where more than 40% of the heat release comes from the incineration of hazardous waste the emission limits for incineration plant apply in full
- special provisions are included for cement kilns, combustion plants and other industrial sectors
- pro-rata emission limits may be applied in some cases, whilst in others single limits apply.
- Rules for the determination of emission limits are included

In general compliance with the standards indicated by the directive will indicate that the BAT is being used. However a justification should be provided that details why further emission limit reductions cannot be achieved through the use of BAT.

3.4 Releases to Water

This section outlines the emission limit and generic requirements arising from European legislation. The techniques that represent BAT for the control of releases to water are outlined in

- Section 2.3.10 Abatement of point source emission to surface water and sewer
- Section 2.3.12 Control of fugitive emissions to surface water, sewer and ground water

Incinerators often do not produce large volumes of water for disposal. In many cases the use of the techniques outlined in this guidance (see Section 2.2.3 - Water use) will result in the elimination of discharges to water.

The main releases to water that can arise from incineration installations are:

- Air abatement equipment (e.g. wet scrubbers);
- Boiler blow-down;
- Cooling water discharges;
- Road drainage;
- Incoming waste handling areas;
- Raw material storage areas;
- Ash and other residue handling areas;
- On-site effluent treatment.

In general, good design should prevent the production of the majority of these effluents from being produced (see Sections 2.3.10 and 2.3.12).

3.4.1 Standards and obligations for releases to water

Both HWID and WID contain provisions in respect of discharges to water. The following paragraphs outline the main requirements and provide guidance on how they may be fulfilled using BAT. Further guidance on the requirements of the Directives can be found in Agency guidance on their implementation.

3.4.1.1 HWID requirements:

The Agency has issued Regulatory Guidance on the application of HWID. This section outlines the basic requirements.

- 1 Discharges to the aquatic environment must be limited as far as possible
- 2 Relevant Community, National and Local legislation must be respected when setting release levels
- 3 Discharges of heavy metals, dioxins and furans to water must be smaller in mass than those to air
- 4 Sites and associated waste storage areas should be designed an operated in such a manner as to prevent the release of polluting substances into soil and groundwater
- 5 Storage capacity to be provided for rainwater run-off and for contaminated water arising from spillages and fire fighting operations. The storage capacity should be sufficient to allow for such waters to be tested and treated prior to discharge where necessary.

3.4.1.2 WID requirements:

The Agency intends to produce guidance on the application of WID, which should be referred to when available. This section provides an outline of that guidance.

Emission limit values

Article 8 of the Directive sets specific emission limit values for water discharges derived from air pollution control equipment (e.g. wet scrubbers or wet EPs), (see Section 3.4.2 below for further guidance).

General requirements

Article 8(7) also sets out some more general requirements in respect of the design and operation of plant such that sites as a whole "...prevent unauthorised and accidental release of polluting substances into soil, surface water or ground water..".

INTRODUCTION	TECHNIQUES	EMISSIONS	IMPACT
Benchmark Comparison	Benchmark Status	Releases to Air	Releases to Water

This applies to all potential sources of pollution on the site, not just to the releases form air pollution control equipment. BAT will normally fulfil the Directive requirements (see Sections 2.3.10 and 2.3.12).

Storage capacity

The Directive specifically states that "...**storage capacity** shall be provided for contaminated rainwater run-off from the... site or for contaminated water arising from spillage or fire fighting operations."

In the context of this requirement, "contaminated rainwater" should also be taken to include all areas where there is reasonable risk of contamination by the installation. This would normally exclude roof water, access road drainage and office or other ancillary operations, where the risk of contamination is likely to be minimal.

In general, contamination of rainwater and spillages should be avoided using BAT, (see Sections 2.3.10 and 2.3.12).

A **risk assessment** process should be used to determine the volume of storage that is required to contain firewater.

At **new sites** it is likely that it will be possible for site drainage to be engineered such that complete containment is provided. This may include, for example, the use of bunding, or the routing of drainage to a holding tank or an on site effluent treatment plant using an emergency valve.

At **existing plants** the cost of civil works and land availability may present significant difficulties. In such cases it will still be necessary for the operator to comply with the Directive. Whilst <u>adequate</u> storage must be provided, in reaching a decision on the actual size of containment required the risk assessment should, in particular, take account of the following factors:

- use of BAT to avoid rainwater contamination;
- average peak rainfall rates (one per year event average);
- existing drainage capability or concerns;
- adequacy of fire prevention and detection measures (advice from Fire Brigade may be required);
- use of BAT for spill prevention / containment;
- cost of additional storage provision;
- advice from Agency water pollution control officers;
- Sensitivity of receiving medium (e.g. local sewer or nature reserve?);;
- Availability of additional off site holding capacity;
- ability to "test and treat before discharge where necessary" (see below).

Article 8(7) also states that "...The storage capacity shall be adequate to ensure that such waters can be tested and treated before discharge where necessary."

This effectively means that where the drainage of such effluents will be to soil, surface or groundwater, suitable provision must be made for complete containment to allow for assessment prior to release. This may include the use of road tankers brought in to deal with effluent, or the provision of an effluent treatment plant with sufficient buffer capacity to allow adequate time for relevant testing and treatment. The tests required will depend on the nature of the ETP, and its ability to treat the pollutants arising.

3.4.2 Legislative emission limit values

This section contains details of the minimum emission limit values that arise from the WID.

It is *important* to stress that, while they represent the minimum performance criteria in respect of effluents arising at incineration installations (particularly those effluents arising form air pollution control devices), *they <u>DO NOT</u> represent BAT for discharges form the site*.

BAT emission limit values for discharges from the site will need to be selected following consideration of the guidance on techniques given in Sections 2.3.10 and 2.3.12 and other cross sector guidance on effluent treatment.

Annexe IV of WID lists the emission limit values for discharge of water (controlled waters or sewer) from air pollution control devices. The point at which they apply, and how combined effluents should be dealt with are also included in the Directive

- The operational control parameters of pH temperature and flow apply to all effluents (including scrubber discharges), whereas
- the ELVs for suspended solids, metals and dioxins only apply where air pollution control liquors are discharged (alone or in combination with other effluents).

The table below gives the ELVs and monitoring requirements.

INTRODUCTION	TECHNIQUES	EMISSIONS	IMPACT
Benchmark Comparison	Benchmark Status	Releases to Air	Releases to Water

Directive Emission limits into Water		
Parameter	Emission Limit Value	Directive Monitoring Requirements
Total suspended solids as defined by Directive 91/271/EEC	95%<30mg/l and 100%<45mg/l	Spot daily sample or 24-hour flow proportional
Mercury and its compounds, expressed as mercury mg/l (from APC effluents)	0.03	Monthly 24-hour flow proportional sample
Cadmium and its compounds, expressed as cadmium mg/l (from APC effluents)	0.05	Monthly 24-hour flow proportional sample
Thallium and its compounds, expressed as thallium mg/l (from APC effluents)	0.05	Monthly 24-hour flow proportional sample
Arsenic and its compounds, expressed as arsenic mg/l (from APC effluents)	0.15	Monthly 24-hour flow proportional sample
Lead and its compounds, expressed as lead mg/l (from APC effluents)	0.2	Monthly 24-hour flow proportional sample
Chromium and its compounds, expressed as chromium mg/l (from APC effluents)	0.5	Monthly 24-hour flow proportional sample
Copper and its compounds, expressed as copper mg/l (from APC effluents)	0.5	Monthly 24-hour flow proportional sample
Nickel and its compounds, expressed as nickel mg/l (from APC effluents)	0.5	Monthly 24-hour flow proportional sample
Zinc and its compounds, expressed as Zinc mg/l (from APC effluents)	1.5	Monthly 24-hour flow proportional sample
Total Dioxins and furans (as TEQ) ng/l (from APC effluents)	0.3	6 monthly 24-hour flow proportional sample
pH range	Site specific	continuous
Temperature °C	Site specific	continuous
Flow I/s	Site specific	continuous

Notes:

1. 2.

Total suspended solids limits apply as <u>spot daily samples</u> OR monthly <u>24hr flow proportional sample</u>. Limits for metals apply as 24hr flow proportional samples. Only 1 sample per year **OR** 5% of annual samples (where more than 20 samples are taken) may exceed the limits stated above. Temperature, flow and pH limits to be measured and applied continuously.

3.

4 IMPACT

4.1 Assessment of the Impact of Emissions on the Environment

The Operator should assess that the emissions resulting from the proposals for the activities/installation will provide a high level of protection for the environment as a whole, in particular having regard to EQSs. etc, revisiting the techniques in Section 2 as necessary (see Section 1.2).



With the Application the Operator should:

- 1. provide a description, including maps as appropriate, of the receiving environment to identify the receptors of pollution. The extent of the area may cover the local, national and international (e.g. transboundary effects) environment as appropriate.
- 2. identify important receptors which may include: areas of human population including noise or odour-sensitive areas, flora and fauna (i.e. Habitat Directive sites, special areas of conservation, Sites of Special Scientific Interest (SSSI or in Northern Ireland, ASSI) or other sensitive areas), soil, water, i.e. groundwater (water below the surface of the ground in the saturation zone and in direct contact with the ground and subsoil) and watercourses (e.g. ditches, streams, brooks, rivers), air including the upper atmosphere, landscape, material assets and the cultural heritage.
- 3. identify the pathways by which the receptors will be exposed (where not self evident).
- 4. carry out an assessment of the potential impact of the total emissions from the activities on these receptors. Ref. 5 provides a systematic method for doing this and will also identify where modelling needs to be carried out, to air or water, to improve the understanding of the dispersion of the emissions. The assessment will include comparison (see IPPC A Practical Guide (Ref. 3) and (Section 3.2) with:
 - community EQS levels;
 - other statutory obligations;
 - non-statutory obligations;
 - environmental action levels (EALs) and the other environmental and regulatory parameters defined in Ref. 5.

in particular, it will be necessary to demonstrate that an appropriate assessment of vent and chimney heights has been made to ensure that there is adequate dispersion of the minimised emission(s) to avoid exceeding local ground-level pollution thresholds and limit national and transboundary pollution impacts. This should be based on the most sensitive receptor, be it human health, soil or terrestrial ecosystems.

for this sector, where higher emission concentrations are expected during shorter averaging periods (e.g. half hourly averages) these should be used as the input parameters for short term modelling assessments, with the lower longer term (typically daily average) concentrations used for predicting longer term impacts. A conservative approach should be adopted, such that impacts may reliably be expected to be less than those expected. An estimate of the degree of this overestimation may be included provided that its derivation is clearly explained along with any other relevant uncertainties.

where appropriate the Operator should also recognise the chimney or vent as an emergency emission point and understand the likely behaviour. Process upsets or equipment failure giving rise to abnormally high emission levels over short periods should be assessed. Even if the Applicant can demonstrate a very low probability of occurrence, the height of the chimney or vent should nevertheless be set to avoid any significant risk to health. The impact of fugitive emissions can also be assessed in many cases.

consider whether the responses to Sections 2 and 3 and this assessment adequately demonstrate that the necessary measures have been taken against pollution, in particular by the application of BAT, and that no significant pollution will be caused. Where there is uncertainty about this, the measures in Section 2 should be revisited as appropriate to make further improvements.

INTRODUCTION	TEC	HNIQUES	EMISSIO	NS	IMPACT
Impact Assessment		Waste Manageme	ent Licensing Regs		Habitats Directive

^{5.} where the same pollutants are being emitted by more than one permitted activity on the installation the Operator should assess the impact both with and without the neighbouring emissions.

Habitats Directive

4.2 The Waste Management Licensing Regulations



In relation to activities involving the disposal or recovery of waste, the Regulators are required to exercise their functions for the purpose of achieving the relevant objectives as set out in Schedule 4 of the Waste Management Licensing Regulations 1994. (For the equivalent Regulations in Scotland, see Appendix 2. In Northern Ireland there are no equivalent regulations at the time of writing. Contact EHS for further information.)

The relevant objectives, contained in paragraph 4, Schedule 4 of the Waste Management Licensing Regulations 1994 (*SI 1994/1056 as amended*) are extensive, but will only require attention for activities which involve the recovery or disposal of waste. Paragraph 4 (1) is as follows:

a) "ensuring the waste is recovered or disposed of without endangering human health and without using process or methods which could harm the environment and in particular without:

risk to water, air, soil, plants or animals; or

- causing nuisance through noise or odours; or
- adversely affecting the countryside or places of special interest;
- b) implementing, as far as material, any plan made under the plan-making provisions".

The application of BAT is likely to already address risks to water, air, soil, plants or animals, odour nuisance and some aspects of effects on the countryside. It will, however, be necessary for the Operator to briefly to consider each of these objectives individually and provide a comment on how they are being addressed by the proposals. It is also necessary to ensure that any places of special concern which could be affected, such as SSSIs, are identified and commented upon although, again, these may have been addressed in the assessment for BAT, in which case a cross-reference may suffice.

Operators should identify any development plans made by the local planning authority, including any waste local plan, and comment on the extent to which the proposals accord with the contents of any such plan (see Section 2.6).

4.3 The Habitats Regulations



European sites are defined in Regulation 10 of the Habitats Regulations to include Special Areas of Conservation (SACs); sites of community importance (sites that have been selected as candidate SACs by member states and adopted by the European Commission but which are not yet formally classified); and Special Protection Areas (SPAs). It is also Government policy (set out in PPG 9 on nature conservation) that potential SPAs and candidate SACs should be considered to be European sites for the purposes of Regulation 10.

Information on the location of European Sites and their conservation objectives is available from

- English Nature (01733 455000), http://www.english-nature.org.uk
- Countryside Council for Wales (01248 385620), http://www.ccw.gov.uk
- Scottish Natural Heritage (0131 447 4784), http://www.snh.org.uk
- Joint Nature Conservation Committee (01733 866852), http://www.jncc.gov.uk
- Environment and Heritage Service, Northern Ireland, http://www.ehsni.gov.uk

The Regulator will need to consider the Operator's initial assessment. If it concludes that the installation is likely to have a significant effect on a European site, then the Regulator will need to carry out an "appropriate assessment" of the implications of the installation in view of that site's conservation objectives. The Regulations impose a duty on the Regulator to carry out these assessments so it cannot rely on the Operator's initial assessments. Therefore the Regulator must be provided with any relevant information upon which the Operator's assessment is based.

Note that in many cases the impact of the Habitats Regulations will have been considered at the planning application stage, in which case the Regulator should be advised of the details.

REFERENCES

REFERENCES

For a full list of available Technical Guidance see Appendix A of the *Guide for Applicants* or visit the Environment Agency Website http://www.environment-agency.gov.uk. Many of the references below are being made available free of charge for viewing or download on the Website. The same information can also be accessed via the SEPA web site http://www.sepa.org, or the NIEHS web site www.ehsni.gov.uk. Most titles will also be available in hard copy from The Stationery Office (TSO). Some existing titles are not yet available on the Website but can be obtained from TSO.

- 1. The Pollution Prevention and Control Act (1999) (www.legislation.hmso.gov.uk).
- 2. The Pollution Prevention and Control Regulations (SI 1973 2000) (www.legislation.hmso.gov.uk).
- 3. IPPC: A Practical Guide (for England and Wales) (or equivalents in Scotland and Northern Ireland) (www.environment.detr.gov.uk).
- 4. IPPC Part A(1) Installations: Guide for Applicants (includes Preparation of a Site Report in a Permit Application) (EA Website).
- 5. Assessment methodologies:
 - E1 BPEO Assessment Methodology for IPC
 - IPPC Environmental Assessments for BAT (in preparation as H1)
- 6. Management system references:
 - Sector specific
- 7. Waste minimisation support references:
 - Environment Agency web site. Waste minimisation information accessible via: www.environment-agency.gov.uk/epns
 - Waste Minimisation an environmental good practice guide for industry (helps industry to minimise waste and achieve national environmental goals). Available free to companies who intend to undertake a waste reduction programme (tel 0345 33 77 00)
 - Profiting from Pollution Prevention 3Es methodology (emissions, efficiency, economics). Video and A4 guide aimed at process industries. Available from Environment Agency, North East region (tel 0113 244 0191, ask for regional PIR)
 - Waste Minimisation Interactive Tools (WIMIT). Produced in association with the ETBPP and the BOC Foundation (a software tool designed for small and medium businesses.). Available free from The Environmental Helpline (tel 0800 585794)
 - Environmental Technology Best Practice Programme ETBPP. A joint DTI/DETR programme, with over 200 separate case studies, good practice guides, leaflets, flyers, software tools and videos covering 12 industry sectors, packaging, solvents and the generic areas of waste minimisation and cleaner technology. The ETBPP is accessible via a FREE and confidential helpline (tel 0800 585794) or via the web site www.etsu.com/etbpp/
 - ETBPP, Increased Profit Through Improved Materials Additions: Management/Technical Guide, GG194/195
 - Waste Management Information Bureau. The UK's national referral centre for help on the full range of waste management issues. It produces a database called Waste Info, which is available for online searching and on CD-ROM. Short enguiries are free (tel 01235 463162)
 - Institution of Chemical Engineers Training Package E07 Waste Minimisation. Basic course which contains guide, video, slides, OHPs etc. (tel 01788 578214)
- 8. Water efficiency references:
 - ETBPP, Simple measures restrict water costs, GC22
 - ETBPP, Effluent costs eliminated by water treatment, GC24
 - ETBPP, Saving money through waste minimisation: Reducing water use, GG26
 - ETBPP Helpline 0800 585794
- 9. Environment Agency (1998) Optimum use of water for industry and agriculture dependent on direct abstraction: Best practice manual. R&D technical report W157, WRc Dissemination Centre, Swindon (tel 01793 865012)
- 10. Releases to air references:
 - BREF on Waste Water and Waste Gas Treatment.
 - A1 Guidance on effective flaring in the gas, petroleum etc industries, 1993, ISBN 0-11-752916-8
 - A2 Pollution abatement technology for the reduction of solvent vapour emissions, 1994, £5.00, 0-11-752925-7
 - A3 Pollution abatement technology for particulate and trace gas removal, 1994, £5.00, 0-11-752983-4
 - Landfill gas flaring
 - Part B PG1/3 Boilers and Furnaces 20-50 MW net thermal input (ISBN 0-11-753146-4-7)
 - Part B PG1/4 Gas Turbines 20-50 MW net thermal input (ISBN 0-11-753147-2)

- 11. Releases to water references:
 - BREF on Waste Water and Waste Gas Treatment
 - A4 Effluent Treatment Techniques, TGN A4, Environment Agency, ISBN 0-11-310127-9 (EA website)
 - Environment Agency, Pollution Prevention Guidance Note Above-ground oil storage tanks, PPG 2, gives information on tanks and bunding which have general relevance beyond just oil (EA website)
 - Mason, P. A, Amies, H. J, Sangarapillai, G. Rose, Construction of bunds for oil storage tanks, Construction Industry Research and Information Association (CIRIA), Report 163, 1997, CIRIA, 6 Storey's Gate, Westminster, London SW1P 3AU. Abbreviated versions are also available for masonry and concrete bunds (www.ciria.org.uk online purchase)
- 12. Dispersion Methodology Guide D1 (EA website summary only)
- 13. IPPC Energy Efficiency Guidance Note (the consultation version, available on the website should be used until the final version is published)
- 14. BS 5908: Code of Practice for Fire Precautions in the Chemical and Allied Industries
- 15. Environment Agency, Pollution Prevention Guidance Note Pollution prevention measures for the control of spillages and fire-fighting run-off, PPG 18, gives information on sizing firewater containment systems (EA website)
- 16. Investigation of the criteria for, and guidance on, the landspreading of industrial wastes final report to the DETR, the Environment Agency and MAFF, May 1998
- 17. Agency guidance on the exemption 7 activity (proposed)
- 18. COMAH guides
 - A Guide to the Control of Major Accident Hazards Regulations 1999, Health and Safety Executive (HSE) Books L111, 1999, ISBN 0 07176 1604 5
 - Preparing Safety Reports: Control of Major Accident Hazards Regulations 1999, HSE Books HS(G)190, 1999
 - Emergency Planning for Major Accidents: Control of Major Accident Hazards Regulations 1999, HSE Books HS(G)191, 1999
 - Guidance on the Environmental Risk Assessment Aspects of COMAH Safety Reports, Environment Agency, 1999 (EA website)
 - Guidance on the Interpretation of Major Accidents to the Environment for the Purposes of the COMAH Regulations, DETR, 1999, ISBN 753501 X, available from the Stationery Office
- 19. Assessment and Control of Environmental Noise and Vibration from Industrial Activities (joint Regulator's guidance in preparation)
- 20. Monitoring Guidance (EA website)
 - M1 Sampling facility requirements for the monitoring of particulates in gaseous releases to atmosphere, March 1993, £5.00, ISBN 0-11-752777-7
 - M2 Monitoring emissions of pollutants at source January 1994, £10.00, ISBN 0-11-752922-2
 - M3 Standards for IPC Monitoring Part 1: Standards, organisations and the measurement infrastructure, August 1995, £11.00, ISBN 0-11-753133-2
 - M4 Standards for IPC Monitoring Part 2 : Standards in support of IPC Monitoring, revised 1998
 - MCERTS approved equipment link via http://www.environment-agency.gov.uk/epns "Guidance for Business and Industry";
 - Direct Toxicity Assessment for Effluent Control: Technical Guidance (2000), UKWIR 00/TX/02/07.
- 21. The Categorisation of Volatile Organic Compounds, DOE Research Report No DOE/HMIP/RR/95/009 (EA website)
- 22. Odour Assessment and Control Guidance for Regulators and Industry (joint agencies guidance in preparation)
- 23. "Policy and Practice for the Protection of Groundwater" (PPPG) (EA website)
- 24. Working at Construction and Demolition-sites (PPG 6) (EA website)
- 25. Waste Incineration Directive 2000/76/EC details to be added
- 26. The Hazardous Waste Incineration Directive 94/67/EC details to be added
- 27. Waste Strategy for England and Wales 2000- details to be added
- 28. Clinical Waste Plant Capacity Estimates details to be added
- 29. BAT Report Review of IPC Technical Guidance for Chapter 5 and 1.3c Vol1 March 2000
- 30. Environment Agency Guidance on the Methods for Securing Representative Ash Samples details to be added

DEFINITIONS

BAT	Best Available Techniques – see IPPC A Practical Guide or the Regulations for further definition
BAT Criteria	The criteria to be taken into account when assessing BAT, given in Schedule 2 of the PPC Regulations
BOD	Biological Oxygen Demand
BREF	BAT Reference Document
CEM	Continuous Emissions Monitoring
CHP	Combined heat and power plant
COD	Chemical Oxygen Demand
EMS	Environmental Management System
ETP	Effluent treatment plant
ITEQ	International Toxicity Equivalents
MCERTS	Monitoring Certification Scheme
NIEHS	Northern Ireland Environment and Heritage Service
SAC	Special Areas of Conservation
SECp	Specific Energy consumption
SEPA	Scottish Environmental Protection Agency
SPA	Special Protection Area
TSS	Suspended solids
TOC	Total Organic Carbon

VOC Volatile organic compounds

Further sector specific acronyms to be added

APPENDIX 1 - SOME COMMON MONITORING AND SAMPLING METHODS

Table A1-1 - Continuous monitoring techniques for the incineration of waste and fuel manufactured from and including waste

Pollutant	Suitable technique	Comments on best practice
O ₂	Paramagnetic	The choice of analyser offering the best performance
	Zirconia probe Electrochemical	depends on the availability of MCERTS certified analysers
CO	Non Dispersive Infra-Red (NDIR)	There is no evidence indicating that a particular type of
SO ₂	In situ and extractive NDIR analysers	sensor or system offers unique advantages over other
	Extractive UV fluorescence analyser	extractive ones. Gas parameters such as temperature.
	FTIR- Fourier transform analyser	humidity dust loading or the presence of interfering
	IR-GFC Gas filter correlation analyser	substances must be taken into consideration and assessed whether they are within the range at which a particular
	In situ and extractive NDUV analyser.	system has been certified.
	Differential optical absorption analyser (DOAS)	For the instruments certified under MCERTS there is sufficient information on individual performance characteristics to be able to quantify the measurement uncertainty. The relevant calculations should be conducted in accordance with GUM (see note below*).
NO _x	In situ and extractive NDIR analyser	As above
	In situ and extractive NDUV analyser	
	IR-GFC Gas filter correlation analyser	
	FTIR- Fourier transform analyser	
	Chemiluminescent analyser	
	Differential optical absorption analyser (DOAS)	
PM	Cross-duct opacity analyser	As above. Triboelectric systems may be affected by
	Triboelectric analyser	which the equipment has been certified.
	Back-scatter analyser	
HC	Extractive Flame Ionisation Detector (FID)	As above
HCI	Gas Filter Correlation Infra-Red (GFC IR);	As above
	Fourier Transform Infra-Red (FTIR);	
	Differential Optical Absorption (DOAS)	
	Tunable diode laser (TDL)	
	Ion mobility spectrometry (IMMS)	
HF	Extractive potentiometric	As above
Metals	Continuous ICP MS sensor	Continuous metals monitoring system based on inductively coupled plasma sensor is in existence but there are no commercially available models yet.

*Note: Measurement uncertainty is defined as total expanded uncertainty at 95% confidence interval calculated in accordance with the Guide to the Expression of Uncertainty in Measurement (GUM), ISBN 92-67-10188-9, 1st Ed., Geneva, Switzerland, ISO 1993.

APPENDIX 2 - EQUIVALENT LEGISLATION IN SCOTLAND & NORTHERN IRELAND

The legislation referred to in the text is that for England and Wales. The following are the equivalents for Scotland and Northern Ireland.

Table A.2.1 -	England and Wales	Scotland	Northern Ireland
Equivalent Legislation	PPC Regulations (England and Wales) 2000	PPC (Scotland) Regulations 2000; SI 200/323	
	Waste Management Licensing Regulations SI:1994 1056	Waste Management Licensing Regulations SI:1994 1056	No NI equivalent
	The Water Resources Act 1991	COPA 1974 (S30A-30E equiv. to Part III WRA91) Natural Heritage (Scotland) Act 1991(Part II equiv. to Part I WRA91)	The Water (NI) Order 1999
	SI 1989 No 317: Clean Air, The Air Quality Standards Regulations 1989	SI 1989/317: Clean Air, The Air Quality Standards Regulations 1989	The Air Quality Standards Regulations (Northern Ireland) 1990. Statutory Rules of Northern Ireland 1990 No 145
	SI 1997 No 3043: Environmental Protection, The Air Quality Regulations 1997	SSI 2000/97 The Air Quality (Scotland) Regs	No NI equivalent
	SI 1989 No 2286 and 1998 No 389 the Surface Water (Dangerous Substances Classification) Regulations. (Values for List II substances are contained in SI 1997/2560 and SI 1998/389)	SI 1990/126 Surface Water (Dangerous Substances) (Classification) (Scotland) Regs	Surface Waters (Dangerous Substances) (Classification) Regulations 1998. Statutory Rules of Northern Ireland 1998 No 397 SI1991/1597:
	SI 1991/1597: Bathing Waters (Classification) Regs.	SI 1991/1609 Bathing Waters (Classification) (Scotland) Regs	The Quality of Bathing Water Regulations (NI) 1993
	SI 1992/1331 and Direction 1997 Surface Waters (Fishlife) (Classification) Regs.	SI 1997/2471 Surface Waters (Fishlife) (Classification) Regs	The Surface Water (Fishlife) (Classification) Regulations (NI) 1997
	SI1997/1332 Surface Waters (Shellfish) (Classification) Regs.	SI 1997/2470 Surface Waters (Shellfish) (Classification) Regs	The Surface Water (Shellfish) (Classification) Regulations (NI) 1997
	SI1994/2716 Conservation (Natural Habitats etc) Regulations 1994	SI 1994/2716 Conservation (Natural Habitats etc) Regs	Conservation (Natural Habitats etc) Regulations (Northern Ireland) 1995
	Control of Major Accident Hazards Regulations 1999 (COMAH)	SI 1999/743 Control of Major Accident Hazards Regs	Control of Major Accident Hazard Regulations (Northern Ireland) 2000